UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
 For the quarterly period ended March 27, 2021

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

to

For the transition period from

Commission File Number 001-36285



RAYONIER ADVANCED MATERIALS INC. Incorporated in the State of Delaware I.R.S. Employer Identification No. 46-4559529 1301 RIVERPLACE BOULEVARD, SUITE 2300 JACKSONVILLE, FL 32207 (Principal Executive Office) Telephone Number: (904) 357-4600

Securities registered pursuant to Section 12(b) of the Act:

No 0

Yes x

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, \$0.01 par value	RYAM	New York Stock Exchange
Indicate by check mark whether the registrant (1) has filed all reports require such shorter period that the registrant was required to file such reports), and (5	

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes X No O

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer o

Accelerated filer

Smaller reporting company

.....

Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. O

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes \Box No x The registrant had 63,597,353 shares of common stock, \$.01 par value per share, outstanding as of May 3, 2021.

Table of Contents

Item		<u>Page</u>
	Part I — Financial Information	
1.	Financial Statements (Unaudited)	
	Consolidated Statements of Income (Loss) and Comprehensive Income (Loss) for the Three Months Ended March 27, 2021 and March 28, 2020	1
	Consolidated Balance Sheets as of March 27, 2021 and December 31, 2020	<u>3</u>
	Consolidated Statements of Cash Flows for the Three Months Ended March 27, 2021 and March 28, 2020	<u>4</u>
	Notes to Consolidated Financial Statements	<u>5</u>
2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>20</u>
3.	Quantitative and Qualitative Disclosures about Market Risk	<u>33</u>
4.	Controls and Procedures	<u>34</u>
	Part II — Other Information	
1.	Legal Proceedings	<u>35</u>
1A.	Risk Factors	<u>35</u>
2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>36</u>
6.	Exhibits	<u>37</u>
	Signature	<u>38</u>

Page

Part I. Financial Information

Item 1. Financial Statements

Rayonier Advanced Materials Inc. Consolidated Statements of Income (Loss) (Unaudited)

(Dollars in thousands, except per share amounts)

	Three Months Ended			
		March 27, 2021		March 28, 2020
Net Sales	\$	465,141	\$	409,808
Cost of Sales		(380,981)		(399,347)
Gross Margin		84,160		10,461
Selling, general and administrative expenses		(17,748)		(20,247)
Duties		(6,515)		(6,451)
Foreign exchange gains (losses)		(691)		5,797
Other operating income (expense), net		(4,410)		(1,568)
Operating Income (Loss)		54,796		(12,008)
Interest expense		(17,963)		(15,225)
Interest income and other, net		(845)		423
Other components of pension and OPEB, excluding service costs		1,203		353
Income (Loss) from Continuing Operations Before Income Taxes		37,191		(26,457)
Income tax (expense) benefit (Note 16)		(63,910)		1,622
Equity in income (loss) of equity method investment		(308)		—
Income (Loss) from Continuing Operations		(27,027)		(24,835)
Income from discontinued operations, net of taxes (Note 2)		—		708
Net Income (Loss)	\$	(27,027)	\$	(24,127)
Basic Earnings Per Common Share (Note 13)				
Income (loss) from continuing operations	\$	(0.43)	\$	(0.39)
Income from discontinued operations		—		0.01
Net income (loss) per common share-basic	\$	(0.43)	\$	(0.38)
Diluted Earnings Per Common Share (Note 13)				
Income (loss) from continuing operations	\$	(0.43)	\$	(0.39)
Income from discontinued operations		_		0.01
Net income (loss) per common share-diluted	\$	(0.43)	\$	(0.38)

See Notes to Consolidated Financial Statements.

Rayonier Advanced Materials Inc. Consolidated Statements of Comprehensive Income (Loss) (Unaudited)

(Dollars in thousands)

	Three Months Ended			
	Mar	ch 27, 2021		March 28, 2020
Net Income (Loss)	\$	(27,027)	\$	(24,127)
Other Comprehensive Income (Loss), net of tax (Note 11):				
Foreign currency translation adjustments		(9,268)		(6,471)
Unrealized gain (loss) on derivative instruments		(1,268)		(20,660)
Net gain from pension and postretirement plans		3,302		6,573
Total other comprehensive income (loss)		(7,234)		(20,558)
Comprehensive Income (Loss)	\$	(34,261)	\$	(44,685)

See Notes to Consolidated Financial Statements.

Rayonier Advanced Materials Inc. Consolidated Balance Sheets (Unaudited) (Dollars in thousands, except per share amounts)

	Ma	rch 27, 2021	Decem	ber 31, 2020
Assets				
Current Assets				
Cash and cash equivalents	\$	106,750	\$	93,653
Accounts receivable, net (Note 3)		182,639		179,208
Inventory (Note 4)		295,544		233,484
Income tax receivable		53,309		58,657
Prepaid and other current assets		61,148		68,570
Total current assets		699,390		633,572
Property, Plant and Equipment (net of accumulated depreciation of \$1,633,320 at March 27, 2021 and \$1,610,454 at December 31, 2020)		1,249,390		1,274,942
Deferred Tax Assets		330,012		385,459
Intangible Assets, net		36,689		38,441
Other Assets		199,794		197,451
Total Assets	\$	2,515,275	\$	2,529,865
Liabilities and Stockholders' Equity				
Current Liabilities				
Accounts payable	\$	145,857	\$	156,721
Accrued and other current liabilities (Note 6)		137,850		110,495
Debt due within one year (Note 7)		17,446		17,100
Current environmental liabilities (Note 8)		8,694		8,684
Total current liabilities		309,847		293,000
Long-Term Debt (Note 7)		1,065,163		1,066,837
Long-Term Environmental Liabilities (Note 8) Pension and Other Postretirement Benefits		162,946		162,995
		258,195		260,708
Deferred Tax Liabilities		30,136 30,234		24,462
Other Long-Term Liabilities		50,254		26,776
Commitments and Contingencies (Note 18)				
Stockholders' Equity				
Common stock, 140,000,000 shares authorized at \$0.01 par value, 63,597,356 and				
63,359,839 issued and outstanding, as of March 27, 2021 and December 31, 2020, respectively		636		633
Additional paid-in capital		403,086		405,161
Retained earnings		395,901		422,928
Accumulated other comprehensive income (loss) (Note 11)		(140,869)		(133,635
Total Stockholders' Equity		658,754		695,087
Total Liabilities and Stockholders' Equity	\$	2,515,275	\$	2,529,865

See Notes to Consolidated Financial Statements.

Rayonier Advanced Materials Inc. Consolidated Statements of Cash Flows (Unaudited)

(Dollars in thousands)

	Three Months Ended			
	March 27, 202	1	March 28, 2020	
Operating Activities				
Net income (loss)	\$ (2	7,027)	\$ (24,127	
Loss (income) from discontinued operations			(708	
Adjustments to reconcile income (loss) from continuing operations to cash provided by operating activities:				
Depreciation and amortization	3	6,489	37,842	
Stock-based incentive compensation expense		(653)	1,943	
Deferred income tax expense (benefit)	6	1,698	17,381	
Net periodic benefit cost of pension and other postretirement plans		2,250	2,677	
Unrealized loss (gain) on derivative instruments		—	6,792	
Unrealized loss (gain) from foreign currency		1,599	(11,368	
Other		879	275	
Changes in operating assets and liabilities:				
Receivables	(6,580)	(10,686	
Inventories	(6	3,133)	(24,924	
Accounts payable		(256)	(1,518	
Accrued liabilities	2	7,375	20,707	
All other operating activities		7,156	(25,292	
Contributions to pension and other postretirement plans	(1,582)	(1,944	
Cash Provided by (Used for) Operating Activities-continuing operations	3	8,215	(12,950	
Cash Provided by (Used for) Operating Activities-discontinued operations			204	
Cash Provided by (Used for) Operating Activities	3	8,215	(12,746	
Investing Activities				
Capital expenditures, net	(1	9,525)	(12,582	
Investment in equity method investment		(987)	_	
Cash Used for Investing Activities	(2	0,512)	(12,582	
Financing Activities				
Revolving credit facility and other borrowings		—	7,592	
Repayment of long-term debt	(1,549)	(1,786	
Short-term financing, net		(520)		
Common stock repurchased	(1,420)	(438	
Cash Provided by (Used for) Financing Activities	(3,489)	5,368	
Cash and Cash Equivalents				
Change in cash and cash equivalents	1	4,214	(19,960	
Net effect of foreign exchange on cash and cash equivalents	(1,117)	(1,393	
Balance, beginning of year	9	3,653	64,025	
Balance, end of period	\$ 10	6,750	\$ 42,672	
•				

See Notes to Consolidated Financial Statements.

(Dollar amounts in thousands unless otherwise stated)

1. Basis of Presentation and New Accounting Pronouncements

Basis of Presentation

The unaudited consolidated financial statements and notes thereto of Rayonier Advanced Materials Inc. (the "Company") have been prepared in accordance with generally accepted accounting principles in the United States of America ("GAAP") for interim financial information and in accordance with the rules and regulations of the Securities and Exchange Commission ("SEC"). In the opinion of management, these consolidated financial statements and notes reflect all adjustments (all of which are normal recurring adjustments) necessary for a fair presentation of the results of operations, financial position and cash flows for the periods presented. These statements and notes should be read in conjunction with the consolidated financial statements and supplementary data included in the Company's Annual Report on Form 10-K for the year ended December 31, 2020, as filed with the SEC on March 1, 2021.

The Company has reclassified certain prior year amounts to conform to the current year's presentation for discontinued operations to reflect the November 2019 sale of its Matane high-yield pulp operations. Unless otherwise stated, information in these notes to consolidated financial statements relates to continuing operations. See Note 2 — *Discontinued Operations* for additional information.

Coronavirus Pandemic

During 2020, the Company's businesses were significantly impacted by the coronavirus ("COVID-19") pandemic. While market demand and pricing for certain of the Company's products began to recover towards the end of the year and continued to improve throughout the first quarter of 2021, the Company's operations remain vulnerable to a reversal of these trends or other continuing negative effects caused by COVID-19.

In its operating facilities and work spaces, the Company continues to maintain protocols previously implemented to reduce the potential spread of COVID-19 and ensure the safety of its employees and continuity of operations.

Due to the financial impacts of COVID-19, the Company has actively monitored the recoverability of the carrying value of its long-term assets. During the three months ended March 27, 2021 and the year ended December 31, 2020, the Company did not recognize any impairment charges related to long-lived assets held for use. The Company will continue to evaluate the recoverability of these and other assets as necessary.

New or Recently Adopted Accounting Pronouncements

There have been no new accounting pronouncements not yet effective or adopted in the current year that we believe have a significant impact, or potential significant impact, to our unaudited consolidated interim financial statements.

Subsequent Events

Sale of lumber and newsprint assets

Events and transactions subsequent to the consolidated balance sheets date have been evaluated for potential recognition and disclosure through the date of issuance of these Consolidated Financial Statements. The following subsequent events warranting disclosure were identified.

On April 12, 2021, the Company announced the sale of its lumber and newsprint assets (the "Purchased Assets") located in Ontario and Québec Canada to GreenFirst Forest Products, Inc. ("GreenFirst"), for approximately \$214 million, including an assumed \$74 million associated with finished goods, work-in-process and raw materials inventory value, which is subject to fluctuation until closing. The purchase price, to be adjusted at closing by changes in the value of the inventory, will be paid 85 percent in cash and the remainder in common shares of GreenFirst. In addition, a credit note will be issued to the Company by GreenFirst in the amount of CDN\$8 million, which may be offset against amounts owed to GreenFirst in the future for wood chip purchases, equally over the next 5 years. The closing of the transaction, which is expected to occur in the second half of 2021, but not before July 31, is subject to customary closing conditions, including receipt of regulatory approvals, the transfer of forestry licenses and the approval of the TSX Venture Exchange.

The Purchased Assets include the Company's six lumber facilities, newsprint facility, inventory and certain real property, machinery, inventory, permits, leases, licenses, pension assets and liabilities and other related assets associated with the successful operations of these businesses. Other assets and liabilities, including accounts receivable, accounts payable, certain retained inventory and rights to softwood lumber duties, generated or incurred through the closing date, are excluded. Since 2017, the Company has paid a total of \$98 million in duties.

In connection with the transaction, the Company will enter into a 20 year wood chip and residual fiber supply agreement with GreenFirst, securing supply for the Company's operations at the Temiscaming plant. Additionally, the parties entered into a Transition Services Agreement ("TSA") whereby the Company will provide certain transitional services to GreenFirst, for a period of time following the closing of the transaction, not to exceed twelve months from such date. The TSA includes support related to information technology, accounting, treasury, human resources and payroll, tax, supply chain and procurement functions. Costs incurred by the Company associated with the TSA will be reimbursed by GreenFirst.

As of March 27, 2021, the carrying value of the net assets included in the transaction was \$238 million, including the carrying value of inventory which was \$107 million. The Company expects a cash tax impact in 2022 of less than \$10 million as a result of this transaction, although the full extent of tax impacts are still being evaluated.

Contingency resolution

On April 19, 2021, the Company reached final settlement in its dispute with the Market Assessment and Compliance Division ("MACD") branch of the Independent Electricity System Operator ("IESO") regarding the investigation of the Kapuskasing facility. See Note 18 — *Contingencies and Guarantees* for additional information.

2. Discontinued Operations

In November 2019, the Company sold its Matane, Quebec pulp mill to Sappi Limited, a global diversified wood fiber company, for a gross purchase price of approximately \$175 million. Income from discontinued operations for the three months ended March 28, 2020 represents an adjustment to the gain on sale of the Matane mill from working capital adjustments that arose following the November 2019 closing.

3. Accounts Receivable, Net

The Company's accounts receivable included the following:

	Marcl	March 27, 2021		ber 31, 2020
Accounts receivable, trade	\$	152,194	\$	140,036
Accounts receivable, other (a)		30,935		39,659
Allowance for doubtful accounts		(490)		(487)
Total accounts receivable, net	\$	182,639	\$	179,208

(a) Accounts receivable, other consists primarily of value added/consumption taxes, grants receivable and accrued billings due from government agencies.

4. Inventory

The Company's inventory included the following:

	March 27, 2021		December 31, 2020
Finished goods	\$ 163,618	\$	138,064
Work-in-progress	19,243		17,246
Raw materials	104,189		70,009
Manufacturing and maintenance supplies	8,494		8,165
Total inventory	\$ 295,544	\$	233,484

5. Leases

The Company's operating and finance leases are primarily for corporate offices, warehouse space, rail cars and equipment. As of March 27, 2021, the Company's leases have remaining lease terms of 1 year to 7.9 years with standard renewal and termination options available at the Company's discretion. Certain equipment leases have purchase options at the end of the term of the lease, which are not included in the Right of Use ("ROU") assets as it is not reasonably certain that the Company will exercise such options. The Company's lease agreements do not contain any material residual value guarantees or material restrictive covenants.

The Company uses its incremental borrowing rate in determining the present value of lease payments unless the lease provides an implicit or explicit interest rate. The weighted average discount rate used in determining the operating lease ROU assets and liabilities as of March 27, 2021 and December 31, 2020 was 5.8 percent and 6.1 percent, respectively. The weighted average discount rate used in determining the finance lease ROU assets and liabilities as of March 27, 2021 and December 31, 2020 was 7.0 percent.

The Company's operating and finance lease cost is as follows:

		Three Months Ended				
	Marc	March 27, 2021		rch 28, 2020		
Operating Leases						
Operating lease expense	\$	1,841	\$	1,804		
Finance Leases						
Amortization of ROU assets		86		80		
Interest		43		49		
Total	\$	1,970	\$	1,933		

As of March 27, 2021, the weighted average remaining lease term is 3.4 years and 5.7 years for operating leases and financing leases, respectively. As of December 31, 2020, the weighted average remaining lease term is 3.6 years and 5.9 years for operating leases and finance leases, respectively. Cash provided by operating activities includes approximately \$2 million and \$2 million from operating lease payments made during the three months ended March 27, 2021 and March 28, 2020, respectively. Finance lease cash flows were immaterial during the three months ended March 27, 2021 and March 28, 2020.

As of March 27, 2021 and December 31, 2020, assets acquired under finance leases of \$2 million and \$2 million, respectively, are reflected in Property, Plant and Equipment, net. The Company's finance leases are included as debt and the maturities for the remainder of 2021 and the next four years and thereafter are included in Note 7 — *Debt and Finance Leases*. The Company's consolidated balance sheet includes the following operating lease assets and liabilities:

	Balance Sheet Classification	Mare	ch 27, 2021	December 31, 2020
Right-of-use assets	Other assets	\$	18,238	\$ 17,566
Lease liabilities, current	Accrued and other current liabilities	\$	6,266	\$ 5,666
Lease liabilities, non-current	Other long-term liabilities	\$	13,140	\$ 13,007

As of March 27, 2021, operating lease maturities for the remainder of 2021 through 2025 and thereafter are as follows:

	Mar	ch 27, 2021
Remainder of 2021	\$	5,437
2022		6,819
2023		5,713
2024		1,835
2025		789
Thereafter		841
Total minimum lease payments	\$	21,434
Less: imputed interest		(2,028)
Present value of future minimum lease payments	\$	19,406

6. Accrued and Other Current Liabilities

The Company's accrued and other current liabilities included the following:

	March 27, 2021	December 31, 2020
Accrued customer incentives and prepayments	\$ 24,651	\$ 29,387
Accrued payroll and benefits	30,118	21,500
Accrued interest	19,343	3,230
Accrued income taxes	6,000	5,052
Accrued stumpage	19,017	10,045
Accrued property and other taxes	6,194	3,995
Other current liabilities	32,527	37,286
Total accrued and other current liabilities	\$ 137,850	\$ 110,495

7. Debt and Finance Leases

The Company's debt and finance leases included the following:

		March 27, 2021	December 31, 2020
ABL Credit Facility due 2025, \$142 million available, bearing interest 0.25% LIBOR floor plus 2.75%, interest rate of 3.00% at March 27, 2021			_
Senior Secured Notes due 2026 at a fixed interest rate of 7.625%	\$	500,000	\$ 500,000
Senior Notes due 2024 at a fixed interest rate of 5.5%		495,647	495,647
Canadian dollar, fixed interest rate term loans with rates ranging from 5.5% to 6.86% and maturity dates ranging from July 2022 through April 2028, secured by certain assets of the Temiscaming mill		73,686	73,791
Other loans (a)		17,082	18,193
Short-term factoring facility-France		4,471	5,089
Finance lease obligation		2,404	2,489
Total debt principal payments due		1,093,290	1,095,209
Less: Debt premium, original issue discount and issuance costs, net		(10,681)	(11,272)
Total debt		1,082,609	1,083,937
Less: Debt due within one year		(17,446)	(17,100)
Long-term debt	\$	1,065,163	\$ 1,066,837
(a) Primarily loans for energy projects in France.	-		

As of March 27, 2021, debt and finance lease payments due during the remainder of 2021, the next four years and thereafter are as follows:

	Finance Lea	Finance Lease Payments				
Remainder of 2021	\$	386	\$	14,649		
2022		515		30,426		
2023		515		10,326		
2024		515		505,895		
2025		515		10,290		
Thereafter		473		519,300		
Total principal payments	\$	2,919	\$	1,090,886		
Less: Imputed interest		515				
Present value minimum finance lease payments	\$	2,404				

8. Environmental Liabilities

An analysis of liabilities for the three months ended March 27, 2021 is as follows:

Balance, December 31, 2020	\$ 171,679
Increase in liabilities	225
Payments	(419)
Foreign currency adjustments	155
Balance, March 27, 2021	171,640
Less: Current portion	(8,694)
Long-term environmental liabilities	\$ 162,946

In addition to the estimated liabilities, the Company is subject to the risk of reasonably possible additional liabilities in excess of the established reserves due to potential changes in circumstances and future events, including, without limitation, changes to current laws and regulations; changes in governmental agency personnel, direction, philosophy and/or enforcement policies; developments in remediation technologies; increases in the cost of remediation, operation, maintenance and monitoring of its environmental liability sites; changes in the volume, nature or extent of contamination to be remediated or monitoring to be undertaken; the outcome of negotiations with governmental agencies and non-governmental parties; and changes in accounting rules or interpretations. Based on information available as of March 27, 2021, the Company estimates this exposure could range up to approximately \$78 million, although no assurances can be given that this amount will not be exceeded given the factors described above. These potential additional costs are attributable to several sites and other applicable liabilities. Further, this estimate excludes reasonably possible liabilities which are not currently estimable primarily due to the factors discussed above.

Subject to the previous paragraph, the Company believes established liabilities are sufficient for probable costs expected to be incurred over the next 20 years with respect to its environmental liabilities. However, no assurances are given they will be sufficient for the reasons described above, and additional liabilities could have a material adverse effect on the Company's financial position, results of operations and cash flows.

9. Derivative Instruments

The Company's earnings and cash flows are subject to fluctuations due to changes in interest rates and foreign currency exchange rates. The Company allows for the use of derivative financial instruments to manage interest rate and foreign currency exchange rate exposure but does not allow derivatives to be used for speculative purposes.

All derivative instruments are recognized on the consolidated balance sheets at their fair value and are either designated as a hedge of a forecasted transaction or undesignated. Changes in the fair value of a derivative designated as a hedge are recorded in other comprehensive income until earnings are affected by the hedged transaction and are then reported in current earnings. Changes in the fair value of undesignated derivative instruments and the ineffective portion of designated derivative instruments are reported in current earnings.

In December 2020, the Company terminated all outstanding derivative instruments, which had been previously designated as hedging instruments and had various maturity dates through 2028. Accumulated gains and losses associated with these instruments were deferred as a component of accumulated other comprehensive income (loss), totaling a net after tax gain of \$2 million as of December 31, 2020, to be recognized in earnings as the underlying hedged transactions occur and affect earnings. During the three months ended March 27, 2021, the Company recognized a \$1.3 million after-tax gain associated with the deferred component in accumulated other comprehensive income (loss) related to these settlements. A \$0.6 million net after tax gain remains deferred within accumulated other comprehensive income (loss) as of March 27, 2021, which will be recognized in earnings as the underlying hedged transactions occur and affect earnings.

Interest Rate Risk

The Company's current debt obligations are primarily fixed and therefore not materially exposed to variability in interest payments due to changes in interest rates. The Company previously entered into interest rate swap agreements to reduce the volatility of interest expense, achieve a desired proportion of fixed-rate versus floating-rate debt and to hedge the variability in cash flows attributable to interest rate risks caused by changes in the LIBOR benchmark.

The Company had designated the swaps as cash flow hedges and assesses their effectiveness using the hypothetical derivative method in conjunction with regression. Effective gains and losses deferred to AOCI are reclassified into earnings over the life of the associated hedge. Ineffective gains and losses are classified to earnings immediately. There was no hedge ineffectiveness during 2020.

Foreign Currency Exchange Rate Risk

Foreign currency fluctuations affect investments in foreign subsidiaries and foreign currency cash flows related to third party purchases, product shipments, and foreign-denominated debt. The Company is also exposed to the translation of foreign currency earnings to the U.S. dollar. Management may use foreign currency forward contracts to selectively hedge its foreign currency cash flows exposure and manage risk associated with changes in currency exchange rates. The Company's principal foreign currency exposure is to the Canadian dollar, and to a lesser extent, the euro.

The effects of derivatives designated as hedging instruments, the related changes in AOCI and the gains and losses in income is as follows:



	Three Months End		
Derivatives Designated as Hedging Instruments	Recognized in OCI erivative	Location on Statement of Income	
Interest rate swaps	\$ _	\$ 	Interest expense
Foreign exchange forward contracts	\$ _	\$ _	Other operating income (expense), net
Foreign exchange forward contracts	\$ _	\$ 1,828	Cost of sales
Foreign exchange forward contracts	\$ 	\$ (100)	Interest income and other, net

	ss) Recognized in OCI on Derivative	Location on Statement of Income	
Interest rate swaps	\$ (1,757)	\$ (120)	Interest expense
Foreign exchange forward contracts	\$ (26,532)	\$ (1,361)	Other operating income (expense), net
Foreign exchange forward contracts	\$ (394)	\$ 394	Cost of sales
Foreign exchange forward contracts	\$ (6,902)	\$ (6,732)	Interest income and other, net

The effects of derivative instruments not designated as hedging instruments on the consolidated statement of income were as follows:

		Three Months Ended		
Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income on Derivative	March 27, 2021	March 28, 202	0
Foreign exchange forward contracts	Other operating income (expense), net	\$ —	\$	(721)

The after-tax amounts of unrealized gains (losses) in AOCI related to hedge derivatives are presented below:

	March 27, 2021		December 31, 2020
Foreign exchange cash flow hedges	\$	566	\$ 1,834

The amount of future reclassifications from AOCI will fluctuate with movements in the underlying markets.

10. Fair Value Measurements

The following table presents the carrying amount, estimated fair values and categorization under the fair value hierarchy for financial instruments held by the Company, using market information and what management believes to be appropriate valuation methodologies:

	_	March 27, 2021											
		Carrying -		Fair Value			Correcting		Carrying		 Fair	Valı	ue
Assets:		Amount		Level 1		Level 2		Amount	Level 1		Level 2		
Cash and cash equivalents	\$	106,750	\$	106,750	\$	—	\$	93,653	\$ 93,653	\$			
Liabilities (a):													
Fixed-rate long-term debt		1,075,734		—		1,104,025		1,076,359	—		1,050,287		

(a) Liabilities exclude finance lease obligation.

The Company uses the following methods and assumptions in estimating the fair value of its financial instruments:

Cash and cash equivalents — The carrying amount is equal to fair market value.

Debt — The fair value of fixed rate debt is based upon quoted market prices for debt with similar terms and maturities.

11. Accumulated Other Comprehensive Income (Loss)

The components of AOCI are as follows:

	Three Months Ended				
	Ma	rch 27, 2021	Ma	rch 28, 2020	
Unrecognized components of employee benefit plans, net of tax:					
Balance, beginning of year	\$	(146,614)	\$	(126,638)	
Other comprehensive gain (loss) before reclassifications		—		4,781	
Income tax on other comprehensive loss		—		(1,238)	
Reclassifications to earnings: (a)					
Amortization of losses		4,082		3,393	
Amortization of prior service costs		138		141	
Income tax on reclassifications		(918)		(504)	
Foreign currency adjustments		_		_	
Net comprehensive gain (loss) on employee benefit plans, net of tax		3,302		6,573	
Balance, end of quarter		(143,312)		(120,065)	
		· · · · · · · · · · · · · · · · · · ·		· · · · ·	
Unrealized gain (loss) on derivative instruments, net of tax:					
Balance, beginning of year		1,834		1,290	
Other comprehensive gain (loss) before reclassifications				(35,585)	
Income tax on other comprehensive income		_		8,255	
Reclassifications to earnings: (b)					
Interest rate contracts		_		120	
Foreign exchange contracts		(1,728)		7,699	
Income tax on reclassifications		460		(1,149)	
Net comprehensive gain (loss) on derivative instruments, net of tax		(1,268)		(20,660)	
Balance, end of quarter		566		(19,370)	
Foreign currency translation adjustments:					
Balance, beginning of year		11,145		(13,879)	
Foreign currency translation adjustment, net of tax of \$0 and \$0		(9,268)		(6,471)	
Balance, end of quarter		1,877	-	(20,350)	
Accumulated other comprehensive income (loss), end of quarter	\$	(140,869)	\$	(159,785)	

(a) The AOCI components for defined benefit pension and post-retirement plans are included in the computation of net periodic benefit cost. See Note 15 — *Employee Benefit Plans* for additional information.

(b) Reclassifications of interest rate contracts are recorded in interest expense. Reclassifications of foreign currency exchange contracts are recorded in cost of sales, other operating income or non-operating income as appropriate. See Note 9 — *Derivative Instruments* for additional information.

12. Stockholders' Equity

An analysis of stockholders' equity is shown below (share amounts not in thousands):

	Common Stock										
	Shares		Par Value			Additional Paid in Retained Capital Earnings		Accumulated Other Comprehensive Loss		To	tal Stockholders' Equity
For the three months ended March 27, 2021											
Balance, January 1, 2021	63,359,839	\$	633	\$	\$ 405,161	\$	422,928	\$	(133,635)	\$	695,087
Net income (loss)	—		—				(27,027)		—		(27,027)
Other comprehensive income (loss), net of tax	_				_		_		(7,234)		(7,234)
Issuance of common stock under incentive stock plans	369,713		4		(4)		_				_
Stock-based compensation	_				(653)		_		_		(653)
Repurchase of common shares (a)	(132,196)		(1)		(1,418)		_		_		(1,419)
Balance, March 27, 2021	63,597,356	\$	636	\$	\$ 403,086	\$	395,901	\$	(140,869)	\$	658,754
For the three months ended March 28, 2020											
Balance, January 1, 2020	63,136,129	\$	632	\$	\$ 399,020	\$	422,373	\$	(139,227)	\$	682,798
Net income (loss)	—				—		(24,127)		—		(24,127)
Other comprehensive income (loss), net of tax	—				—		_		(20,558)		(20,558)
Issuance of common stock under incentive stock plans	290,689		3		(3)		_		_		_
Stock-based compensation	_		_		1,943		_		_		1,943
Repurchase of common shares (a)	(179,951)		(3)		(435)		_		_		(438)
Balance, March 28, 2020	63,246,867	\$	632	\$	\$ 400,525	\$	398,246	\$	(159,785)	\$	639,618

(a) Repurchased to satisfy the tax withholding requirements related to the issuance of stock under the Rayonier Advanced Materials Incentive Stock Plan.

Common Stock Buyback

On January 29, 2018, the Board of Directors authorized a share buyback program pursuant to which the Company may, from time to time, purchase shares of its common stock with an aggregate purchase price of up to \$100 million. During the three months ended March 27, 2021 and March 28, 2020, the Company did not repurchase any common shares under this buyback program. As of March 27, 2021, there was approximately \$60 million of share repurchase authorization remaining under the program. The Company does not expect to utilize any further authorization in the near future.



13. Earnings Per Share of Common Stock

The following table provides details of the calculations of basic and diluted earnings per share:

	Three Months Ended				
		March 27, 2021		March 28, 2020	
Income (loss) from continuing operations	\$	(27,027)	\$	(24,835)	
Income (loss) from discontinued operations		—		708	
Net income (loss) available for common stockholders	\$	(27,027)	\$	(24,127)	
Shares used for determining basic earnings per share of common stock		63,430,601		62,982,735	
Dilutive effect of:					
Stock options		—		—	
Performance and restricted stock		—		—	
Preferred stock		—		—	
Shares used for determining diluted earnings per share of common stock		63,430,601	_	62,982,735	
Basic per share amounts					
Income (loss) from continuing operations	\$	(0.43)	\$	(0.39)	
Income (loss) from discontinued operations		—		0.01	
Net income (loss)	\$	(0.43)	\$	(0.38)	
Diluted per share amounts					
Income (loss) from continuing operations	\$	(0.43)	\$	(0.39)	
Income (loss) from discontinued operations		—		0.01	
Net income (loss)	\$	(0.43)	\$	(0.38)	

Anti-dilutive instruments excluded from the computation of diluted earnings per share:

	Three Months Ended				
	March 27, 2021 Marc				
Stock options	122,525	157,033			
Performance and restricted stock	2,338,111	448,812			
Total anti-dilutive instruments	2,460,636	605,845			

14. Incentive Stock Plans

The Company's total stock-based compensation for the three months ended March 27, 2021 and March 28, 2020 was a benefit of \$1 million and expense of \$2 million, respectively.

The Company made new grants of restricted stock units and performance-based stock units to certain employees during the first three months of 2021. The 2021 restricted stock unit awards cliff vest after three years. The 2021 performance-based stock unit awards measure total shareholder return ("TSR") on an absolute basis and relative to peers. Participants can earn between 0 and 200 percent of the target award. Performance below the threshold for the absolute TSR would result in a 0 payout for the TSR metric. There is a performance-based stock award and cash unit stock award that will be measured using the same objectives but paid and accounted for separately. As required by Accounting Standards Codification 718, *Compensation-Stock Compensation*, the portion of the award to be settled in cash is classified as a liability and remeasured to fair value at the end of each reporting period until settlement.

In March 2021, the performance-based share units granted in 2018 were settled at an average of 60 percent of the performance-based stock units awarded, resulting in the issuance of 182,811 shares of common stock.

The following table summarizes the activity on the Company's incentive stock awards for the three months ended March 27, 2021:

	Stock Options			Restricted Stoc	k ar	nd Stock Units	Performance-Based Stock Units			
	Options		ighted Average xercise Price	Awards		Veighted Average Grant Date Fair Value	Awards		eighted Average Grant Date Fair Value	
Outstanding at January 1, 2021	152,281	\$	38.26	828,955	\$	10.27	1,821,402	\$	8.77	
Granted	—		—	409,912		10.46	135,699		17.61	
Forfeited	_			(71,647)		14.11	(296,864)		9.01	
Exercised or settled	_		_	(186,830)		22.21	(302,516)		22.76	
Expired or cancelled	(29,756)		33.39	_		—	_		_	
Outstanding at March 27, 2021	122,525	\$	39.44	980,390	\$	7.79	1,357,721	\$	6.49	

15. Employee Benefit Plans

The Company has defined benefit pension and other long-term and postretirement benefit plans covering certain union and non-union employees, primarily in the U.S., Canada and France. The defined benefit pension plans are closed to new participants. The liabilities for these plans are calculated using actuarial estimates and management assumptions. These estimates are based on historical information, along with certain assumptions about future events.

The components of net periodic benefit costs from these plans that have been recorded are shown in the following table:

	Pension						Postretirement				
]	Three Mo	nths	s Ended	Three Months Ended						
Components of Net Periodic Benefit Cost	March 27, 2021 March 2			March 28, 2020	March	n 27, 2021	March 28, 2020				
Service cost	\$	3,083	\$	2,667	\$	370	\$	363			
Interest cost		4,806		6,340		207		318			
Expected return on plan assets		(10,436)		(10,545)		—		—			
Amortization of prior service cost		176		179		(38)		(38)			
Amortization of losses		4,079		3,427		3		(34)			
Total net periodic benefit cost	\$	1,708	\$	2,068	\$	542	\$	609			

Service cost is included in cost of sales and selling, general and administrative expenses in the statements of income, as appropriate. Interest cost, expected return on plan assets, amortization of prior service cost and amortization of losses are included in other components of pension and OPEB, excluding service cost on the consolidated statement of income.

16. Income Taxes

The Company's effective tax rate from continuing operations for the three months ended March 27, 2021 was an expense of 172 percent, compared with a benefit of 6 percent for the three months ended March 28, 2020.

The current quarter March 27, 2021 effective rate differs from the federal statutory rate of 21 percent primarily due to Global Intangible Low Taxed Income ("GILTI"), disallowed interest deductions in the U.S., and different statutory tax rates of foreign operations. The application of GILTI effectively results in double book taxation of the majority of the Company's high Canadian earnings. The Company currently expects minimal cash taxes to be paid in 2021 or 2022 as a result of 2021 earnings. The effective tax rate benefit for the three months ended March 28, 2020 differs from the federal statutory rate primarily due to nondeductible interest expense in the U.S. and lower tax deductions on vested stock compensation, partially offset by benefits from the CARES Act (see below).

On March 27, 2020, the United States Congress passed the CARES Act to provide taxpayer protection against the economic impacts of COVID-19. As part of the CARES Act, the Company is able to carry 2019 and 2020 tax net operating losses back to tax years when the U.S. Federal Statutory rate was 35 percent compared with the current 21 percent. To date, the

Company has recorded a \$33 million current receivable related the 2019 loss carryback which is expected to be received in 2021 and a \$9 million noncurrent receivable related to the 2020 loss carryback expected to be filed in the second quarter of 2021. Separately, the Company has a \$22 million receivable related to tax years under examination by the IRS, \$17 million of which is expected to be received within the next twelve months.

There have been no material changes to the balance of unrecognized tax benefits reported at December 31, 2020.

17. Segment and Geographical Information

The Company currently operates in the following five business segments: High Purity Cellulose, Forest Products, Paperboard, Pulp & Newsprint and Corporate. All prior period amounts presented herein have been reclassified to conform to this segment structure. The Corporate operations consist primarily of senior management, accounting, information systems, human resources, treasury, tax and legal administrative functions that provide support services to the operating business units. The Company allocates a portion of the cost of maintaining these support functions to its operating units.

The Company evaluates the performance of its segments based on operating income. Intersegment sales consist primarily of wood chips sales from Forest Products to High Purity Cellulose, Paperboard and Pulp & Newsprint segments and high-yield pulp sales from Pulp & Newsprint to Paperboard. Intersegment sales prices are at rates that approximate market for the respective operating area.

Net sales, disaggregated by product-line, was comprised of the following:

	Three Months Ended					
	March 27, 2021			March 28, 2020		
High Purity Cellulose						
Cellulose Specialties	\$	167,837	\$	160,235		
Commodity Products		58,933		66,524		
Other sales (a)		22,991	_	22,819		
Total High Purity Cellulose		249,761		249,578		
Forest Products						
Lumber		127,711		60,549		
Other sales (b)		19,060		21,750		
Total Forest Products		146,771		82,299		
Paperboard						
Paperboard		47,849		50,486		
Pulp & Newsprint						
Pulp		27,544		29,975		
Newsprint		11,491		16,703		
Total Pulp & Newsprint		39,035		46,678		
Eliminations		(18,275)		(19,233)		
Total net sales	\$	465,141	\$	409,808		

(a) Other sales include sales of electricity, lignin and other by-products to third-parties

(b) Other sales include sales of logs, wood chips and other by-products to other Company segments and third-parties

Operating income (loss) by segment was comprised of the following:

	Three Months Ended					
	March 27, 2021	March 28, 2020				
High Purity Cellulose	\$ 6,42	7 \$ (4,858)				
Forest Products	60,75	0 (1,311)				
Paperboard	5,75	5 4,674				
Pulp & Newsprint	(5,90)) (5,876)				
Corporate	(12,23	6) (4,637)				
Total operating income (loss)	\$ 54,79	5 (12,008)				

Identifiable assets by segment were as follows:

	 March 27, 2021	December 31, 2020
High Purity Cellulose	\$ 1,511,465	\$ 1,528,929
Forest Products	235,996	186,321
Paperboard	127,525	129,871
Pulp & Newsprint	100,844	99,374
Corporate	539,445	585,370
Total identifiable assets	\$ 2,515,275	\$ 2,529,865

18. Commitments and Contingencies

Commitments

The Company has no material changes to the purchase obligations presented in Note 22 — *Commitments and Contingencies* in the Company's Annual Report on Form 10-K for the year ended December 31, 2020, as filed with the SEC on March 1, 2021, that are outside the normal course of business for the three months ended March 27, 2021. The Company's purchase obligations continue to primarily consist of commitments for the purchase of natural gas, steam energy and electricity contracts entered into within the normal course of business.

The Company leases certain buildings, machinery and equipment under various operating leases. See Note 5 — *Leases*, for additional information.

Litigation and Contingencies

<u>Final Settlement Reached in Dispute with IESO Relating to Investigation of the Kapuskasing Newsprint Facility.</u> From the period from 2014 to early 2021, the Market Assessment and Compliance Division ("MACD") branch of the Independent Electricity System Operator ("IESO"), the governmental agency responsible for operating the wholesale electricity market and directing the operation of the bulk electrical system in the province of Ontario, Canada, had been engaged in reviewing the Company's compliance with the published rules that govern the operation of the wholesale electricity market in Ontario, Canada. The inquiry was focused primarily on payments made by IESO to the Company between 2010 and 2019 under market rules in connection with multiple planned, extended and unplanned forced outages that caused extensive downtime, in full or in part, of the Company's Kapuskasing, Ontario newsprint facility.

In May 2020, MACD finalized two of its four investigations into the Company's electricity management practices at its Kapuskasing newsprint facility and issued orders asserting penalties of CAD \$25 million. These orders called for the Company to pay penalties of CAD \$3 million immediately and CAD \$12 million over a 10 year period, with the remaining CAD \$10 million to be deferred and ultimately forgiven assuming the Company otherwise complied with the orders' remaining terms. The Company, which maintained it had complied in all material respects with the published rules, vigorously contested IESO's orders, including through the filing of judicial review proceedings with the divisional Court (Superior Court of Justice) of Ontario seeking invalidation of the orders. At the time these orders were issued, the remaining two investigations remained open, subjecting the Company to the risk that MACD may in the future issue additional orders upon finalization of these additional investigations.



On April 19, 2021, the Company and IESO entered into Minutes of Settlement ("MOS") pursuant to which the parties agreed to fully and finally settle all claims relating to all four of the investigations (whether completed or not) and related orders, the judicial review proceedings and underlying disputes. As part of the settlement, the Company agreed to a fixed obligation to pay a sum of CAD \$12 million over a period of 5 years comprised of a CAD \$4.5 million up-front payment and a CAD \$7.5 million payment to be spread (on a front-weighted basis) over the next 5 anniversaries of the MOS, without interest. In addition to the foregoing, the MOS provides that a "suspended" sum of CAD \$10.4 million would become due and payable in the event the Company fails to comply with any of the terms and conditions of the MOS or commits an event of default, as defined under the applicable market rules, unless such breach or event of default is remedied on a timely basis. This contingent "suspended" sum decreases annually as the scheduled fixed, or non-contingent, payments are made under the MOS. Assuming no uncured event of default or breach occurs during the repayment period, upon full payment of the CAD \$12 million, the entire "suspended" sum shall be extinguished and RYAM shall be released from any payment obligation with respect thereto.

Given the parties' finalization of and entry into the MOS, the Company considers this matter concluded (subject only to the parties' obligations yet to be performed under the MOS).

Duties on Canadian softwood lumber sold to the U.S. The Company operates six softwood lumber mills in Ontario and Quebec, Canada and exports softwood lumber into the United States from Canada. In 2017, anti-dumping and countervailing duties were assessed by the United States Department of Commerce ("USDOC") on lumber exported into the United States, with the Company being assigned an anti-dumping duty rate of 6 percent and a countervailing duty rate of 14 percent. In December 2020, following its administrative review of the period of April 28, 2017 through December 31, 2018, USDOC determined revised rates for anti-dumping and countervailing duties, and the Company is now subject to an anti-dumping duty rate of approximately 1.6 percent and a countervailing duty rate of approximately 7.4 percent. The reduced rates will be applied by the Company for lumber sold into the U.S. in the future. Canada's legal challenge to the USDOC's assessment of duties continues in spite of the recent revision in rates.

The Company has paid approximately \$98 million in lumber duties to date, recorded as expense in the periods incurred. The Company currently has a \$21 million long-term receivable associated with the December 2020 determination of the revised rates for the 2017 and 2018 periods. Cash is not expected to return to the Company until final resolution of the softwood lumber dispute, which remains subject to legal challenges and to USDOC further administrative review processes covering periods after December 31, 2018.

<u>Other</u>. In addition to the above, the Company is engaged in various legal and regulatory actions and proceedings, and has been named as a defendant in various lawsuits and claims arising in the ordinary course of its business. While the Company has procured reasonable and customary insurance covering risks normally occurring in connection with its businesses, the Company has in certain cases retained some risk through the operation of self-insurance, primarily in the areas of workers' compensation, property insurance and general liability. These other lawsuits and claims, either individually or in aggregate, are not expected to have a material adverse effect on the Company's financial position, results of operations or cash flows.

Guarantees and Other

The Company provides financial guarantees as required by creditors, insurance programs and various governmental agencies. As of March 27, 2021, the Company had net exposure of \$44 million from various standby letters of credit, primarily for financial assurance relating to environmental remediation, credit support for natural gas and electricity purchases, and guarantees related to foreign retirement plan obligations. These standby letters of credit represent a contingent liability. The Company would only be liable upon its default on the related payment obligations. The letters of credit have various expiration dates and will be renewed as required.

The Company had surety bonds of \$85 million as of March 27, 2021, primarily to comply with financial assurance requirements relating to environmental remediation and post closure care, to provide collateral for the Company's workers' compensation program, and to guarantee taxes and duties for products shipped internationally. These surety bonds expire at various dates and are expected to be renewed annually as required.

LignoTech Florida ("LTF") is a venture in which the Company owns 45 percent and its partner Borregaard ASA owns 55 percent. The Company is a guarantor of LTF's financing agreements and, in the event of default, expects it would only be liable for its proportional share of any repayment under the agreements. The Company's proportion of the LTF financing agreement guarantee was \$33 million at March 27, 2021.

The Company has not recorded any liabilities for these financial guarantees in its consolidated balance sheets, either because the Company has recorded the underlying liability associated with the guarantee or the guarantee is dependent on the

Company's own performance and, therefore, is not subject to the measurement requirements or because the Company has calculated the estimated fair value of the guarantee and determined it to be immaterial based upon the current facts and circumstances that would trigger a payment obligation.

It is not possible to determine the maximum potential amount of the liability under these potential obligations due to the unique set of facts and circumstances likely to be involved with each provision.

As of March 27, 2021, all of the Company's collective bargaining agreements covering its unionized employees are current.

Supplemental Disclosures of Cash Flow Information 19.

Supplemental disclosures of cash flows information were comprised of the following for the three months ended:

	 March 27, 2021	 March 28, 2020
Interest paid	\$ 851	\$ 8,115
Income taxes paid (received)	\$ 54	\$ (390)
Capital assets purchased on account	\$ 12,848	\$ 7,622
Assets acquired under operating leases	\$ 1,768	\$ 119

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

When we refer to "we," "us," "our" or "the Company," we mean Rayonier Advanced Materials Inc. and its consolidated subsidiaries. References herein to "Notes to Consolidated Financial Statements" refer to the Notes to the Consolidated Financial Statements of Rayonier Advanced Materials Inc. included in Item 1 of this Quarterly Report on Form 10-Q (the "Report.")

This Management's Discussion and Analysis of Financial Condition and Results of Operations ("MD&A") is intended to provide a reader of our consolidated financial statements with a narrative from the perspective of management on our financial condition, results of operations, liquidity and certain other factors which may affect future results. This MD&A should be read in conjunction with our 2020 Annual Report on Form 10-K and information contained in our subsequent Forms 8-K and other reports to the U.S. Securities and Exchange Commission (the "SEC").

Note About Forward-Looking Statements

Certain statements in this Report regarding anticipated financial, business, legal or other outcomes including business and market conditions, outlook and other similar statements relating to Rayonier Advanced Materials' ("the Company") future events, developments, or financial or operational performance or results, are "forward-looking statements" made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995 and other federal securities laws. These forward-looking statements are identified by the use of words such as "may," "will," "should," "expect," "estimate," "believe," "intend," "forecast," "anticipate" "guidance" and other similar language. However, the absence of these or similar words or expressions does not mean a statement is not forward-looking. While we believe these forward-looking statements are reasonable when made, forward-looking statements are not guarantees of future performance or events and undue reliance should not be placed on these statements. Although we believe the expectations reflected in any forward-looking statements are based on reasonable assumptions, we can give no assurance these expectations will be attained, and it is possible actual results may differ materially from those indicated by these forward-looking statements due to a variety of risks and uncertainties. The following risk factors and those contained in Item 1A — *Risk Factors*, among others, could cause actual results or events to differ materially from the Company's historical experience and those expressed in forward-looking statements made in this document.

Amounts contained in this Report may not always add due to rounding.

Our operations are subject to a number of risks and uncertainties including, but not limited to, those listed below. When considering an investment in our securities, you should carefully read and consider these risks, together with all other information in our Annual Report on Form 10-K for the year ended December 31, 2020 as filed with the SEC and our other filings and submissions to the SEC, which provide much more information and detail on the risks described below. If any of the events described in the following risk factors actually occur, our business, financial condition or operating results, as well as the market price of our securities, could be materially adversely affected. These risks and events include, without limitation:

Epidemics and Pandemic Risks

We are subject to risks associated with epidemics and pandemics, including the COVID-19 pandemic and related impacts. The nature and extent of
ongoing and future impacts of the pandemic are highly uncertain and unpredictable.

Macroeconomic and Industry Risks

- The businesses we operate are highly competitive and many of them are cyclical, which may result in fluctuations in pricing and volume that can adversely affect our business, financial condition and results of operations.
- Changes in raw material and energy availability and prices could have a material adverse effect on our business, results of operations and financial condition..
- We are subject to risks associated with doing business outside of the United States.
- Currency fluctuations may have a negative impact on our business, financial condition and results of operations.
- Restrictions on trade through tariffs, countervailing and anti-dumping duties, quotas and other trade barriers, in the United States and internationally, could materially adversely affect our ability to access certain markets.

Business and Operating Risks

• Our ten largest customers represent approximately 31 percent of our 2020 revenue, and the loss of all or a substantial portion of our revenue from these large customers could have a material adverse effect on our business.

- A material disruption at one of our major manufacturing facilities could prevent us from meeting customer demand, reduce our sales and profitability, increase our cost of production and capital needs, or otherwise materially adversely affect our business, financial condition and results of operation.
- The availability of, and prices for, wood fiber may have a material adverse impact on our business, results of operations and financial condition.
- Our operations require substantial capital.
- We depend on third parties for transportation services and increases in costs and the availability of transportation could materially adversely affect our business.
- Our failure to maintain satisfactory labor relations could have a material adverse effect on our business.
- We are dependent upon attracting and retaining key personnel, the loss of whom could materially adversely affect our business.
- Failure to develop new products or discover new applications for our existing products, or our inability to protect the intellectual property underlying such new products or applications, could have a material negative impact on our business.
- The risk of loss of the Company's intellectual property and sensitive data, or disruption of its manufacturing operations, in each case due to cyberattacks or cybersecurity breaches, could materially adversely impact the Company.

Regulatory Risks

- Our business is subject to extensive environmental laws, regulations and permits that may materially restrict or adversely affect how we conduct business and our financial results.
- The potential longer-term impacts of climate related risks remain uncertain at this time.
- The Company considers and evaluates climate-related risks in three general categories; Regulatory, Transition to a low-carbon economy, and Physical risks related to climate-change.

Financial Risks

- We may need to make significant additional cash contributions to our retirement benefit plans if investment returns on pension assets are lower than expected or interest rates decline, and/or due to changes to regulatory, accounting and actuarial requirements.
- We have debt obligations that could materially adversely affect our business and our ability to meet our obligations.
- The phase-out of the London Inter Bank Office Rate ("LIBOR") as an interest rate benchmark in 2023 may impact our borrowing costs.
- · Challenges in the commercial and credit environments may materially adversely affect our future access to capital.
- We may need additional financing in the future to meet our capital needs or to make acquisitions, and such financing may not be available on favorable terms, if at all, and may be dilutive to existing stockholders.

Company's Common Stock and Certain Corporate Matters Risks

- Your percentage of ownership in the Company may be diluted in the future.
- Certain provisions in our amended and restated certificate of incorporation and bylaws, and of Delaware law, could prevent or delay an acquisition of the Company, which could decrease the price of our common stock.

Forward-looking statements are only as of the date they are made, and the Company undertakes no duty to update its forward-looking statements except as required by law. You are advised, however, to review any further disclosures we have made or may make in our filings and other submissions to the U.S. Securities and Exchange Commission (the "SEC"), including those on Forms 10-Q, 10-K, 8-K and other reports. Details on each of the above risk factors are more specifically described in Item 1A - *Risk Factors*.

Note About Non-GAAP Financial Measures

A "non-GAAP financial measure" is generally defined as a numerical measure of a company's historical or future performance that excludes or includes amounts, or is subject to adjustments, so as to be different from the most directly comparable measure calculated and presented in accordance with U.S. Generally Accepted Accounting Principles ("GAAP"). This Report contains certain non-GAAP financial measures, including Earnings Before Interest, Taxes, Depreciation and Amortization ("EBITDA"), adjusted EBITDA, and adjusted free cash flows. These non-GAAP measures are reconciled to each of their respective most directly comparable GAAP financial measures in Item 7 — *Management's Discussion and Analysis of Financial Condition and Results of Operations*.

We believe these non-GAAP measures provide useful information to our Board of Directors, management and investors regarding certain trends relating to our financial condition and results of operations. Our management uses these non-GAAP



measures to compare our performance to that of prior periods for trend analyses, purposes of determining management incentive compensation and budgeting, forecasting and planning purposes.

We do not consider non-GAAP measures an alternative to financial measures determined in accordance with GAAP. The principal limitation of these non-GAAP financial measures is they may exclude significant expense and income items that are required by GAAP to be recognized in our consolidated financial statements. In addition, they reflect the exercise of management's judgment about which expense and income items are excluded or included in determining these non-GAAP financial measures. In order to compensate for these limitations, reconciliations of the non-GAAP financial measures we use to their most directly comparable GAAP measures are provided. Non-GAAP financial measures should not be relied upon, in whole or part, in evaluating the financial condition, results of operations or future prospects of the Company.

Business

We are a global leader of cellulose-based technologies, which comprise a broad offering of high purity cellulose specialties, a natural polymer commonly used in the production of specialty chemicals and polymers for use in producing LCD displays, filters, fibers, performance additives for pharmaceutical, food and other industrial applications. Starting from a tree and building upon more than 90 years of experience in cellulose chemistry, we provide high quality high-purity cellulose pulp products that make up the essential building blocks for our customers' products while providing exceptional service and value. In addition, we produce lumber, paperboard, newsprint and high-yield pulp for use in consumer products.

Recent developments

Sale of lumber and newsprint assets

On April 12, 2021, we announced the sale of our lumber and newsprint assets located in Ontario and Québec Canada to GreenFirst, for approximately \$214 million, including an assumed \$74 million associated with finished goods, work-in-process and raw materials inventory value, which is subject to fluctuation until closing. The purchase price, to be adjusted at closing by changes in the value of the inventory, will be paid 85% in cash and the remainder in common shares of GreenFirst. In addition, a credit note will be issued to us by GreenFirst in the amount of CDN\$8 million, which may be offset against amounts owed by us to GreenFirst in the future for wood chip purchases, equally over the next 5 years. The closing of the transaction, which is expected to occur in the second half of 2021, but not before July 31, is subject to customary closing conditions, including receipt of regulatory approvals, the transfer of forestry licenses and the approval of the TSX Venture Exchange. See Note 1 —*Basis of Presentation and New Accounting Pronouncements* for additional information.

In 2020, the assets generated approximately \$439 million of revenue and \$55 million of operating income and \$51 million of Adjusted EBITDA.

Final Settlement Reached in Dispute with IESO Relating to Investigation of the Kapuskasing Newsprint Facility

We had previously been engaged in litigation with the Market Assessment and Compliance Division ("MACD") branch of the Independent Electricity System Operator ("IESO") regarding their investigations into the Company's compliance with the published rules that govern the operation of the wholesale electricity market in Ontario, Canada. On April 19, 2021, we and the IESO entered into Minutes of Settlement ("MOS") pursuant to which the parties agreed to fully and finally settle all claims relating to their investigations. As part of the settlement, the Company agreed to a fixed obligation to pay a sum of CAD \$12 million over a period of 5 years comprised of a CAD \$4.5 million up-front payment and a CAD \$7.5 payment to be spread (on a front-weighted basis) over the next 5 anniversaries of the MOS, without interest. In addition to the foregoing, the MOS provides that a "suspended" sum of CAD \$10.4 million would become due and payable in the event the Company fails to comply with any of the terms and conditions of the MOS or commits an event of default, as defined under the applicable market rules, unless such breach or event of default is remedied on a timely basis. See Note 18 — *Commitments and Contingencies* for additional information.

We consider this matter concluded (subject only to the parties' obligations yet to be performed under the MOS).

Coronavirus-Update

Our businesses were significantly impacted by the coronavirus ("COVID-19") pandemic in 2020. While market demand and pricing for certain of our products began to recover towards the end of the year and continued to improve throughout the first quarter of 2021, our operations remain vulnerable to a reversal of these trends or other continuing negative effects caused by COVID-19.

In our operating facilities and work spaces, we continue to maintain protocols previously implemented to reduce the potential spread of COVID-19 and ensure the safety of our employees and continuity of operations.

Due to the financial impacts of COVID-19, we have actively monitored the recoverability of the carrying value of our long-term assets. During the three months ended March 27, 2021 and the year ended December 31, 2020, we did not recognize any impairment charges related to long-lived assets held for use. We will continue to evaluate the recoverability of these and other assets as necessary.

Market Assessment

The market assessment represents our best current estimate of each business in this environment.

High Purity Cellulose

Pricing levels for our commodity products increased during the first quarter and are forecasted to increase further in the second quarter. Prices for cellulose specialties declined slightly and in line with expectations for the full year. Total High Purity Cellulose volumes are expected to remain stable for the full year, however, we expect a more favorable mix towards cellulose specialties. The annual maintenance outage at the Jesup facility in the second quarter is currently ongoing.

Key costs, including energy and commodity chemical prices have increased rapidly during the first quarter and remain difficult to predict. We remain committed to investing in this core business to reduce costs, improve reliability and provide new platforms for growth. The sale of the lumber and newsprint assets discussed below will provide us with a key opportunity to reinvest in our core high purity cellulose business and our BioFuture.

Forest Products

Prices for lumber continue to strengthen to record levels driven by the exceptional demand in the U.S housing market and high levels of repair and remodel activity. Housing starts in March 2021 reached 1.7 million units, seasonally adjusted, while building permits reached 1.8 million. We are focused on ensuring reliable operations and capturing the benefit of high lumber prices through completion of the sale, which is expected in the second half of 2021.

Paperboard

Paperboard prices have increased 5 percent from fourth quarter and are expected to increase further, helping offset increases in raw material cost.

Pulp & Newsprint

High-yield pulp and newsprint markets have experienced price increases during the first quarter and additional price increases are expected in the nearterm. Additionally, we continue to manage production at the newsprint facility to minimize costs and improve sales mix while experiencing early success in the expansion of the EnvirosmartTM food service bag, which targets the quick service restaurant end-market and used for items such as sandwiches and togo orders.

Growing RYAM's BioFuture

Upon the consummation of the sale of the lumber and newsprint assets, we will be focusing on and investing in further leveraging our four high purity cellulose plants as biorefineries. We believe we are in a unique position, starting with natural, renewably sourced feedstock, to capitalize on the global demand for more sustainable products with our leading cellulose specialties offerings as alternatives for petroleum-based incumbents and specialized assets capable of generating green fuels, electricity and other biomaterials. We have executed on several high return projects to enhance the value of these assets, including recent investments in green energy in Tartas, France. Additionally, TemSilk™, a new product being produced in Temiscaming, Quebec, is a critical input in the production of Lyocell, a more sustainable textile. We are also a lead investor in Anomera, Inc. ("Anomera"), a Canadian start-up corporation headquartered in Montreal, Quebec. Anomera manufactures Carboxylated Cellulose Nanocrystals (CNC), a patented, biodegradable product, with uses in the cosmetics industry and various other industrial applications, including concrete, inks and pigments, polymer composites, coatings and adhesives industries. The investment in Anomera provides a new platform for cellulose specialties as we leverage the combined knowledge of cellulose chemistry to develop new markets for cellulose. We will seek further opportunities to leverage our core knowledge of creating the remarkable from the renewable to drive incremental value in our BioFuture.

Critical Accounting Policies and Use of Estimates

The preparation of financial statements requires us to make estimates, assumptions and judgments that affect our assets, liabilities, revenues and expenses and disclosure of contingent assets and liabilities. We base these estimates and assumptions

on historical data and trends, current fact patterns, expectations and other sources of information we believe are reasonable. Actual results may differ from these estimates.

For a full description of our critical accounting policies, see Item 7 — *Management's Discussion and Analysis of Financial Condition and Results of Operations* in our 2020 Annual Report on Form 10-K. For recent accounting pronouncements see Item 1 of Part I, Financial Statements — Note 1 — *Basis of Presentation and New Accounting Pronouncements* for additional information.

Results of Operations

Financial Information	Three Months Ended						
(in millions, except percentages)	 March 27, 2021		March 28, 2020	Change			
Net Sales	\$ 465	\$	410	13%			
Cost of Sales	(381)		(399)				
Gross Margin	 84		11	664%			
Selling, general and administrative expenses	(18)		(20)				
Duties	(7)		(6)				
Foreign exchange gains (losses)	(1)		6				
Other operating income (expense), net	 (4)		(2)				
Operating Income (Loss)	 55		(12)	558%			
Interest expense	(18)		(15)				
Interest income and other, net	(1)		—				
Net periodic pension and OPEB income (expense), excluding service costs	1		—				
Income (Loss) From Continuing Operations Before Income Taxes	 37		(27)	237%			
Income tax benefit (expense)	(64)		2				
Equity in income (loss) of equity method investment	\$ —	\$	—				
Income (Loss) from Continuing Operations	\$ (27)	\$	(25)	(8)%			
Income (loss) from discontinued operations, net of taxes	_		1				
Net Income (Loss)	\$ (27)	\$	(24)				
Gross Margin %	18 %		3 %				
Operating Margin %	12 %		(3)%				
Effective Tax Rate %	(172)%		(6)%				

Net sales by segment were as follows:

		Ended		
Net sales (in millions)	Mar	rch 27, 2021		March 28, 2020
High Purity Cellulose	\$	250	\$	250
Forest Products		147		82
Paperboard		48		50
Pulp & Newsprint		39		47
Eliminations		(19)		(19)
Total net sales	\$	465	\$	410

Net sales increased by \$55 million during the three months ended March 27, 2021 when compared to the three months ended March 28, 2020 primarily driven by higher lumber and High Purity Cellulose commodity prices along with cellulose specialties sales volumes. The increases were partially offset by lower High Purity Cellulose commodity product sales volumes. For further discussion, see Operating Results by Segment.

Operating income (loss) by segment was as follows:

	Three Months Ended							
Operating income (loss) (in millions)	March 27, 2021	March 28, 2020						
High Purity Cellulose	\$6	\$ (5)						
Forest Products	61	(1)						
Paperboard	6	5						
Pulp & Newsprint	(6)	(6)						
Corporate	(12)	(5)						
Total operating income (loss)	\$ 55	\$ (12)						

The operating results for the three month period ended March 27, 2021 improved by \$67 million, to operating income of \$55 million when compared to the same prior year period primarily due to strong lumber prices, lower operating costs, higher High Purity Cellulose commodity prices and increased cellulose specialties sales volumes.

Non-operating Expenses

Interest expense increased \$3 million to \$18 million for the three months ended March 27, 2021 when compared to the same prior year period. The increase was principally driven by the higher interest rate and additional amortization of debt issuance costs related to the December 2020 refinancing of certain debt instruments. See Note 7 — **Debt and Finance Leases** for further information.

Income Tax Benefit (Expense)

The effective tax rate expense for the first quarter of 2021 was 172 percent compared to a benefit rate of 6 percent in the same period of 2020. The 2021 effective tax rate differs from the statutory rate of 21 percent primarily due to Global Intangible Low Taxed Income ("GILTI") on foreign earnings, disallowed interest deductions in the U.S., and different statutory tax rates of foreign operations. The application of GILTI effectively results in double book taxation of the majority of the Company's high Canadian earnings. The Company currently expects minimal cash taxes to be paid in 2021 or 2022 as a result of 2021 earnings. See Note 16 — *Income Taxes* for additional information.

Discontinued Operations

The Company has presented the operating results for its Matane operations that was sold in November 2019 as discontinued operations. The Company had an after tax benefit of \$1 million as a result of the working capital adjustments as required by the sale agreement for the three months ended March 28, 2020.

Operating Results by Segment

High Purity Cellulose

	Three Months Ended							
(in millions)	Mar	March 27, 2021 March 28, 202						
Net Sales	\$	250	\$	250				
Operating income (loss)	\$	6	\$	(5)				
Average Sales Prices (\$ per metric ton):								
High Purity Cellulose	\$	1,046		963				
Sales Volumes (thousands of metric tons):								
High Purity Cellulose		217		235				

Changes in High Purity Cellulose net sales are as follows:

Three Months Ended		Changes Attributable to:								
Net Sales (in millions)	Marc	March 28, 2020 Price			Volume/Mix/Other			March 27, 2021		
Cellulose Specialties	\$	160	\$	(2)	\$	10	\$	168		
Commodity Products		67		8		(16)		59		
Other sales (a)		23		_		—		23		
Total Net Sales	\$	250	\$	6	\$	(6)	\$	250		
(a) Other sales consist of electricity	lignin and other h	v-products to third-r	arties		-					

(a) Other sales consist of electricity, lignin and other by-products to third-parties.

Total net sales for the three months ended March 27, 2021 were comparable to the same prior year period at \$250 million. Cellulose specialties volumes improved by 6 percent while sales prices declined slightly by 1 percent during the quarter ended March 27, 2021. Commodity product sales prices increased 15 percent during the first quarter of 2021, however sales volumes decreased by 23 percent.

Changes in High Purity Cellulose operating income are as follows

Three Months Ended	Gross Margin Changes Attributable to (a):											
(in millions)	Marc	h 28, 2020	Sales Sales Price Volume/Mix/Other Cost						SG&	A and other	Mare	ch 27, 2021
Operating income (loss)	\$	(5)	\$	6	\$	(1)	\$	5	\$	1	\$	6
Operating margin %		(2.0)%		2.4 %		(0.4)%		2.0 %		0.4 %		2.4 %

(a) Sales Volume computed based on contribution margin.

Operating results improved by \$11 million during the three months ended March 27, 2021 to operating income of \$6 million when compared to the same prior year period. Sales prices for the segment increased 9 percent during the current three-month period driven by higher commodity prices and increased cellulose specialties sales when compared to the same prior year period. Compared to the prior year, sales volumes for the segment were impacted by shipping constraints and declined 8 percent during the current three-month period driven by lower commodities volumes, partially offset by higher cellulose specialties sales volumes driven by improved demand. Additionally, better productivity and the impact of shutdowns improved operational costs.

Forest Products

		Ended		
(in millions)		March 27, 2021		March 28, 2020
Net Sales	\$	147	\$	82
Operating income (loss)	\$	61	\$	(1)
Average Sales Prices (\$ per thousand board feet):				
Lumber	\$	888	\$	407
Sales Volumes (millions of board feet):				
Lumber		144		149

Changes in Forest Products net sales are as follows:

Three Months Ended		Changes Attributable to:						
Net Sales (in millions)	March 28, 2020	Price	Volume/Mix/Other	March 27, 2021				
Lumber	\$ 61	\$ 70	\$ (3)	\$ 128				
Other sales (a)	21	—	(2)	19				
Total Net Sales	\$ 82	\$ 70	\$ (5)	\$ 147				

(a) Other sales consist of sales of logs, wood chips, and other by-products to other segments and third-parties

Total net sales for the three months ended March 27, 2021 increased \$65 million, or 78 percent when compared to the same prior year period ended 2020. The improvement is due to a 118 percent increase in lumber prices. Sales volumes were slightly lower than in the prior year resulting from timing of shipments and logistic constraints.

Changes in Forest Products operating income are as follows:

Three Months Ended

Gross Margin Changes Attributable to (a)

(in millions)	Marc	h 28, 2020	Sales Price	V	Sales /olume/Mix/Other	Cost	SG&	A and other	Ma	rch 27, 2021
Operating income (loss)	\$	(1)	\$ 70	\$	(5)	\$ (3)	\$		\$	61
Operating margin %		(1.2)%	 46.6 %		(1.9)%	 (2.0)%		— %		41.5 %

(a) Sales Volume computed based on contribution margin.

Operating results improved by \$62 million during the three months ended March 27, 2021 to an operating income of \$61 million. The favorable results were driven by higher lumber prices partly offset by lower sales volumes discussed above and the impact of higher residual stumpage costs. The Company deposited \$7 million and \$6 million of softwood lumber duties in the quarter ended March 27, 2021 and 2020, respectively. While the Company benefits from the December 2020 rate reduction from 20% to 9%, the increase in lumber prices resulted in an increase to the total value of the duties paid.

Paperboard

	 Three Months Ended						
(in millions)	March 27, 2021		March 28, 2020				
Net Sales	\$ 48	\$	50				
Operating income	\$ 6	\$	5				
Average Sales Prices (\$ per metric tons):							
Paperboard	\$ 1,111	\$	1,107				
Sales Volumes (in thousands of metric tons):							
Paperboard	43		46				

Changes in Paperboard net sales are as follows:

Three Months Ended	Changes Attributable to:						
Net Sales (in millions)	March 28, 2020		Price	Volume/Mix		March 27, 2021	
Paperboard	\$	0 \$		\$	(2)	\$ 48	8

Sales for the three months ended March 27, 2021 decreased \$2 million as sales volumes declined 7 percent driven by timing of sales compared to the prior year.

Three Months Ended			 Gross Margin Changes Attributable to (a):								
(in millions)	Mare	ch 28, 2020	Sales Price	Sale	s Volume/Mix		Cost	SG8	&A and other	Ma	rch 27, 2021
Operating income (loss)	\$	5	\$ 	\$	(1)	\$	2	\$	_	\$	6
Operating margin %		10.0 %	— %		(1.7)%		4.2 %		— %		12.5 %

(a) Computed based on contribution margin.

Operating results improved by \$1 million during the three months ended March 27, 2021 to operating income of \$6 million. The improvement was primarily due to lower operational costs.

Pulp and Newsprint

Three Months Ended							
March	March 27, 2021						
\$	39 \$	47					
\$	(6) \$	(6)					
\$	474 \$	463					
\$	468 \$	417					
	44	52					
	25	40					
	\$ \$ \$	March 27, 2021 \$ 39 \$ \$ (6) \$ \$ 474 \$ \$ 468 \$ 44 44 44					

(a) Average sales prices and volumes for external sales only. For the three month period ended March 27, 2021 and March 28, 2020, the Pulp & Newsprint segment sold approximately 17,000 metric tons and 16,000 metric tons of high-yield pulp for \$7 million and \$6 million, respectively, to the Paperboard segment.

Changes in Pulp & Newsprint net sales are as follows:

Three Months Ended			ble to:			
Net Sales (in millions)	Marc	h 28, 2020	Price		Volume/Mix	March 27, 2021
Pulp	\$	30 \$	1	\$	(4)	\$ 28
Newsprint		17	1		(6)	11
Total Net Sales	\$	47 \$	2	\$	(10)	\$ 39

Total sales for the three months ended March 27, 2021 were down \$8 million or 16 percent when compared to the same three months ended March 28, 2020. High-yield pulp and newsprint sales prices were up 2 percent and 12 percent, respectively, while high-yield pulp and newsprint sales volumes decreased 15 percent and 38 percent, respectively. Declines in newsprint volumes were due to continued weak demand stemming from the COVID-19 pandemic and management's decision to produce on only one of its newsprint two operating lines.

Changes in Pulp & Newsprint operating income are as follows:

Three Months Ended			 Gross Margin Changes Attributable to (a):								
(in millions)	Mar	ch 28, 2020	Sales Price	Sale	es Volume/Mix		Cost	SG8	A and other	Ma	rch 27, 2021
Operating income (loss)	\$	(6)	\$ 2	\$	(4)	\$	2	\$	_	\$	(6)
Operating margin %		(12.8)%	 4.6 %		(12.3)%		5.1 %		— %		(15.4)%

(a) Sales Volume computed based on contribution margin.

Operating results for the three months ended March 27, 2021 remained flat when compared to the same prior year period. Lower newsprint and high-yield pulp sales volumes were offset by improvements in high-yield pulp and newsprint sales prices. Declines in high-yield pulp volumes were impacted by the timing of shipments and logistics constraints.

Corporate

		Three Months	Ended
Operating Income (Loss) (in millions)	Ma	rch 27, 2021	March 28, 2020
Operating loss	\$	(12) \$	(5)

The operating loss for the three months ended March 27, 2021 increased \$7 million when compared to the same prior year period primarily due to unfavorable foreign exchange impacts and an increased reserve associated with the dispute settlement discussed above.

Liquidity and Capital Resources

Cash flows from operations, primarily driven by operating results, have historically been our primary source of liquidity and capital resources. However, our operating cash flows have been volatile in recent years due to decreases in market prices for our commodity products as well as the impact on demand driven by the COVID-19 pandemic. In response, we maintain a key focus on cash, managing working capital closely and optimizing the timing and level of our capital expenditures.

As of March 27, 2021, we are in compliance with all financial and other customary covenants. We believe our future cash flows from operations and availability under our ABL Credit Facility, as well as our ability to access the capital markets, if necessary or desirable, will be adequate to fund our operations and anticipated long-term funding requirements, including capital expenditures, defined benefit plan contributions, and repayment of debt maturities.

The non-guarantor subsidiaries had assets of \$719 million, year-to-date revenue of \$54 million, covenant EBITDA for the last twelve months is a \$33 million loss and liabilities of \$285 million as of March 27, 2021.

On September 6, 2019, our Board of Directors suspended our quarterly common stock dividend. No dividends have been declared since. The declaration and payment of future common stock dividends, if any, will be at the discretion of the Board of Directors and will be dependent upon our financial condition, results of operations, capital requirements and other factors the Board of Directors deem relevant. In addition, our debt facilities place limitations on the declaration and payment of future dividends.

On January 29, 2018, our Board of Directors authorized a \$100 million common stock share buyback program. For the three months ended March 27, 2021 and March 28, 2020, we did not repurchase any common shares under this buyback program. We do not expect to utilize any further authorization in the near future.

A summary of liquidity and capital resources is shown below (in millions of dollars):

	Μ	larch 27, 2021	December 31, 2020
Cash and cash equivalents (a)	\$	107	\$ 94
Availability under the ABL Credit Facility (b)		142	102
Total debt (c)		1,083	1,084
Stockholders' equity		659	695
Total capitalization (total debt plus equity)	\$	1,742	\$ 1,779
Debt to capital ratio		62 %	61 %

(a) Cash and cash equivalents consisted of cash, money market deposits and time deposits with original maturities of 90 days or less.

(b) Amounts available under the ABL Credit Facility fluctuate based on eligible accounts receivable and inventory levels. At March 27, 2021, we had \$186 million of gross availability and net available borrowings of \$142 million after taking into account standby letters of credit of approximately \$44 million. In addition to the availability under the ABL Credit Facility, we have \$19 million available under an accounts receivable factoring line of credit in France.

(c) See Note 7 — *Debt and Finance Leases* of our consolidated financial statements for additional information.

Cash Flows (in millions of dollars)

The following table summarizes our cash flows from operating, investing and financing activities for the three months ended:

<u>Cash Flows Provided by (Used for):</u>	March 27, 2021				
Operating activities	\$	38	\$	(13)	
Investing activities	\$	(21)	\$	(13)	
Financing activities	\$	(3)	\$	5	

Cash flows provided by operating activities improved \$51 million during the three months ended March 27, 2021 to \$38 million when compared to the same prior year period due to favorable operating results driven from higher lumber and High Purity Cellulose prices and lower operating costs partially offset by seasonal working capital increases. Higher non-cash expenses primarily related to deferred tax expense were partly offset by unfavorable changes to working capital. The three

months ended March 28, 2020 included a \$20 million increase to the U.S. income tax receivable from the passage of the CARES Act in March of 2020.

Cash flows used for investing activities increased \$8 million during the three months ended March 27, 2021 when compared to the same prior year period from increased capital spending. The increase also includes a \$1 million non-voting investment in Anomera, Inc.

Cash flows from financing activities changed by \$8 million during the three months ended March 27, 2021 to cash used for investing activities of \$3 million when compared to the same prior year period. The decline is primarily from a decrease in borrowings, net of payments, during the three months ended March 27, 2021, partially offset by cash paid for common stock repurchased in lieu of income taxes from the vesting of incentive stock grants which was \$1 million higher during the current period. See Note 7 — *Debt and Finance Leases* and Note 12 — *Stockholders' Equity*, to our consolidated financial statements for additional information.

Performance and Liquidity Indicators

The discussion below is presented to enhance the reader's understanding of our operating performance, liquidity, ability to generate cash and satisfy rating agency and creditor requirements. This information includes the following measures of financial results: EBITDA, adjusted EBITDA and adjusted free cash flows. These measures are not defined by U.S. Generally Accepted Accounting Principles ("GAAP") and the discussion of EBITDA, adjusted EBITDA and adjusted free cash flows is not intended to conflict with or change any of the GAAP disclosures described above. Our management uses these non-GAAP measures to compare our performance to that of prior periods for trend analyses, purposes of determining management incentive compensation and budgeting, forecasting and planning purposes. Our management considers these measures, in addition to operating income, to be important to estimate the enterprise and stockholder values of the Company, and for making strategic and operating decisions. In addition, analysts, investors and creditors use these measures when analyzing our operating performance, financial condition and cash generating ability. Our management uses EBITDA and adjusted EBITDA as performance measures and adjusted free cash flows as a liquidity measure. See "Note about Non-GAAP Financial Measures" on page 21 for limitations associated with non-GAAP measures.

EBITDA is defined by SEC rules as earnings before interest, taxes, depreciation and amortization. EBITDA is not necessarily indicative of results that may be generated in future periods.

Below is a reconciliation of Income (Loss) from Continuing Operations to EBITDA by segment (in millions of dollars):

Three Months Ended	Forest Products		Paperboard		Pulp & Newsprint (a)			High Purity Cellulose	Corporate & Other		Total	
March 27, 2021					Newsprint		Pulp					
Income (loss) from continuing operations	\$	61	\$	6\$	5 (5)	\$	_	\$ 7	\$	(96)	\$	(27)
Depreciation and amortization		2		4	1		_	28		1		36
Interest expense, net		_	-	_	_		—	_		18		18
Income tax expense (benefit)		—		-	—		—			64		64
EBITDA	\$	63	\$ 1	0 \$	6 (4)	\$		\$ 35	\$	(13)	\$	91
March 28, 2020												
Income (loss) from continuing operations	\$	(1)	\$	5\$	6 (4)	\$	(1)	\$ (5)	\$	(19)	\$	(25)
Depreciation and amortization		2		4	—		1	30				38
Interest expense, net		—	_	-	—		—			15		15
Income tax expense (benefit)		_								(2)		(2)
EBITDA	\$	1	\$	9 \$	6 (4)	\$		\$ 26	\$	(5)	\$	27

(a) Due to the announced sale of our lumber and newsprint assets, we are providing a break out of EBITDA for the Pulp & Newsprint segment.

EBITDA for the three months ended March 27, 2021 improved by \$64 million when compared to the quarter ended March 28, 2020, primarily from favorable operating results driven by higher lumber and High Purity Cellulose prices and lower costs. For the full discussion of changes to operating income, see *Management's Discussion of Results of Operations*.

Adjusted free cash flows is defined as cash provided by operating activities of continuing operations adjusted for capital expenditures, net of proceeds from sale of assets, excluding strategic capital expenditures. Adjusted free cash flows, as defined by the Company, is a non-GAAP measure of cash generated during a period which is available for debt reduction, strategic capital expenditures and acquisitions and repurchase of the Company's common stock. Adjusted free cash flows is not necessarily indicative of the adjusted free cash flows that may be generated in future periods.

Below is a reconciliation of cash flows from operations to adjusted free cash flows for the respective periods (in millions of dollars):

		Three Months Ended				
Cash Flows from Operations to Adjusted Free Cash Flows Reconciliation	March 27, 2021			March 28, 2020		
Cash provided by (used for) operating activities - continuing operations	\$	38	\$	(13)		
Capital expenditures (a)		(17)		(10)		
Adjusted Free Cash Flows	\$	21	\$	(23)		

(a) Capital expenditures exclude strategic capital expenditures which are deemed discretionary by management. Strategic expenditures for the first three months of 2021 were approximately \$2 million. Strategic capital expenditures for the same period of 2020 were approximately \$2 million.

Adjusted free cash flows improved primarily due to favorable operating results primarily from higher lumber and High Purity Cellulose prices partly offset by higher capital expenditure requirements. For the full discussion of operating cash flows, see *Management's Discussion and Analysis of Cash Flows*.

Contractual Financial Obligations and Off-Balance Sheet Arrangements

We have no material changes outside the ordinary course of business to the Contractual Financial Obligations table as presented in Item 7 — *Management's Discussion and Analysis of Financial Condition and Results of Operations* in our 2020 Annual Report on Form 10-K.

See Note 18 — Commitments and Contingencies for details on our letters of credit and surety bonds as of March 27, 2021.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Market and Other Economic Risks

We are exposed to various market risks, primarily changes in interest rates, currency and commodity prices. Our objective is to minimize the economic impact of these market risks. We may use derivatives in accordance with policies and procedures approved by the Audit Committee of our Board of Directors. Derivatives are managed by a senior executive committee whose responsibilities include initiating, managing and monitoring resulting exposures. See Note 9 — *Derivative Instruments* for additional information.

We manage our foreign currency exposures by balancing certain assets and liabilities denominated in foreign currencies. We may also use foreign currency forward contracts to manage these exposures. The principal objective of such contracts is to minimize the potential volatility and financial impact of changes in foreign currency exchange rates. We do not utilize financial instruments for trading or other speculative purposes.

The prices, sales volumes and margins of the commodity products of our High Purity Cellulose segment and all the products of the Forest Products and Pulp & Newsprint segments have historically been cyclically affected by economic and market shifts, fluctuations in capacity, and changes in foreign currency exchange rates. In general, these products are commodities that are widely available from other producers; because these products have few distinguishing qualities from producer to producer, competition is based primarily on price, which is determined by supply relative to demand. The overall levels of demand for the products we manufacture, and consequently our sales and profitability, reflect fluctuations in end user demand. Our cellulose specialties product prices are impacted by market supply and demand, raw material and processing costs, changes in global currencies and other factors. While these prices are not directly correlated to commodity dissolving wood pulp and paper pulp prices, changes in commodity dissolving wood pulp and paper pulp prices may impact competitors' actions

which can lead to an impact in prices for cellulose specialties products. In addition, approximately half of our cellulose specialties contracted volumes are under multi-year contracts that expire between 2021 and 2023.

As of March 27, 2021, we had \$4 million of variable rate debt which is subject to interest rate risk. At this borrowing level, a hypothetical onepercentage point increase/decrease in interest rates would result in an immaterial increase/decrease in interest payments and expense over a 12-month period.

The fair market value of our long-term fixed interest rate debt is also subject to interest rate risk. However, we intend to hold most of our debt until maturity. The estimated fair value of our fixed-rate debt at March 27, 2021 was \$1,104 million compared to the \$1,076 million carrying value of principal amount. We use quoted market prices to estimate the fair value of our fixed-rate debt. Generally, the fair market value of fixed-rate debt will increase as interest rates fall and decrease as interest rates rise.

We may periodically enter into commodity forward contracts to fix some of our energy costs that are subject to price volatility caused by weather, supply conditions, political and economic variables and other unpredictable factors. Such forward contracts partially mitigate the risk of changes to our gross margins resulting from an increase or decrease in these costs. Forward contracts which are derivative instruments are reported in the consolidated balance sheets at their fair values, unless they qualify for the normal purchase normal sale ("NPNS") exception and such exception has been elected. If the NPNS exception is elected, the fair values of such contracts are not recognized on the balance sheet.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

Our management is responsible for establishing and maintaining adequate disclosure controls and procedures. Disclosure controls and procedures (as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934 (the "Exchange Act")), are designed with the objective of ensuring that information required to be disclosed in reports filed under the Exchange Act, such as this quarterly report on Form 10-Q, is (1) recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and (2) accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure.

Because of the inherent limitations in all control systems, no control evaluation can provide absolute assurance that all control exceptions and instances of fraud have been prevented or detected on a timely basis. Even systems determined to be effective can provide only reasonable assurance their objectives are achieved.

Based on an evaluation of our disclosure controls and procedures as of the end of the period covered by this quarterly report on Form 10-Q, our management, including the Chief Executive Officer and Chief Financial Officer, concluded the design and operation of the disclosure controls and procedures were effective as of March 27, 2021.

During the quarter ended March 27, 2021, based upon the evaluation required by paragraph (d) of SEC Rule 13a-15, there were no changes in our internal control over financial reporting that would materially affect or are reasonably likely to materially affect our internal control over financial reporting.
Part II. Other Information

Item 1. Legal Proceedings

The Company is engaged in various legal and regulatory actions and proceedings, and has been named as a defendant in various lawsuits and claims arising in the ordinary course of its business. While the Company has procured reasonable and customary insurance covering risks normally occurring in connection with its businesses, the Company has in certain cases retained some risk through the operation of self-insurance, primarily in the areas of workers' compensation, property insurance, business interruption and general liability. While there can be no assurance, the ultimate outcome of these actions, either individually or in the aggregate, is not expected to have a material adverse effect on the Company's financial position, results of operations or cash flows, except as may be noted below.

Final Settlement Reached in Dispute with IESO Relating to Investigation of the Kapuskasing Newsprint Facility

From the period from 2014 to early 2021, the Market Assessment and Compliance Division ("MACD") branch of the Independent Electricity System Operator ("IESO"), the governmental agency responsible for operating the wholesale electricity market and directing the operation of the bulk electrical system in the province of Ontario, Canada, had been engaged in reviewing the Company's compliance with the published rules that govern the operation of the wholesale electricity market in Ontario, Canada. The inquiry was focused primarily on payments made by IESO to the Company between 2010 and 2019 under market rules in connection with multiple planned, extended and unplanned forced outages that caused extensive downtime, in full or in part, of the Company's Kapuskasing, Ontario newsprint facility.

In May 2020, MACD finalized two of its four investigations into the Company's electricity management practices at its Kapuskasing newsprint facility and issued orders asserting penalties of CAD \$25 million. These orders called for the Company to pay penalties of CAD \$3 million immediately and CAD \$12 million over a 10-year period, with the remaining CAD \$10 million to be deferred and ultimately forgiven assuming the Company otherwise complied with the orders' remaining terms. The Company, which maintained it had complied in all material respects with the published rules, vigorously contested IESO's orders, including through the filing of judicial review proceedings with the divisional Court (Superior Court of Justice) of Ontario seeking invalidation of the orders. At the time these orders were issued, the remaining two investigations remained open, subjecting the Company to the risk that MACD may in the future issue additional orders upon finalization of these additional investigations.

On April 19, 2021, the Company and IESO entered into Minutes of Settlement ("MOS") pursuant to which the parties agreed to fully and finally settle all claims relating to all four of the investigations (whether completed or not) and related orders, the judicial review proceedings and underlying disputes. As part of the settlement, the Company agreed to a fixed obligation to pay a sum of CAD \$12 million over a period of 5 years comprised of a CAD \$4.5 million up-front payment and a CAD \$7.5 payment to be spread (on a front-weighted basis) over the next 5 anniversaries of the MOS, without interest. In addition to the foregoing, the MOS provides that a "suspended" sum of CAD \$10.4 million would become due and payable in the event the Company fails to comply with any of the terms and conditions of the MOS or commits an event of default, as defined under the applicable market rules, unless such breach or event of default is remedied on a timely basis. This contingent "suspended" sum decreases annually as the scheduled fixed, or non-contingent, payments are made under the MOS. Assuming no uncured event of default or breach occurs during the repayment period, upon full payment of the CAD \$12 million, the entire "suspended" sum shall be extinguished and RYAM shall be released from any payment obligation with respect thereto.

Given the parties' finalization of and entry into the MOS, the Company considers this matter concluded (subject only to the parties' obligations yet to be performed under the MOS).

Item 1A. Risk Factors

There have been no material changes to the risk factors previously disclosed in Part I, Item 1A, of our 2020 Annual Report on Form 10-K during the period covered by this report.



Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Issuer Purchases of Equity Securities

The following table provides information regarding our purchases of Rayonier Advanced Materials common stock during the quarter ended March 27, 2021:

<u>Period</u>	Total Number of Shares Purchased (b)	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	of	pproximate Dollar Value f Shares that May Yet Be ırchased Under the Plans or Programs (a)
January 1 to January 30	_	\$ _		\$	60,294,000
January 31 to February 27	3,828	\$ 12.04	—	\$	60,294,000
February 28 to March 27	127,801	\$ 10.70	—	\$	60,294,000
Total	131,629				

(a) As of March 27, 2021, approximately \$60 million of share repurchase authorization remains under the authorization declared by the Board of Directors on January 29, 2018.

(b) Repurchased to satisfy the tax withholding requirements related to the issuance of stock under the Rayonier Advanced Materials Incentive Stock Plan.

Item 6. Exhibits

<u>3.1</u>	Amended and Restated Certificate of Incorporation of Rayonier Advanced Materials Inc.	Incorporated herein by reference to Exhibit 3.1 to the Registrant's Form 8-K filed on June 30, 2014
<u>3.2</u>	Certificate of Designations of 8.00% Series A Mandatory Convertible Preferred Stock of Rayonier Advanced Materials Inc., filed with the Secretary of State of the State of Delaware and effective August 10, 2016	Incorporated herein by reference to Exhibit 3.1 to the Registrant's Form 8-K filed on August 10, 2016
<u>3.3</u>	Amended and Restated Bylaws of Rayonier Advanced Materials Inc	Incorporated herein by reference to Exhibit 3.2 to the Registrant's Form 8-K filed on June 30, 2014
<u>10.1</u>	Asset Purchase Agreement by and between 9437-6001 Quebec Inc., as purchaser and GreenFirst Forest Products Inc., as Purchaser guarantor, and Rayonier A.M. Canada G.P. and Rayonier A.M. Canada Industries Inc., collectively the Seller, dated as of April 10, 2021 ¹	Filed herewith
<u>31.1</u>	Chief Executive Officer's Certification Pursuant to Rule 13a-14(a)/15d-14(a) and pursuant to Section 302 of the Sarbanes-Oxley Act of 2002	Filed herewith
<u>31.2</u>	Chief Financial Officer's Certification Pursuant to Rule 13a-14(a)/15d-14(a) and pursuant to Section 302 of the Sarbanes-Oxley Act of 2002	Filed herewith
<u>32</u>	Certification of Periodic Financial Reports Under Section 906 of the Sarbanes-Oxley Act of 2002	Furnished herewith
101	The following financial information from our Quarterly Report on Form 10-Q for the three months ended March 27, 2021 formatted in Extensible Business Reporting Language ("XBRL"), includes: (i) the Condensed Consolidated Statements of Income (Loss) and Comprehensive Income (Loss) for the Three Month Ended March 27, 2021 and March 28, 2020; (ii) the Condensed Consolidated Balance Sheets as of March 27, 2021 and December 31, 2020; (iii) the Condensed Consolidated Statements of Cash Flows for the Three Months Ended March 27, 2021 and March 28, 2020; and (iv) the Notes to Condensed Consolidated Financial Statements	Filed herewith
104	Cover Page Interactive Data File - formatted as Inline XBRL and contained in Exhibit 101	

¹ Certain confidential portions of this exhibit were omitted by means of marking such portions with asterisks because the identified confidential portions (i) are not material and (ii) would be competitively harmful if publicly disclosed.

37

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Rayonier Advanced Materials Inc. (Registrant)

By: /s/ MARCUS J. MOELTNER

Marcus J. Moeltner Chief Financial Officer and Senior Vice President, Finance (Duly Authorized Officer and Principal Financial Officer)

Date: May 6, 2021

38

Confidential Material Omitted – To be filed separately with the Securities and Exchange Commission upon request. Double asterisks denote omissions.

9437-6001 QUEBEC INC.

as the Purchaser

and

GREENFIRST FOREST PRODUCTS INC.

as the Parent Guarantor

and

RAYONIER A.M. CANADA G.P.

and

RAYONIER A.M. CANADA INDUSTRIES INC.

collectively as the Seller

ASSET PURCHASE AGREEMENT

April 10, 2021

TABLE OF CONTENTS

Article 1 INTERPRETATION 1 Section 1.1 Defined Terms. 1 Section 1.2 Gender and Number. 18 Section 1.3 Headings, etc. 18 Section 1.4 Currency. 18 Section 1.5 Certain Phrases, etc. 18 Section 1.6 Accounting Terms. 18 Section 1.7 Incorporation of Disclosure Letter, schedules and Exhibits. 18 Section 1.8 References to Persons and Agreements. 18 Section 1.9 Statutes. 18 Section 1.10 NonBusiness Days. 19 Section 1.11 Knowledge. 19 Section 1.12 Made Available. 19 Article 2 PURCHASE AND SALE 19 Section 2.1 Purchase and Sale. 19 Section 2.2 Excluded Assets. 21 Section 2.3 Purchase Price. 22 Section 2.4 Allocation of Purchase Price. 22 Section 2.5 Payment on Closing and after Closing. 23 Section 2.6 Estimated Inventory Amount. 23 Section 2.7 Preparation of Closing Statement. 23 Section 2.8 Settlement of Objections. 23 Section 2.9 Payment of Adjustment to Purchase Price. 25 Section 2.10 No Effect on Other Rights. 26 Section 2.11 Share Exchange. 26 Section 2.12 Tax Elections 26 Section 2.13 Real Estate and Other Adjustments. 27 Section 2.14 Assumption of Obligations and Liabilities. 28 Section 2.15 Excluded Obligations and Liabilities. 28 Section 2.16 NonAssignable Contracts, Commitments and Permits. 28 Section 2.17 Replacement of Security Bonds. 29 Article 3 REPRESENTATIONS AND WARRANTIES 30 Section 3.1 Representations and Warranties of the Seller. 30 Section 3.2 Disclaimer of Other Representations and Warranties 48 Article 4 REPRESENTATIONS AND WARRANTIES OF THE PURCHASER 49 Section 4.1 Representations and Warranties of the Purchaser. 49 Section 4.2 Disclaimer of Other Representations and Warranties. 51 Article 5 REPRESENTATIONS AND WARRANTIES OF THE PARENT GUARANTOR 51 Section 5.1 Representations and Warranties of the Parent Guarantor. 51

Page

Section 5.2 Disclaimer of Other Representations and Warranties. 56 Article 6 COVENANTS 56 Section 6.1 Conduct of Operations Prior to Closing. 56 Section 6.2 Actions to Satisfy Closing Conditions. 58 Section 6.3 Confidentiality. 58 Section 6.4 Request for Consents. 59 Section 6.5 Authorizations and Regulatory Approvals. 59 Section 6.6 Financing. 61 Section 6.7 Access. 62 Section 6.8 Contact with Customers, Suppliers, Employees and Other Operations Relations. 63 Section 6.9 Notice of Untrue Representation or Warranty. 63 Section 6.10 Transfer Taxes. 63 Section 6.11 Cooperation on Tax Matters. 64 Section 6.12 TSXV Required Information. 64 Section 6.13 Treatment of Purchaser Shares and Parent Shares 64 Section 6.14 Commitment to Financing 65 Article 7 CONDITIONS OF CLOSING 65 Section 7.1 Conditions for the Benefit of the Purchaser. 65 Section 7.2 Conditions for the Benefit of the Seller. 68 Article 8 CLOSING 69 Section 8.1 Date, Time and Place of Closing. 69 Section 8.2 Risk of Loss. 70 Section 8.3 Single Transaction. 70 Article 9 TERMINATION 70 Section 9.1 Termination Rights. 70 Section 9.2 Effect of Termination. 71 Article 10 INDEMNIFICATION 72 Section 10.1 Survival. 72 Section 10.2 Indemnification in Favour of the Purchaser. 72 Section 10.3 Indemnification in Favour of the Seller. 73 Section 10.4 Limitations. 74 Section 10.5 Notice of Claim. 74 Section 10.6 Direct Claims. 75 Section 10.7 Procedure for Third Party Claims. 75 Section 10.8 Manner of Payment. 77 Section 10.9 Exclusion of Other Remedies. 77 Section 10.10 Duty to Mitigate. 78 Section 10.11 One Recovery. 78 Section 10.12 Adjustments to Purchase Price. 78 Section 10.13 Guarantee by Parent Guarantor 78 Article 11 POSTCLOSING COVENANTS AND OTHER MATTERS 79 Section 11.1 Employees. 79

Section 11.2 Pension Plans and Other Benefit Plans. 80 Section 11.3 Pension Plan Arrangements. 81 Section 11.4 Access to Purchaser Books and Records. 82 Section 11.5 Access to Seller Books and Records. 82 Section 11.6 Duties. 83 Section 11.7 Insurance Policies. 83 Section 11.8 Further Assurances. 83 Article 12 MISCELLANEOUS 84 Section 12.1 Notices. 84 Section 12.2 Time of the Essence. 85 Section 12.3 Announcements and Public Communications. 85 Section 12.4 Third Party Beneficiaries. 86 Section 12.5 Expenses. 86 Section 12.6 Amendments. 86 Section 12.7 Waiver. 87 Section 12.8 NonMerger. 87 Section 12.9 Entire Agreement. 87 Section 12.10 Successors and Assigns. 87 Section 12.11 Severability. 87 Section 12.12 Governing Law; Waiver of Jury Trial. 88 Section 12.13 Counterparts. 88 Section 12.14 Legal Representation. 88 Exhibit A Chip Purchase Agreement A-1 Exhibit B Non-Competition Agreement B-1 Exhibit C Services Agreement C-1 Exhibit D Transition Services Agreement D-1 Exhibit E GST and QST Certificate E-1 Exhibit F Pension Plan Assignment and Assumption Agreement F-1 Exhibit G Chip Purchase Set-Off Note G-1 Exhibit H Support in Forestry Matters H-1

ASSET PURCHASE AGREEMENT

Asset Purchase Agreement dated April 10, 2021 between **9437-6001 QUEBEC INC.**, a corporation incorporated under the laws of the Province of Quebec (the "**Purchaser**"), **GREENFIRST FOREST PRODUCTS INC.** (the "**Parent Guarantor**"), a corporation incorporated under the laws of British Columbia, **RAYONIER A.M. CANADA G.P.**, a partnership duly formed under the laws of the Province of Québec and **RAYONIER A.M. CANADA INDUSTRIES INC.**, a corporation incorporated under the laws of Canada (collectively the "**Seller**").

RECITALS:

- A. The Seller is the registered and beneficial owner of the Purchased Assets.
- B. The Purchaser wishes to purchase the Purchased Assets from the Seller and to undertake and assume performance of the Assumed Obligations, and the Seller wishes to sell the Purchased Assets to the Purchaser, upon and subject to the terms and conditions of this Agreement.
- C. The Purchaser is the wholly-owned subsidiary of the Parent Guarantor and the Parent Guarantor has agreed to guarantee the Purchaser's obligations under this Agreement.

NOW THEREFORE in consideration of the foregoing, and the respective covenants, agreements, representations and warranties of the Parties contained herein and for other good and valuable consideration (the receipt and adequacy of which are acknowledged), the Parties agree as follows:

Article 1 INTERPRETATION

Section 1.1 Defined Terms.

As used in this Agreement, the following terms have the following meanings:

- (1) "Accounts Receivable" means all accounts receivable, notes receivable, trade receivables, rights to receive payment, book debts and other amounts, due, owing or accruing due to the Seller arising out of, from or in connection with the Operations together with any security interest, letters of credit or other credit support documents granted in favour of the Seller as security therefor and Duties related to the Operations.
- (2) **"Acquisition Agreements**" means the Chip Purchase Agreement, the Transition Services Agreement, the Services Agreement, the Non-Competition Agreement, the Pension Plan and Assumption Agreement, the Reverse TSA Arrangement and the Registration Rights Agreement.
- (3) "Advance Ruling Certificate" means an advance ruling certificate issued pursuant to Section 102 of the Competition Act.
- (4) "Affiliate" means, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. As used in this

definition, the term "control" (including, with its correlative meanings, "controlled by" and "under common control with") means the possession, directly or indirectly, of the powers to direct or cause the direction of management or policies of a Person, through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

- (5) "Agreement" means this asset purchase agreement, the Disclosure Letter and the exhibits attached to it, as the same may be amended or supplemented from time to time.
- (6) "Ancillary Forestry Authorizations" has the meaning specified in Section 3.1(37)(d).
- (7) "Antitrust Law" means any Laws that is designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or eliminating, restricting or lessening of competition (including the HSR Act, the Competition Act and similar Laws).
- (8) "Assumed Benefit Plan" has the meaning specified in Section 11.2(2).
- (9) "Assumed Obligations" means (i) all claims, obligations and liabilities, other than Excluded Obligations, of the Seller arising in respect of the period from and after the Closing, in each case to the extent relating to or arising in connection with the Operations or the Purchased Assets, including all asset retirement obligations in relation thereto, (ii) subject to and in accordance with Section 11.1, all liabilities related to the Employees; (iii) the Environmental Liabilities (except for the environmental liabilities described at Section 10.2(1)(e) and the Excluded Environmental Liabilities); (iv) all liabilities and obligations related to the Assumed Benefit Plans; and (v) all liabilities and obligations relating to the benefit plans described in Section 11.2(4).
- (10) "Authorization" means, with respect to any Person, any order, permit, approval, consent, waiver, licence, or similar authorization issued, granted, given or authorized by, or made applicable under the authority of, of any Governmental Entity having jurisdiction over the Person.
- (11) "Base Amount" has the meaning specified in Section 2.3(1).
- (12) "Benefit Plan" means any plan, policy, agreement or arrangement (whether written or unwritten) relating to stock options, stock purchases, stock awards, deferred compensation, bonus, profit sharing, severance, retention, termination, retirement, pension, supplemental pension, health, dental or other medical, life, disability or other insurance (whether insured or selfinsured), mortgage insurance, employee loan, employee assistance, supplemental benefits or other employee benefits, supplementary retirement, welfare benefits, change of control, fringe benefits, supplemental benefits or other employee benefits, in each case, sponsored, maintained or contributed to, or required to be sponsored, maintained or contributed to, by an employer, for the benefit of one or more current or former employees, officers, directors or individual independent contractors, other than, the Canada Pension Plan, the Québec Pension Plan, any health or drug plan established and administered by a Province, and any employment, parental or workers' compensation insurance provided by Canadian federal or provincial Laws.

- (13) "Books and Records" means all information in any form relating to the Operations, including books of account, financial and accounting information and records, tax records, personnel records, sales and purchase records, cost and pricing information, customer and supplier lists, referral sources, research and development reports and records, production reports and records, plan and projections, equipment logs, operating guides and manuals, technical documents including specifications, marketing and advertising materials and all other documents, files, and corporate records.
- (14) "Business Day" means any day of the year, other than a Saturday, Sunday or any day on which major banks are generally closed for business in Montreal, Québec, or Toronto, Ontario, Canada.
- (15) "Cap" means twenty-five million dollars (\$25,000,000).
- (16) "Carve-out Financial Statements" means the audited annual and reviewed interim financial statements prepared by the Seller in respect of the Operations as the Purchaser has requested the Seller to prepare.
- (17) "Chip Purchase Agreement" means the chip purchase agreement, substantially in the form attached as Exhibit A.
- (18) "Chip Purchase Set-Off Amount" has the meaning specified in Section 2.3(3).
- (19) "Chip Purchase Set-Off Note" has the meaning specified in Section 2.5(1)(c).
- (20) "Claim" has the meaning specified in Section 10.5(1).
- (21) "Closing" means the completion of the transaction of purchase and sale of the Purchased Assets contemplated in this Agreement.
- (22) "**Closing Cash Payment**" means an amount equal to: (i) the Base Amount, plus (ii) the Estimated Inventory Amount, minus (iii) an amount equal to 1.5 multiplied by the result of the division between the number of Consideration Shares issued in accordance with Section 2.5(1)(b) hereof, and the US Dollar to Canadian Dollar daily exchange rate published by the Bank of Canada for the last published day immediately prior to Closing.
- (23) "Closing Date" means (a) the fifth Business Day after the day in which the last of the conditions set forth in Section 7.1 and Section 7.2 (other than any such conditions which by their nature are not capable of being satisfied until the Closing Date) is satisfied or waived, (b) if such date is prior to July 31, 2021, July 31, 2021, (c) or on such other date as the Parties may mutually determine.
- (24) "Collective Bargaining Agreement" has the meaning specified in Section 3.1(22)(a).
- (25) **"Commissioner of Competition**" means the Commissioner of Competition appointed under the Competition Act or any Person authorized to exercise his powers and perform his duties.
- (26) "Competition Act" means the Competition Act (Canada).

- (27) **"Competition Act Clearance**" means one of the following has occurred in respect of the transactions contemplated by this Agreement:
 - (a) the Commissioner of Competition shall have issued an Advance Ruling Certificate; or
 - (b) either (i) the requirement to notify the Commissioner of Competition and supply information shall have been waived pursuant to Section 113(c) of the Competition Act or (ii) the applicable waiting period under Section 123 of the Competition Act shall have expired or been terminated and, in the case of (i) or (ii), the Commissioner of Competition shall have issued a No-Action Letter, and such No-Action Letter remains in force and effect at Closing.
- (28) "Confidentiality Agreement" has the meaning specified in Section 6.3.
- (29) "Consent" means any consent, approval, waiver, notice or other authorization required under a Contract.
- (30) "Consideration Shares" means such number of Purchaser as is equal to the lesser of:
 - (a) the number of Purchaser Shares that would result in RYAM GP holding, if such Purchaser Shares were (notwithstanding the timing terms thereof) immediately upon Closing exchanged for Parent Shares, 19.9% of the Parent Shares issued and outstanding on the Closing Date (for greater certainty, the calculation of the outstanding Parent Shares shall include for these purposes, (a) the number of Parent Shares underlying the Purchaser Shares and (b) and Parent Shares issued in connection with the Financing), and
 - (b) (A x B x C) where

A is equal to the Base Amount plus the Estimated Inventory Amount;

B is equal to 0.099333; and

C is equal to the US Dollar to Canadian Dollar daily exchange rate published by the Bank of Canada for the last published day immediately prior to Closing.

- (31) "Contract" means any written, electronic or oral agreement, contract, obligation, promise, commitment, loan or credit agreement, employment or severance agreement, note, mortgage, bond, indenture, lease, benefit plan, permit, franchise, license or other instrument, understanding, undertaking or arrangement and for greater certainty excludes the Real Property Leases.
- (32) "Custodial Trustee" means the CIBC Mellon Trust Company.
- (33) "**Damages**" means any loss, liability, damage, cost or expense (including reasonable legal or other professional fees and expenses).

- (34) "Data Room" means the virtual data room of the Seller maintained by Donnelly Financial Solutions "Venue", access to which was provided to the Purchaser and its representatives for purposes of their due diligence investigations in connection with the transactions contemplated by this Agreement.
- (35) "DC SERP Benefits" means the defined contribution benefits accrued by [**] under the Supplementary Retirement Plan for Designated Senior Executives of Rayonier A.M. Canada Industries Inc. and its Affiliates.
- (36) "Debt Commitment Letter" means the commitment letter from the Debt Financing Sources dated on or about the date hereof to provide Debt Financing or any replacement thereof.
- (37) "Debt Financing" means the debt financing provided by the Debt Financing Sources as set forth in the Debt Commitment Letter.
- (38) "Debt Financing Sources" means [**] and each party to any joinder agreement, indenture or credit agreement entered into in connection with the Debt Financing, including the Debt Commitment Letter, or relating thereto, together with their respective Affiliates, and their respective Affiliates' officers, directors, employees, agents and representatives and their successors and assigns.
- (39) "**Deductible**" means \$610,000.
- (40) "Direct Claim" means any cause, matter, thing, act, omission or state of facts not involving a Third Party Claim which entitles an Indemnified Party to make a claim for indemnification under this Agreement.
- (41) "Disclosure Letter" means the disclosure letter delivered by the Seller to the Purchaser contemporaneously with the execution and delivery of this Agreement.
- (42) "**DRA**" has the meaning specified in Section 8.1.
- (43) "Draft Closing Statement" has the meaning specified in Section 2.7.
- (44) "Duties" means any duties or taxes levied or imposed on imported goods under (i) the Customs Tariff (Canada), the Excise Act (Canada), Excise Act, 2001 (Canada), the Excise Tax Act (Canada), the Special Import Measures Act (Canada), or any other similar legislation enacted in Canada (but not including the goods and services tax or harmonized sales tax imposed under Part IX of the Excise Tax Act (Canada)), (ii) the Trade Act (United States), the Tariff Act of 1930 (United States) or any other similar legislation enacted in the United States or in a foreign jurisdiction.
- (45) "**Employees**" means all Unionized Employees and Non-Unionized Employees, including without limitation, those on temporary layoff or an approved or statutory leave of absence.
- (46) "Environment" means all or any part of the following media (alone or in combination): air (including the atmosphere and the air within any structure or underground space); water (including sea, seabed, marine sediments and any marine organisms and ecosystem,

groundwater and any water under or within land and coastal and inland waters); land (including subsoil and land under water) and any organisms or systems supported by those media (including habitats, species and humans).

- (47) "Environmental Laws" means all Laws relating to (a) protection of human health and/or the Environment, including but not limited to, those relating to pollution, waste, emissions, discharges, or Releases of Hazardous Substances or any other solid, liquid, gas, odour, heat, sound, vibration, or radiation, (b) protection and conservation of natural resources, including but not limited to, climate, air, surface water, groundwater, wetlands, land surface, subsurface strata, wildlife, aquatic, terrestrial, avian or microbial species and vegetation, (c) the manufacture, generation, handling, transport, transfer, labelling, packaging, sale, distribution, import, export, use, processing, treatment, recycling, storage, destruction, or disposal of, or exposure to, Hazardous Substances, and/or (c) any other criminal, civil, equitable, or common law principle concerning any act or omission relating to the Environment or Hazardous Substances. "Environmental Laws" excludes Forestry Laws.
- (48) "Environmental Liabilities" means all liabilities, obligations, responsibilities, response, remedial and removal costs, investigation costs, capital costs, operation and maintenance costs, losses, damages, including damages for diminution of the value of land, property damages, natural resource damages, costs and expenses, fines, penalties and sanctions incurred as a result of or related to any matter or condition arising under or related to any Environmental Laws or environmental Permits, in each case, relating to or arising from or in connection with the Operations or the Purchased Assets, but excluding the Excluded Environmental Liabilities.
- (49) "Estimated Inventory Amount" has the meaning set out in Section 2.6.
- (50) "Excluded Assets" has the meaning set out in Section 2.2.
- (51) "Excluded Benefit Plan" has the meaning specified in Section 11.2(1).
- (52) "Excluded Employees" has the meaning specified in Section 11.1(2).
- (53) "Excluded Environmental Liabilities" means, collectively:
 - (a) any charge or other amount owning as a result of the greenhouse gas emissions of the Operations and/or the Purchased Assets for any compliance period prior to the Closing Date under any Environmental Laws;
 - (b) any fine or monetary penalty imposed by a Governmental Entity in relation to any breach or non-compliance with Environmental Laws or environmental Permits in relation to the Operations and/or the Purchased Assets that occurred at any time prior to the Closing Date; and
 - (c) any expense resulting from an administrative Order imposed by a Governmental Entity relating to Hazardous Substances above the applicable concentration standards and/or value limits as set out in any Environmental Laws or the *Soil, ground water and sediment standards for use under Part XV.1 of the Environmental Protection Act* (in the case of lands in Ontario) or the *Guide*

d'intervention – Protection des sols et rehabilitation des terrains contaminés (in the case of lands in Quebec), as in force on the Closing Date, that have migrated from the Owned Real Property or Leased Real Property and that resulted from the Operations at any time prior to the Closing Date, excluding any aggravation thereof after the Closing Date,

in each case, relating to or arising from or in connection with the Operations or the Purchased Assets.

- (54) "Excluded Obligations" means all obligations and liabilities of the Seller that are not Assumed Obligations, including obligations and liabilities of the Seller: (i) for any Taxes of the Seller, including for greater certainty any liability for Taxes relating to the Operations for any period (or portion thereof) ending on or prior to the Closing Date, (ii) in respect of Indebtedness, (iii) for any fees and expenses incurred by the Seller in connection with negotiating, preparing, closing and carrying out this Agreement and the transactions contemplated hereby (including any liability to an Indemnified Party in connection with any breach of a representation, warranty or covenant contained in this Agreement), (iv) to pay any amount to or for the benefit of any officers, directors, employees or consultants of the Seller as a result of the completion of the Agreement, (v) to pay any amount to or for the benefit of claims, actions or legal proceedings, whether or not pending on the Closing Date, relating to periods prior to the Closing Date, (vii) relating to, or arising out of, the matters described in Section 10.2(1)(e), (viii) relating to the Excluded Assets or, (ix) the Excluded Environmental Liabilities.
- (55) **"Executive Participation Commitment Letters**" means the letters addressed to [**], dated on or about the date hereof and countersigned by the Parent Guarantor, providing for an undertaking from each of [**] to exercise (directly or through an affiliate or family member), collectively, at least US\$4 million of rights (or other existing convertible or exercisable securities) on the terms contemplated by the term sheet to the rights offering attached to the Standby Purchaser Backstop Letter.
- (56) **"Exhibits**" means the exhibits attached to this Agreement.
- (57) "Final Closing Statement" has the meaning specified in Section 2.8(3) and Section 2.8(4).
- (58) "Final Negative Adjustment Amount" has the meaning specified in Section 2.9(2).
- (59) "Final Positive Adjustment Amount" has the meaning specified in Section 2.9(3).
- (60) **"Financial Information**" means, collectively, (i) the Parent Financial Statements, and (ii) the management's discussion and analysis of financial results for the financial year ended December 31, 2020, (iii) the management's discussion and analysis of financial results for the three and nine-month periods ended December 31, 2020.
- (61) "Financial Statements" means, collectively, the unaudited annual management financial statements described in Section 3.1(9) of the Disclosure Letter.

- (62) **"Financing**" means, collectively, the (a) Debt Financing and (b) the asset based lending (ABL) financing in form and substance as required pursuant to the terms of the Debt Financing, (c) the rights offering of approximately C\$99 million in form and substance as contemplated by the terms set out in the Standby Purchaser Backstop Letter, and (d) the financing agreed to in the Executive Participation Commitment Letters, each undertaken to finance the Purchase Price; provided that the Purchaser may consider an alternative superior financing with prior consultation with the Seller.
- (63) "Financing Sources" means the Debt Financing Sources in respect of the Debt Financing, Senvest Management, LLC in respect of the Standby Purchaser Backstop Letter and each of [**] (directly or through an affiliate or family member) in respect of the Executive Participation Commitment Letters.
- (64) "First Nations" means any Indian or Indian Band (as those terms are defined in the *Indian Act* (Canada)), Inuit person or people, First Nation person or people, Métis person or people, indigenous person or people or aboriginal person or people of Canada and any person or group representing, or purporting to represent, any of the foregoing.
- (65) "First Nations' Claims" means any claims, assertions or demands, written or oral, whether proven or unproven, in respect of First Nations' rights, aboriginal rights, aboriginal title, treaty rights or any other aboriginal interest in or to all or any portion of the Owned Real Property and the Leased Real Property.
- (66) "Forest Roads" means the road network, including bridges, built or to be built by the Seller or its Affiliates prior to Closing relating to the Operations.
- (67) **"Forestry Laws**" means all applicable Laws concerning forestry operations (including timber harvesting, road construction, silviculture and other forest management activities) in the provinces of Ontario and Quebec, including the *Forests Act* (Ontario) and the *Sustainable Forest Development Act* (Québec) insofar as it applies to timber harvesting, road construction, silviculture and other forest management activities.
- (68) **"Free Market Timber**" means the free market timber allocated to the Seller in respect of the Operations under the *Sustainable Forest Development Act* (Québec), and available for harvest by the Purchaser from and after the Closing Date or such other volume of free market timber purchased by Seller as same shall be confirmed by the Purchaser prior to Closing.
- (69) **"Fundamental Representations**" has the meaning specified in Section 7.1(1).
- (70) "GAAP" means generally accepted accounting principles as in effect in the United States of America, at the relevant time applied on a consistent basis.
- (71) "Governmental Antitrust Entity" means any state, local or foreign court or other Governmental Entity with jurisdiction over enforcement of any applicable Antitrust Laws.
- (72) **"Governmental Entity**" means any (a) multinational, federal, provincial, territorial, state, municipal, local or other governmental or public department, central bank, court, commission, board, arbitrator, tribunal, bureau or agency, domestic or foreign, (b) any subdivision or authority of any of the above, or (c) any quasi-governmental or private

body exercising any regulatory, expropriation or tax authority under or for the account of any of the above.

- (73) "**GST**" means the taxes (including the goods and services tax and harmonized sales tax) imposed under Part IX of the *Excise Tax Act* (Canada) and the regulations made thereunder.
- (74) "GST and QST Certificate" has the meaning specified in Section 6.10(4).
- (75) "Hazardous Substance" means any substance, chemical, mixture, or material, whether animate or inanimate, that is or may be harmful or hazardous to human, animal, or plant life, any property, any activity, or to the Environment or any natural resources, and includes but is not limited to, anything that is regulated under any Laws as a "contaminant", "source of contaminant", "pollutant", "pesticide", "fuel", "deleterious substance", "toxic substance", "hazardous substance", "controlled substance", "designated substance", "domestic substance", "non-domestic substance", "priority substance", "prohibited substance", "substance subject to notification or consent", "restricted substance", "ozone-depleting substance", "nuclear substance", "hazardous product", "dangerous good", "waste", "hazardous waste", or "hazardous recyclable material".
- (76) "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended and the rules and regulations promulgated thereunder.
- (77) "HSR Act Clearance" means that an early termination shall have been granted or any waiting periods applicable to the consummation of the transactions contemplated hereby under the HSR Act shall have expired or been terminated following filing of a Notification and Report Form.
- (78) **"Indebtedness**" means all (i) indebtedness of the Seller for borrowed money, (ii) any off-balance sheet financing of the Seller, (iii) any liability of the Seller relating to hedging obligations, (iv) any unpaid interest, prepayment premiums or penalties accrued or owing on any such indebtedness of the Seller, (v) any amount owing to a related party of the Seller, (vi) any liabilities for any drawn letters of credit, performance bonds, surety bonds and similar obligations of the Operations (other than, until replaced by the Purchaser, Nominated Security Bonds), (vi) any obligations for the payment of deferred purchase price or similar consideration, (vii) any restructuring reserves, (viii) all accounts payable and other obligations of the Seller in connection with or related to the Operations or the Purchased Assets, (ix) the present value of the liabilities of each Excluded Benefit Plan (calculated using the actuarial assumptions and methods used by the actuary to that Excluded Benefit Plan in its most recent valuation of that Benefit Plan for solvency purposes) to the extent that it exceeds the fair market value of the assets of each such Excluded Benefit Plan that could reasonably be expected to result in liability to the Purchaser; and (x) guarantees of any of the obligations set forth above; and for greater certainty, excludes the capital expenditures described in Section 1.1(78) of the Disclosure Letter.
- (79) "Indemnified Party" means a Person with indemnification rights or benefits under Section 10.2 or Section 10.3, or otherwise under this Agreement.

- (80) "Indemnifying Party" means a Party against which a claim may be made for indemnification under this Agreement, including pursuant to Article 10.
- (81) "Information Technology" means all computer systems, communication systems, software (other than off-the-shelf software) and hardware, whether owned, used or licensed.
- (82) "Insurance Policies" has the meaning specified in Section 3.1(39).
- (83) "Intellectual Property" has the meaning specified in Section 3.1(25)(a).
- (84) "Intellectual Property Rights" means rights in the following: (a) all trademark rights, trade dress, service marks, trade names, domain names, and brand names exclusively used in the Operations, whether registered or existing at common law; (b) all registered and unregistered statutory and common law copyrights and all other rights associated therewith and the underlying works of authorship; (c) all patents rights including patent applications and disclosures; (d) all inventions, know how, discoveries, improvements and trade secrets and (e) all other intellectual property rights owned, licensed, controlled or used by a Person, in any and all relevant jurisdictions in the world.
- (85) "Interim Period" means the period between the date of this Agreement and the Closing.
- (86) "**Inventory**" means all inventory of the Operations, including all finished goods, work–in-progress and raw materials, consumed or sold in the Operations as well as:
 - (a) all unsold merchantable logs, lumber, pellets and other finished goods produced by the Lumber Mills that are at the Lumber Mills or under consignment;
 - (b) all logs suitable for processing into lumber at the Lumber Mills and which (i) have been felled and bucked in cut blocks or road rights of way on the Tenures but have not yet been loaded onto trucks or railway for transport, (ii) are *en route* to the Lumber Mills, or (iii) are at the Lumber Mills awaiting processing;
 - (c) all pulp and other finished goods produced by the Newsprint Mill that are at the Newsprint Mill or under consignment;
 - (d) all logs then being processed into lumber in the Lumber Mills;
 - (e) all unsold merchantable logs, lumber, pellets and other finished goods produced by the Lumber Mills that are at the Lumber Mills or under consignment;
 - (f) all merchantable wood chips, bark, sawdust and hog fuel produced by the Lumber Mills which are at the Lumber Mills; and
 - (g) all supplies, fuel, chemicals, pallets, packaging materials and other manufacturing supplies used primarily or exclusively at the Lumber Mills, the Newsprint Mill or relating to Woodland Operations;

- 11 -

but not including any lumber, pellet inventory of finished goods at the Lumber Mills and at the Newsprint Mill or in transit that has been sold and title has been transferred to another Person before the particular time.

- (87) "Inventory Amount" means the aggregate amount of the Inventory (including for certainty, the Retained Inventory) as at the Closing Date, as finally determined in the Final Closing Statement.
- (88) "Investment Canada Act" means the Investment Canada Act (Canada).
- (89) "Law" means all (a) constitutions, treaties, laws, statutes, codes, ordinances, rules, regulations, and by-laws, whether federal, provincial, municipal, regional, domestic, foreign or international; and (b) all Orders; and (c) policies, guidelines and protocols to the extent they have force of law.
- (90) "Leased Real Property" has the meaning specified in Section 3.1(31)(b).
- (91) "Legal Proceeding" has the meaning specified in Section 3.1(16).
- (92) "Licensed Intellectual Property" has the meaning set out in Section 2.1(17).
- (93) "Liens" means (a) mortgages, hypothecs, charges, pledges, assignments, liens (statutory or otherwise), security interests, easements, servitudes, pre-emptive right or right of first refusal, ownership or title retention agreement, restrictive covenant or conditional sale agreement, imperfections of title or encroachments, rightsofway, rights of refusal, restriction on use or other restrictions, preferential right, or other real, immovable, movable or personal right, prior claims, legal hypothecs or encumbrances of any nature whatsoever, whether contingent or absolute, and any agreement, option, right or privilege (whether by Law, Contract or otherwise) capable of becoming any of the foregoing and (b) any other encumbrance of any nature or any arrangement or condition which, in substance, secures payment or performance of an obligation.
- (94) "Lumber Mills" means the timber processing facilities described in Section 1.1(94) of the Disclosure Letter.
- (95) "MACD" has the meaning set out in Section 10.2(1)(e).
- (96) "Material Adverse Effect" means, any event, change, effect, condition, fact, development, occurrence or circumstance that, individually or in the aggregate with all other events, changes, effects, conditions, facts, developments, occurrences or circumstances, is or would reasonably be expected to be material and adverse to the business, results of operations, capital, properties, condition (financial or otherwise), assets, obligations or liabilities of the Operations with respect to the Seller or the business carried on by any other Person, except any such event, change, effect, condition, fact, development, occurrence or circumstance resulting from or arising out of (a) any change in global, national or regional political, regulatory or legislative conditions or in general economic, business, regulatory, political or market conditions or in national or global financial or capital markets (including credit market or securities markets); (b) any change generally affecting the industries, in which such Person or any of its Subsidiaries, operates; (c) any natural disasters, weather conditions and other force

majeure events; (d) any adoption, proposed implementation or change in applicable Laws, or any changes in generally accepted accounting practices (or any change in the authoritative interpretation by a Governmental Entity of any of the foregoing); (e) any global or national changes or developments in or relating to currency exchange rates or interest rates; (f) any action taken or refrained from being taken, in each case to which the other Party has expressly approved, consented to or requested in writing following the date of this Agreement; (g) any change in the market price or trading volume of such Person's securities (it being understood that the facts and circumstances that may have given rise to such change that are not otherwise excluded from the definition of a Material Adverse Effect may be taken into account in determining whether there has been a Material Adverse Effect); (h) any failure by such Person to meet any internal or public financial projections, forecasts or estimates of revenues, earnings or other financial metrics for any period (it being understood that the facts and circumstances that may have given rise to such failure that are not otherwise excluded from the definition of a Material Adverse Effect may be taken into account in determining whether there has been a Material Adverse Effect); or (i) the negotiation, execution, performance or announcement of this Agreement or the consummation of the transactions contemplated herein, except, in the case of clauses (a)(e) above, to the extent that any such event, change, effect, condition, fact, development, occurrence or circumstance has a materially disproportionate effect on such Person and its Subsidiaries, taken as a whole, relative to other participants in the markets and industries in which such Person operates.

- (97) "Material Contracts" has the meaning specified in Section 3.1(27)(a).
- (98) "Newsprint Mill" means the newsprint operation located at 1 Government Road, P.O. Box 100, Kapuskasing (Ontario) P5N 2Y2.
- (99) "No-Action Letter" means written confirmation from the Commissioner of Competition that he does not, at that time, intend to make an application under Section 92 of the Competition Act in respect of the transactions contemplated by this Agreement.
- (100) "Nominated Permits" has the meaning specified in Section 2.16(3).
- (101) "Nominated Security Bonds" has the meaning specified in Section 2.17(2).
- (102) "Non-Competition Agreement" means the non-competition agreement, substantially in the form attached as Exhibit B.
- (103) "Non-Unionized Employees" means employees of the Operations whose terms and conditions of employment are not covered by a Collective Bargaining Agreement and identified as such in Section 1.1(103) of the Disclosure Letter.
- (104) "Notice" has the meaning specified in Section 12.1.
- (105) "Notice of Claim" has the meaning specified in Section 10.5(1).
- (106) "Ongoing MACD Investigations" has the meaning set out in Section 10.2(1)(e).

- (108) "**Operations**" means the operations of (i) distribution, marketing and sale of lumber newsprint, (ii) harvesting and (iii) processing/ manufacturing, waste disposal and such other activities in each case as conducted, or have been conducted, in connection with the Purchased Assets.
- (109) "**Order**" means any order, writ, injunction, judgment, decision, ruling, awards, plan and decree of any Governmental Entity.
- (110) "Ordinary Course" means, with respect to an action taken by or with respect to a Person, that such action is taken in the ordinary course of the business of the applicable Person consistent with past practices.
- (111) "Outside Date" means September 30, 2021.
- (112) "Owned Intellectual Property" has the meaning set out in Section 2.1(16).
- (113) "Owned Real Property" means the real or immovable property set forth in Section 3.1(31)(b) of the Disclosure Letter.
- (114) "Parent Financial Statements" means the consolidated audited financial statements of the Parent Guarantor for the financial year ended December 31, 2018, 2091 and 2020.
- (115) "Parent Guarantor" has the meaning specified in the introductory paragraph to this Agreement.
- (116) "Parent Shares" means the fully paid and non-assessable common share in the capital of the Parent Guarantor.
- (117) "Parties" means, collectively, the Purchaser, the Parent Guarantor, the Seller and any other Person who may become a party to this Agreement, and "Party" means any one of them.
- (118) "Per Claim Threshold" means \$60,000.
- (119) "**Permits**" means all permits, consents, waivers, licences, certificates, certificates of authorization, approvals, authorizations, registrations, franchises, rights, privileges, quotas and exemptions, or any item with a similar effect, issued, granted, given or authorized by any Governmental Entity. "**Permits**" excludes Ancillary Forestry Authorizations.

- (120)"Permitted Lien" means (a) any Lien for Taxes or governmental assessments, charges or claims of payment not yet delinquent or being contested in good faith by appropriate Legal Proceedings and, in each case, for which adequate accruals, provisions or reserves (based on good faith estimates of management) have been set aside for the payment thereof, (b) vendors', mechanics', materialmens', carriers', workers', landlords', repairmen's, warehousemen's, construction and other similar Liens arising or incurred in the ordinary course of business on a basis consistent with past practice or with respect to liabilities that are not yet due and payable or, if due, are not delinquent or are being contested in good faith by appropriate Legal Proceedings and, in each case, for which adequate accruals or reserves (based on good faith estimates of management) have been set aside for the payment thereof, (c) Liens imposed or promulgated by applicable Laws with respect to real or immovable property, including zoning, building or similar restrictions, as well as Liens consisting of (i) easements, rights of way, servitudes, restrictions and similar rights in real or immovable property and (ii) the reservations, limitations, provisos and conditions, if any, expressed in any original grant from the Crown of any real or immovable property or any interest therein or in any similar grant from a Governmental Entity in jurisdictions other than Canada; provided that any such items would not, individually or in the aggregate, reasonably be expected to interfere in any material respect with the ordinary conduct of the Operations or materially impair the continued use or operation of such real or immovable property for the purpose for which it is currently used, (d) pledges or deposits in the ordinary course of business on a basis consistent with past practice in connection with (A) workers' compensation, unemployment insurance, social security legislation or any other legislation or regulation applicable to the Person concerned or its business or assets or (B) security for bids, tenders, Contracts (other than Contracts for the payment of money) or leases to which the Person concerned is a party as lessee, and (e) Liens identified in Section 1.1(120) of the Disclosure Letter.
- (121) "**Person**" includes an individual, a corporation, a partnership, limited partnership, a limited liability company, an unlimited liability company, an association, a trust or any other entity or organization, including a Governmental Entity.
- (122) "Personal Information" means information that is protected by any Privacy Law.
- (123) "**Privacy Laws**" means the *Personal Information Protection and Electronic Documents Act* (Canada) and any similar Laws governing the protection of personal information.
- (124) "Proceeding" means any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative.
- (125) "Purchase Price" has the meaning specified in Section 2.3.
- (126) "Purchased Assets" means the assets referred to or described in Section 2.1.
- (127) "Purchaser" has the meaning specified in the introductory paragraph to this Agreement.
- (128) "Purchaser Indemnitees" has the meaning specified in Section 10.2(1).
- (129) "Public Record" means all information filed by or on behalf of the Parent Guarantor and its predecessor entities with any of the securities commissions in compliance, or

intended compliance, with any applicable securities Laws from December 31, 2018 to the Closing Date, in each case publicly available on the SEDAR website, other than public information or disclosures that have been modified or superseded by subsequent public information or disclosures by the Parent Guarantor, to the extent so modified or superseded.

- (130) "Purchaser Shares" means the fully paid and non-assessable shares in the capital of the Purchaser.
- (131) "**QST**" means the taxes imposed under Title 1 of an *Act Respecting the Quebec Sales Tax* and the regulations made thereunder.
- (132) "Real Property Leases" has the meaning specified in Section 3.1(31)(b).
- (133) "Registration Documents" has the meaning specified in Section 8.1.
- (134) "Registration Rights Agreement" means the registration rights agreement between RYAM GP and the Parent Guarantor providing for, inter alia, (i) two demand registration rights per year, (ii) piggyback right (exercisable pro rata) in the event that other holders of registrations rights (including Senvest Management, LLC) were to exercise registration rights and (iii) provided RYAM GP proposes to sell Consideration Shares under the Registration Rights Agreement for less than C\$1.50 per share, payment or reimbursement by the Parent Guarantor of all reasonable underwriting fees associated with such sale.
- (135) "Regulatory Approvals" means the Competition Act Clearance and, if required, the HSR Act Clearance, the consent of the Ontario Ministry of Natural Resources to the transfer of the Sustainable Forest Licences pursuant to this Agreement, the consent of the Ministère des Forêts, de la Faune et des Parcs to the transfer of the Supply Guarantees pursuant to this Agreement, the transfer or reissuance of any Permits required by Environmental Laws for the Operations and/or Purchased Assets, and any other Authorizations required to be obtained pursuant to applicable Laws in connection with the transactions contemplated by this Agreement.
- (136) "Reimbursement Amount" has the meaning specified in Section 12.5(2).
- (137) **"Release**" means any release or discharge of any Hazardous Substance, including any discharge, spray, injection, inoculation, abandonment, deposit, spillage, leakage, seepage, pouring, emission, emptying, throwing, dumping, placing, exhausting, escape, leach, migration, dispersal, dispensing or disposal.
- (138) **"Retained Inventory**" means a portion of the Inventory at Closing, comprised of finished goods logs or lumber which has, at the time of Closing, a market value of C\$6 million.
- (139) **"Reverse TSA Arrangement**" means an amendment to the Transition Services Agreement or other arrangement between the Seller and the Purchaser for the Purchaser to provide the services described in Exhibit H to the Seller after Closing, the form of this arrangement is to be agreed between Purchaser and Seller during the Interim Period, each party acting reasonably.

- (140) "RYAM GP" means Rayonier A.M. Canada G.P. and any successor thereof.
- (141) "Security Bonds" has the meaning specified in Section 2.1(17).
- (142) "Seller" has the meaning specified in the introductory paragraph to this Agreement.
- (143) "Seller Benefit Plan" means each Benefit Plan that is sponsored, maintained, administered, contributed to, or required to be contributed to by the Seller for the benefit of one or more Employees.
- (144) "Seller Counsel" means McCarthy Tétrault LLP.
- (145) "Seller Indemnitees" has the meaning specified in Section 10.3.
- (146) "Seller Pre-Closing Communications" has the meaning specified in Section 12.14.
- (147) "SERP" has the meaning specified in Section 11.2(4).
- (148) "Services Agreement" means the services agreement, substantially in the form attached as Exhibit C.
- (149) "SFL Holdcos" means the Persons to which Sustainable Forest Licences have been granted and in which the Seller holds an interest, and "SFL Holdco" means any of such Persons.
- (150) "SFL Shares" has the meaning specified in Section 2.1(15).
- (151) "Share Exchange" has the meaning specified in Section 2.11(1).
- (152) **"Standby Purchaser Backstop Letter**" means the letter addressed to **[**]** of the Parent Guarantor, from Senvest Management, LLC, dated on or about the date hereof and countersigned by the Parent Guarantor, **[**]**, providing for a backstop from Senvest Management, LLC in an amount of \$75 million and includes as a schedule the terms of the rights offering to be effected by the Parent Guarantor.
- (153) "Statement of Adjustments" has the meaning specified in Section 2.13(2).
- (154) **"Subsidiary**" of any Person means any corporation or other form of legal entity (a) an amount of the outstanding voting securities of which is sufficient to elect at least a majority of its board of directors or other governing body (or, if there are not such voting securities, more than 50% of the equity interests of which) is owned or controlled, directly or indirectly, by such Person or by one or more of its Subsidiaries or (b) with respect to which such Person or one or more of its Subsidiaries is the general partner or the managing member or has similar authority.
- (155) **"Supply Guarantees**" means the timber volume supply guarantees granted to Seller in respect of the Operations under the Sustainable Forest Development Act (Québec) by the Ministère des Forêts, de la Faune et des Parcs of Québec.

- 17 -

- (156) **"Sustainable Forest Licences**" means the sustainable forest licences granted, with respect to the Operations, to the Seller or to a Person in which the Seller holds an interest under the *Crown Forest Sustainability Act* (Ontario) by the Ministry of Natural Resources and Forestry of Ontario in connection with the Operations.
- (157) **"Tax"** or **"Taxes"** means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Entity including, (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales, sales and use, real or immovable property, land transfer, personal or movable property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant person, and (iii) all employment insurance premiums, Canada Pension Plan contributions, Québec Pension Plan contributions, including any fine, penalty, interest, or addition to such taxes.
- (158) **"Tax Act**" means the *Income Tax Act* (Canada), as now in effect and as it may be promulgated or amended from time to time.
- (159) **"Tax Return**" means a return, information return, withholding statement, report, election, designation, estimate, claim for refund or other information, form or statement required to be filed in accordance with applicable Laws in connection with, the calculation, determination, assessment, collection or payment of any Taxes, including, where permitted or required, combined or consolidated returns for a group of entities and including any amendment thereof, including any schedule or attachment thereto.
- (160) **"Tenures**" means the rights of the Seller to harvest timber from certain lands owned by the Province of Ontario and Province of Quebec, all as more particularly set out in Section 3.1(37) of the Disclosure Letter and includes all forest licenses, timber sale licenses, Free Market Timber, timber quotas, pulpwood agreements, Supply Guarantees and other Contracts granting harvest rights under the *Sustainable Forest Development Act* (Québec) and the *Crown Forest Sustainability Act* (Ontario) or similar legislation in any other jurisdictions held by the Seller.
- (161) "Tenure Deposits" means the amounts in trust deposited in connection with the Tenures, including, but not limited to forest renewal trusts, forestry futures trusts and silvicultural credits and credits for Forest Roads, all as set out at Section 3.1(37) of the Disclosure Letter.
- (162) **"Tenure Related Assets**" means the rights, title and interest of the Seller in any active, inactive and closed Forest Roads, bridges, dumps, camps, quarries or other facilities used or occupied at any time in connection with the Tenures or the Ancillary Forestry Authorizations.
- (163) "Third Party Auditors" has the meaning specified in Section 2.8(2).
- (164) **"Third Party Claim**" means any action, suit, Proceeding, arbitration, claim or demand that is instituted or asserted by a third party, including a Governmental Entity, against an

Indemnified Party which entitles the Indemnified Party to make a claim for indemnification under this Agreement.

- (165) **"Transfer Taxes**" means all transfer, land transfer, value added, ad-valorem, excise, sales, use, consumption, goods or services, harmonized sales, provincial sales, retail sales, social services, or other similar taxes or duties and any interest, fines and penalties imposed by any Governmental Entity, whether disputed or not.
- (166) "Transferred Employees" has the meaning specified in Section 11.1.
- (167) **"Transition Services Agreement**" means the transition services agreement, substantially in the form attached as Exhibit D.
- (168) "TSXV" means the TSX Venture Exchange.
- (169) **"Unionized Employees**" means employees of the Operations whose terms and conditions of employment are covered by a Collective Bargaining Agreement and identified as such in Section 1.1(169) of the Disclosure Letter.
- (170) **"Woodland Operations**" means the woodland and forestry operations (including silviculture activities) of the Seller or its Affiliates in Quebec and Ontario located at or conducted upon the Tenures and including all machinery, Inventory, Forest Roads, bridges, supplies, rights of way, access rights and other assets related thereto.
- (171) "Wound Up Pension Plans" has the meaning specified in Section 3.1(21)(k).

Section 1.2 Gender and Number.

Any reference in this Agreement to gender includes all genders. Words importing the singular number only shall include the plural and vice versa.

Section 1.3 Headings, etc.

The provision of a Table of Contents, the division of this Agreement into Articles, Sections and Exhibits and the insertion of headings are for convenient reference only and are not to affect its interpretation.

Section 1.4 Currency.

All references in this Agreement to dollars, or to \$ are expressed in U.S. currency unless otherwise specifically indicated.

Section 1.5 Certain Phrases, etc.

In this Agreement, (i) the words "including", "includes" and "include" mean "including (or includes or include) without limitation", and (ii) the phrase "the aggregate of", "the total of", "the sum of", or a phrase of similar meaning means "the aggregate (or total or sum), without duplication, of". In the computation of periods of time from a specified date to a later specified date, unless otherwise expressly stated, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding".

Section 1.6 Accounting Terms.

All accounting terms not specifically defined in this Agreement are to be interpreted in accordance with GAAP.

Section 1.7 Incorporation of Disclosure Letter, schedules and Exhibits.

The Disclosure Letter and any schedule or Exhibit attached to this Agreement is an integral part of this Agreement for all purposes of this Agreement.

Section 1.8 References to Persons and Agreements.

Any reference in this Agreement to a Person or Persons includes his, her or their, directors, officers, administrators, executors, personal legal representatives, successors and permitted assigns. Except as otherwise provided in this Agreement, any reference to this Agreement or any other agreement or document is a reference to this Agreement, such other agreement or document as it may have been, or may from time to time be amended, restated, replaced, supplemented or novated and shall include all schedules to it.

Section 1.9 Statutes.

Except as otherwise provided in this Agreement, any reference in this Agreement to a statute refers to such statute and all rules and regulations made under it, as it or they may have been or may from time to time be amended or reenacted.

Section 1.10 NonBusiness Days.

Whenever payments are to be made or an action is to be taken on, or not later than, a day which is not a Business Day, such payment shall be made or such action shall be taken on the next succeeding Business Day.

Section 1.11 Knowledge.

References in this Agreement to the knowledge of the Seller, will be deemed to refer to the actual knowledge of any of [**], in each case that such individual has, or would reasonably be expected to have, after making diligent inquiry of their direct reports who would reasonably be expected to have knowledge or awareness relevant to the representation and warranty in question.

Section 1.12 Made Available.

References in this Agreement to "made available", "delivered" or "provided" mean that a document or other item of information was made available or provided to the Purchaser and its representatives for viewing in the Data Room, as that site existed as of noon (Eastern time) on April 8, 2021.

Article 2 PURCHASE AND SALE

Section 2.1 Purchase and Sale.

Subject to the terms and conditions of this Agreement, the Seller shall sell, assign, deliver and transfer to the Purchaser free and clear of all Liens (other than Permitted Liens), and the Purchaser shall purchase from the Seller, on the Closing Date, all rights, title and interest in and to all of the property and assets relating to the Operations (other than the Excluded Assets), whether movable or immovable, corporeal or incorporeal, of every kind and description and wheresoever situated, as a going concern (collectively, the "**Purchased Assets**"), including:

- (1) all Owned Real Property together with the buildings, structures, improvements and appurtenances situate thereon;
- (2) all rights as lessee in the Leased Real Property and under the Real Property Leases together with the Seller's interest in all fixtures and leasehold improvements located on the Leased Real Properties;
- all fixtures owned by the Seller, including all fixed machinery and fixed equipment, situated on the Owned Real Property or the Leased Real Property;
- (4) all other machinery and equipment and all vehicles, tools, handling equipment, furniture, furnishings, computer hardware and peripheral equipment, supplies and accessories, spare parts and other tangible property normally situated on the Owned Real Property or the Leased Real Property or used primarily in connection with the Operations, in each case, owned by the Seller;
- (5) all leases (and all rights and benefits thereunder) of machinery and equipment and other tangible property normally situated on the Owned Real Property or Leased Real Property or used primarily in connection with the Operations, to which, in each case, the Seller is a party;
- (6) the Tenures and Ancillary Forestry Authorizations and all Tenure Related Assets;
- (7) the Lumber Mills and Woodland Operations;
- (8) the Newsprint Mill;
- (9) the Forest Roads;
- (10) all Inventory other than Retained Inventory;
- (11) all Free Market Timber;
- (12) all production, shipping and packaging supplies relating to the Operations (but only subject to the rights in respect thereof set forth in the Transition Services Agreement);
- (13) all Contracts or commitments relating primarily to the Operations (and all rights and benefits thereunder), including:

- (b) all forward commitments by the Seller for supplies or materials entered into in the usual and Ordinary Course of the Operations for use in the Operations; and
- (c) those listed in Section 2.1(13) of the Disclosure Letter;
- (14) all Permits applicable to the Operations and its operation in its usual and Ordinary Course, including the Permits listed or described in Section 2.1(14) of the Disclosure Letter, except such Permits that are not assignable or transferable by the Seller to the Purchaser pursuant to applicable Laws;
- (15) all of the interests in the share capital or ownership interests owned by the Seller in the SFL Holdcos and listed in Section 2.1(15) of the Disclosure Letter (the "SFL Shares");
- (16) all Intellectual Property owned by the Seller and belonging to or primarily used in the Operations (the "**Owned Intellectual Property**"), including the Owned Intellectual Property listed in Section 2.1(16) of the Disclosure Letter;
- (17) all Intellectual Property not owned by the Seller but belonging to or used in the Operations (the "Licensed Intellectual **Property**"), including the Licensed Intellectual Property listed in Section 2.1(17) of the Disclosure Letter;
- (18) all of the goodwill attributable to the Operations, together with the right for the Purchaser to represent itself as carrying on the Operations in succession to the Seller;
- (19) all prepaid expenses and deposits (including Tenure Deposits) relating to the Operations;
- (20) the net surplus position of the Assumed Benefit Plans;
- (21) all rights and Claims under all warranties and guarantees in respect of any of the Purchased Assets;
- (22) all claims, causes of action, choses in action, rights of recovery and rights of set-off of any kind (including the right to sue and recover) against third parties relating to the Operations, other than related to Duties paid by Seller prior to the Closing Date;
- (23) any benefits payable under Insurance Policies relating to the Operations or the Purchased Assets relating to damages or losses assumed by the Purchaser (and not indemnifiable by Seller pursuant to Article 10) unless and only to the extent that the Seller has expended money to rectify the matter which is recoverable by the Seller under such insurance claim; and
- (24) all Books and Records, including all correspondence, personnel records, inspection records, and other records, books, documents and data bases recorded or stored by means of any device, including in electronic form, relating solely to the Operations, the Purchased Assets and those employees who are, pursuant to the provisions of this Agreement, to be employed by the Purchaser as are in the possession or under the control of the Seller (or the part thereof relating to the Operations).

Section 2.2 Excluded Assets.

The Purchased Assets shall not include the following property and assets (collectively, the "Excluded Assets"):

- (1) the names "Rayonier", "Rayonier Advanced Materials", "Rayonier A.M.", "Advanced Materials", "RYAM" and any derived forms thereof, either alone or as part of any trade name or trademark which is or has been used in connection with the Operations, together with all product packaging and shipping material bearing such names or trademarks, subject to the right to use such product packaging and shipping material as contemplated in the Transition Services Agreement;
- (2) all insurance policies of the Seller, including those associated with the Operations, and except as provided for under Section 2.1(23), any proceeds paid in connection with such insurance policies and any prepaid insurance premiums;
- (3) all cash on hand, cash equivalents and bank deposits;
- (4) the Accounts Receivable;
- (5) all refunds of Duties, Tax refunds, Tax credits receivable, energy disbursement and any other governmental receivables by the Seller, in each case applicable to periods prior to Closing;
- (6) all personal property of the Seller consumed after the date hereof and prior to the Closing Date in the Ordinary Course in a manner consistent with past practice;
- (7) all personnel records that the Seller is required by Law to retain in its possession (copies of which will be provided to the Purchaser);
- (8) all books and records (including Tax Returns) not relating solely to the Purchased Assets or the Operations, provided that copies of such books and records shall be provided to the extent that they relate to the Operations;
- (9) all amounts owing to the Seller by any of its Affiliates;
- (10) all Seller Pre-Closing Communications;
- (11) any and all rights to future litigation in connection with pre-Closing matters and unrelated to Assumed Obligations or Purchased Assets;
- (12) Intellectual Property, information about suppliers and clients, all rights of the Seller in any Contracts and any other assets of the Seller, in each case which are used in connection with the Operations but primarily used in connection with any other businesses conducted by the Seller or its Affiliates including the Owned Intellectual Property identified as Excluded Assets in Section 2.1(16) of the Disclosure Letter, provided that to the extend that any such Intellectual Property, information or Contract is severable, transferable or assignable in part, such part shall be included as a Purchased Asset; and

- (13) the Contracts identified as Excluded Assets in Section 2.1(13) of the Disclosure Letter; and
- (14) the Retained Inventory.

Section 2.3 Purchase Price.

The aggregate consideration payable by the Purchaser to the Seller for the Purchased Assets (the "**Purchase Price**") will, subject to adjustments provided for in Section 2.9 and Section 2.13, be:

- (1) \$140,000,000 (the "Base Amount"); plus
- (2) the Inventory Amount, <u>plus</u>
- (3) C\$7,900,000 (the "Chip Purchase Set-Off Amount").

Section 2.4 Allocation of Purchase Price.

- (1) At least five (5) days before Closing, the Purchaser will prepare and deliver to the Seller a preliminary statement setting out the allocation of the Purchase Price among the Purchased Assets, including for the purposes of determining any Transfer Taxes as may be payable under Section 6.11. The Seller may review and propose changes to the allocation statement not later than three (3) days after receipt of the statement. The failure of the Seller to propose any changes or otherwise approve the allocation statement within such time period will be deemed to be evidence of its approval thereof.
- (2) Within 30 days following the date of this Agreement, the Purchaser will prepare and deliver to the Seller a statement setting out the allocation of the Purchase Price among the Purchased Assets, including for the purposes of determining any Transfer Taxes as may be payable under Section 6.10. The Seller may review and propose changes to the allocation statement not later than 30 days after receipt of the statement. The failure of the Seller to propose any changes within such time period will be deemed to be evidence of its approval thereof. In any event, the final allocation must be agreed to by the Seller and the Purchaser prior to the Closing Date.
- (3) The allocation statement shall also provide for the allocation of the Consideration Shares amongst the assets, which shall be determined by the Purchaser.
- (4) The Seller and the Purchaser shall each complete all Tax Returns, designations and elections in a manner consistent with the final allocation and otherwise follow the final allocation for all Tax purposes on and subsequent to the Closing Date and not take any position inconsistent with the final allocation, except as may be required by applicable Law. If such allocation is disputed by any Governmental Entity, the Party receiving notice of such dispute shall promptly notify the other Party and the Parties shall use their commercially reasonable efforts to sustain the final allocation.

Section 2.5 Payment on Closing and after Closing.

- (1) The Purchase Price will be paid and satisfied at Closing as set out below in accordance with the terms and conditions set forth in Section 8.1 hereof subject to adjustments provided for in Section 2.9 and Section 2.13:
 - (a) the Purchaser shall pay, by wire transfer of immediately available funds to the accounts designated by the Seller in such proportions between the Seller as determined by the Seller, an amount equal to the Closing Cash Payment;
 - (b) the Purchaser will issue to RYAM GP the Consideration Shares; and
 - (c) the Purchaser shall issue to the Seller, in such proportions between the Seller as determined by the Seller, one or more non-interest bearing promissory having aggregate principal amounts equal to the Chip Purchase Set-Off Amount, which promissory note shall be in substantially in the form set out in Exhibit G (the "Chip Purchase Set-Off Note").

Section 2.6 Estimated Inventory Amount.

Not later than five (5) Business Days prior to the Closing Date, the Seller shall deliver to the Purchaser a statement setting out a good faith estimate of the expected Inventory Amount (the "**Estimated Inventory Amount**") at 11:59 p.m. on the day before the Closing Date, together with reasonable supporting documentation.

Section 2.7 Preparation of Closing Statement.

Within ninety (90) days following the Closing Date (or such other date as is mutually agreed to by the Seller and the Purchaser in writing), the Purchaser will prepare and deliver to the Seller a draft statement setting out the Inventory Amount as at 11:59 p.m. on the day before the Closing Date (the "**Draft Closing Statement**").

Section 2.8 Settlement of Objections.

- (1) The Seller will have a period of forty-five (45) days to review the Draft Closing Statement following receipt of it. The Seller must notify the Purchaser in writing if the Seller has any objections to the Draft Closing Statement within such forty-five (45) day period. The notice of objection must contain a statement and description of the Seller's objections and each amount in dispute. The Purchaser shall, upon request, provide the Seller and its representatives with reasonable access to all Books and Records and employees necessary or desirable (in the Seller's opinion, acting reasonably) to permit the Seller to verify the accuracy, presentation and other matters relating to the preparation of the Draft Closing Statement.
- (2) If the Seller sends a notice of objection of the Draft Closing Statement within the forty-five (45) day period set forth in Section 2.8(1), the Seller and the Purchaser will work expeditiously and in good faith in an attempt to resolve such objections and amounts in dispute within a further period of twenty (20) days following the date of notification by the Seller to the Purchaser of such objection. Failing resolution of any objection to the Draft Closing Statement raised by the Seller, the objections will be submitted for determination

to an independent firm of chartered accountants (the "Third Party Auditors") mutually agreed to by the Seller and the Purchaser (and, failing such agreement between the Seller and the Purchaser within a further period of five (5) Business Days, such independent firm of chartered accountants shall be KPMG LLP, or if such firm is unable to act or is not independent to all Parties at such time, the Parties shall apply to a court with competent authority to appoint an independent accounting firm). The determination of the Third Party Auditors shall be final and binding upon the Seller and the Purchaser and will not be subject to appeal, absent manifest error. The Seller and the Purchaser shall use commercially reasonable efforts to cause the Third Party Auditors to make their determination of the amounts in dispute that have been referred to it within thirty (30) days of their engagement. While the Third Party Auditors are performing their engagement, the Parties shall not communicate with the Third Party Auditors on the subject matter of their work relating to this Agreement, except by joint conference call, joint meeting or letter with copy simultaneously delivered to the other Parties. The Third Party Auditors shall allow each Party to present their respective positions regarding the Draft Closing Statement, and each Party will have the right to present additional documents, materials and other information, and make an oral presentation to the Third Party Auditors regarding the objections. The Third Party Auditors shall consider such additional documents, materials and other information and such oral presentations. Any such other documents, materials or other information will be copied to each Party and each Party will be entitled to attend any such oral presentation, and to reply thereto. The Third Party Auditors will be acting as experts not as arbitrators and, in making its determination, the Third Party Auditors will only consider the amounts in dispute placed before it as contained in the notice contemplated hereby.

- (3) If the Seller does not notify the Purchaser of any objection within the forty-five (45) day period set forth in Section 2.8(1), the Seller will be deemed to have accepted and approved the Draft Closing Statement and such Draft Closing Statement will be final, conclusive and binding upon the Seller and the Purchaser, and will not be subject to appeal. The Draft Closing Statement will become the "Final Closing Statement" on the next Business Day following the earlier of (i) the end of such forty-five (45) day period and (ii) the date on which the Seller notifies the Purchaser, in writing, that the Seller has no objection to the Draft Closing Statement.
- (4) If the Seller sends a notice of objection within the forty-five (45) day period set forth in Section 2.8(1), the Purchaser will revise the Draft Closing Statement to reflect the final resolution or final determination of such objections under Section 2.8(2) within two (2) Business Days following such final resolution or determination. Such revised Draft Closing Statement will be final, conclusive and binding upon the Seller and the Purchaser, and will not be subject to appeal. The Draft Closing Statement will become the "Final Closing Statement" on the next Business Day following revision of the Draft Closing Statement under this Section 2.8(4). The Seller and the Purchaser will each bear their own fees and expenses in preparing or reviewing, as the case may be, the Draft Closing Statement and the Final Closing Statement. In the case of objections and the retention of the Third Party Auditors as provided in Section 2.8(2), the costs and expenses of such firm of chartered accountants shall be equally borne and shared by both the Seller and the Purchaser except that such cost and expenses shall be entirely borne by: (i) the Purchaser if the Inventory Amount set out in the Final Closing Statement by an amount

greater than 25% of the Inventory Amount set out in the Draft Closing Statement; or (ii) the Seller if the Inventory Amount set out in the Seller's statement of objection provided pursuant to Section 2.8(1) exceeds the Inventory Amount set out in the Final Closing Statement by an amount greater than 25% of the Inventory Amount set out in the Draft Closing Statement. The Seller and the Purchaser will each bear their own costs in presenting their respective cases to such firm of chartered accountants.

(5) The Seller and the Purchaser agree that the procedure set forth in this Section 2.8(1) for resolving objections with respect to the Draft Closing Statement is the sole and exclusive method of resolving such objections. This Section 2.8(4) will not prohibit the Seller or the Purchaser from instigating litigation to compel specific performance of Section 2.8(1) or to enforce the determination of the Third Party Auditors.

Section 2.9 Payment of Adjustment to Purchase Price.

- (1) The aggregate Purchase Price payable to the Seller shall be increased or decreased, as the case may be, if the Inventory Amount is more or less than the Estimated Inventory Amount, as follows:
 - (a) if the Inventory Amount is less than the Estimated Inventory Amount, there shall be a corresponding decrease to the Purchase Price, on a dollar for dollar basis, the amount of such decrease will be owed by the Seller to the Purchaser and paid as hereinafter provided; or
 - (b) if the Inventory Amount is greater than the Estimated Inventory Amount, there shall be a corresponding increase to the Purchase Price, on a dollar for dollar basis, the amount of such increase will be owed by the Purchaser to the Seller and paid as hereinafter provided.
- (2) If the net amount based on the calculations set forth in Section 2.9(1) is owed by the Seller to the Purchaser (the "Final Negative Adjustment Amount"), then within five (5) Business Days of the determination of the Final Closing Statement, the Seller will pay or cause to be paid to the Purchaser by wire transfer of immediately available funds the Final Negative Adjustment Amount for and on behalf of the Seller.
- (3) If the net amount based on the calculations set forth in Section 2.9(1) is owed by the Purchaser to the Seller (the "Final Positive Adjustment Amount"), then within five (5) Business Days of the determination of the Final Closing Statement, the Purchaser will pay to the Seller the Final Positive Adjustment Amount by wire transfer of immediately available funds to the account designated by the Seller.
- (4) All Purchase Price adjustments contemplated by this Section 2.9 will be made in respect of the Closing Cash Payment and, for certainty, not in respect of the Consideration Shares or the Canadian Dollar Note.

Section 2.10 No Effect on Other Rights.

The determination and adjustment of the Purchase Price in accordance with the provisions of Section 2.6, Section 2.7, Section 2.8, Section 2.9 and Section 2.13 shall not limit or affect any other rights or causes of action either the Purchaser or the Seller, as the case may be, may

have with respect to the representations, warranties, covenants and indemnities in their favour contained in this Agreement.

Section 2.11 Share Exchange.

- (1) Immediately after the issuance of the Consideration Shares in accordance Section 2.5, RYAM GP covenants and agrees to sell, assign, transfer and deliver the Consideration Shares to the Parent Guarantor in exchange for the issuance by the Parent Guarantor, of one Parent Share for each Consideration Share (the "Share Exchange").
- (2) RYAM GP shall deliver to the Parent Guarantor the share certificates issued to it, and representing the Consideration Shares duly endorsed in blank for transfer purposes or accompanied by duly executed transfer forms.
- (3) Upon receipt of the share certificates referred to in Section 2.11(2), the Parent Guarantor shall issue the corresponding Parent Shares to RYAM GP and deliver to RYAM GP the share certificates representing the Parent Shares issued to RYAM GP (or evidence thereof, which Parent Shares are expected to represent approximately 17.5%, any event, in accordance with the definition of Consideration Shares, less than 20%, of the Parent Shares issued and outstanding on the Closing Date (for greater certainty, the calculation of the outstanding Parent Shares shall include for these purposes, (a) the number of Parent Shares underlying the Purchaser Shares and (b) and Parent Shares issued in connection with the Financing).

Section 2.12 Tax Elections

- (1) At the request of RYAM GP, the Purchaser and the partners of RYAM GP shall jointly make and execute one or more elections, in the prescribed form and within the prescribed time limits, to have Section 85 of the Tax Act (and the equivalent provision of any applicable provincial tax legislation) apply in respect of the disposition of the Purchased Assets by RYAM GP in consideration for, *inter alia*, the Consideration Shares issuable to RYAM GP. The "agreed amount" for purposes of such election for each of the Purchased Assets shall be as determined by RYAM GP, in its sole discretion, within the limits of Section 85 of the Tax Act.
- (2) At the request of RYAM GP, the Parent Guarantor and the partners of RYAM GP shall jointly make and execute an election, in the prescribed form and within the prescribed time limits, to have Section 85 of the Tax Act (and the equivalent provision of any applicable provincial tax legislation) apply in respect of the disposition of the Consideration Shares by RYAM GP in consideration for the issuance of Parent Shares to RYAM GP in connection with the exchange of the Consideration Shares pursuant to the Share Exchange. The "agreed amount" for purposes of such election shall be as determined by RYAM GP, in its sole discretion, within the limits of Section 85 of the Tax Act.
- (3) RYAM GP will be solely responsible for filing the elections described in Section 2.12(1) and Section 2.12(2) within the time prescribed by Section 85(6) of the Tax Act (and the equivalent provision of any applicable provincial tax legislation), or at such later time along with payment of the corresponding late-filing penalties. For greater certainty, the Seller will be solely responsible for the payment of any interest, penalties and other costs

(4) If applicable, and at the request of the Seller, the Seller and the Purchaser shall execute and file, on a timely basis and using any prescribed form, a joint election under subsection 20(24) of the Tax Act and the equivalent provision of any applicable provincial tax statute, in respect of amounts paid by the Seller to the Purchaser for assuming future obligations. In this regard, the Seller and the Purchaser acknowledge that a portion of the Purchased Assets transferred by the Seller to the Purchaser pursuant to this Agreement and having a value equal to the amount elected under subsection 20(24) of the Tax Act or under the provisions of any applicable provincial tax statute is being transferred by the Seller as payment of the assumption by the Purchaser of such future obligations.

Section 2.13 Real Estate and Other Adjustments.

- (1) Subject to Section 2.17, adjustments shall be made as of the period ending at 11:59 p.m. on the day prior to the Closing Date and shall be paid on the Closing Date for all prepaid expenses, deposits (including Tenure Deposits), prepaid Duties, energy disbursements relating to the Operations, municipal and school taxes, property taxes, local improvement rates and charges, water and assessment rates, utilities, fuel, and all other items normally adjusted between a vendor and purchaser in respect of the sale of property similar to the Owned Real Property in the Province of Québec or Ontario (and the Leased Real Property to the extent the Seller, as tenant, is responsible for the payment of such property taxes), as the case may be. In addition, the adjustments shall include the other matters referred to in this Agreement which are stated to be the subject of adjustment and shall exclude the other matters in this Agreement which are stated not to be the subject of adjustment. The Seller will be liable for the portion of such adjustments based on the number of days in the year occurring prior to the Closing Date, and the Purchaser will be liable for the portion of such adjustments based on the number of days in the year occurring on and after the Closing Date. For any year in which an apportionment is required, the Purchaser will, if required, file all required Tax Returns incident to these taxes assessed for the year in which the Closing Date occurs that are not paid by the Seller as of the Closing Date.
- (2) A draft statement of adjustments in the form to be agreed by the Parties, acting reasonably, shall be delivered to the Purchaser by the Seller at least ten (10) Business Days prior to the Closing Date and shall have annexed to it details of the calculations used by the Seller to arrive at all debits and credits on such statement of adjustments and, where applicable, photocopies of supporting invoices and back-up materials (the "Statement of Adjustments").

Section 2.14 Assumption of Obligations and Liabilities.

- (1) The Purchaser shall, from and after the Closing Date, pay, satisfy, discharge, assume, perform and fulfill in a timely manner the Assumed Obligations.
- (2) The Purchaser's obligations under this Section 2.14 shall not be subject to set-off or reduction by reason of any actual or alleged breach of any representation, warranty or
covenant contained in this Agreement or any right or alleged right to indemnification hereunder.

(3) The Purchaser's assumption of the Assumed Obligations shall in no way be deemed a waiver or release by the Purchaser of any rights, at law or in equity, which the Purchaser may have against the Seller (but only to the extent provided for in this Agreement) as a result of any Claim arising out of the breach by the Seller of any representation, warranty or covenant of the Seller under this Agreement.

Section 2.15 Excluded Obligations and Liabilities.

The Purchaser shall not assume and shall have no obligation to discharge, perform or fulfil the Excluded Obligations.

Section 2.16 NonAssignable Contracts, Commitments and Permits.

- (1) The Seller will use commercially reasonable efforts (other than the payment of money or assumption of obligations in excess of *de minimis* amounts or the amount set out in the applicable Contract or other instrument) to obtain all third party consents or waivers, or give all such notices, as are, in each case, required in accordance with the terms thereof to permit the assignment to, and assumption by, the Purchaser of all the Contracts to be assigned to and assumed by the Purchaser pursuant to this Agreement.
- Nothing in this Agreement will constitute an agreement to assign or an attempted assignment of any Contract or other (2)commitment for which any requisite consent or waiver to the assignment thereof has not been obtained. To the extent permitted by applicable Law, if any requisite consent or waiver has not been obtained on or prior to the Closing, the applicable Contract or other commitment will be held by the Seller as agent for the benefit of the Purchaser and the Purchaser will perform the obligations of the Seller thereunder and be entitled to receive all money becoming due and payable under and other benefits derived from the Contract or other commitment immediately after receipt by the Seller. without set-off or reduction by reason of any actual or alleged breach of any representation, warranty or covenant contained in this Agreement or any other right or alleged right to indemnification hereunder and the Seller shall cooperate with the Purchaser in any reasonable and lawful arrangements designed to provide the benefits of such nonassignable right to the Purchaser, including holding any such non-assignable rights in trust for the Purchaser or acting as agent for the Purchaser, provided that pursuant to such arrangements the Purchaser fully indemnifies the Seller for all costs, obligations or liabilities incurred thereunder or in connection therewith. The Purchaser shall be responsible for and shall pay, satisfy, discharge, assume, perform and fulfill in a timely manner any and all liabilities and obligations arising after the Closing in connection with any such Contract. The holding as agent or trustee by the Seller of any such Contract shall not create a relationship of association, partnership or joint venture between the Seller and the Purchaser.
- (3) With respect to any Permits referred to in Section 2.1(14) which may not be assigned, transferred or reissued to the Purchaser as contemplated hereunder at Closing without the approval or consent of any Governmental Entity (collectively, "**Nominated Permits**"), notwithstanding any other provision hereof, such Nominated Permits shall not be

assigned or transferred pursuant to this Agreement without first obtaining all such approvals or consents required in connection with such assignment or transfer; and from the Closing until such time as all such approvals or consents are obtained and such Permits are assigned, transferred or reissued to the Purchaser as contemplated hereunder:

- (a) the Nominated Permits shall be held as agent by the Seller, for the Purchaser, subject to applicable Laws;
- (b) the Purchaser shall apply for and use all commercially reasonable efforts to obtain all consents and approvals required to assign, transfer or reissue the Nominated Permits to the Purchaser, at its sole cost and expense;
- (c) the Seller shall: (i) act in relation to the Nominated Permits only in accordance with the written instructions and directions of the Purchaser, and not otherwise; (ii) promptly transmit to the Purchaser copies of all notices, claims, demands or other communications that the Seller may receive and that relate in any way to the Nominated Permits, and (iii) at the Purchaser's sole cost and expense, cooperate with the Purchaser and take all such actions and do, or cause to be done, all such things at the request of the Purchaser as shall reasonably be necessary and proper in order that the value of any Nominated Permits be preserved to the benefit of the Purchaser;
- (d) the Purchaser shall be responsible for and shall pay, satisfy, discharge, assume, perform and fulfill in a timely manner any and all liabilities and obligations arising after the Closing in connection with the Nominated Permits;
- (e) the Purchaser shall indemnify and hold the Seller harmless from and against any Damages suffered or incurred after the Closing and that are in any way related to the Nominated Permits, including any and all Damages related to actions taken by the Seller under the authorization, instruction or direction of the Purchaser pursuant to Section 2.16(3)(c), save and except for any Damages arising from fraud or wilful misconduct of the Seller;
- (f) the holding as agent by the Seller of any of the Nominated Permits shall not create a relationship of association, partnership or joint venture between the Seller and the Purchaser; and
- (g) none of the Purchaser Indemnitees shall have any right to indemnification by the Seller pursuant to the terms of this Agreement with respect to any Damages relating to any such Nominated Permits not having been assigned or transferred to the Purchaser as of the Closing or not being issued or reissued in the name of the Purchaser.

Section 2.17 Replacement of Security Bonds.

(1) The Purchaser shall use commercially reasonable efforts to replace as of the Closing Date or as soon as reasonably practicable thereafter all existing security deposits/bonds, letters of credit and Tenure Deposits in the name of the Seller relating to Operations,

including those listed in Section 2.17 of the Disclosure Letter (collectively, the "Security Bonds").

- (2) With respect to any Security Bonds which cannot be replaced by the Purchaser using commercially reasonable efforts at Closing (collectively, "**Nominated Security Bonds**"), notwithstanding any other provision hereof, such Nominated Security Bonds shall not be replaced; and from the Closing until such time as their replacement is possible:
 - (a) the Purchaser shall apply for and use all its commercially reasonable efforts to replace the Nominated Security Bonds, at its sole cost and expense, as soon as possible;
 - (b) the Purchaser shall be responsible for and shall pay, satisfy, discharge, assume, perform and fulfill in a timely manner any and all liabilities and obligations arising and accruing after the Closing in connection with the Nominated Security Bonds, including to pay the fees relating thereto and other related costs incurred by Seller to maintain the Nominated Security Bonds after the Closing Date;
 - (c) the Purchaser shall indemnify and hold the Seller harmless from and against any Damages suffered or incurred after the Closing and that are in any way related to the Nominated Security Bonds remaining in place after Closing, save and except for any Damages arising from gross negligence, fraud or wilful misconduct of the Seller or to the extent such Damages relate to matters arising or accruing before Closing; and
 - (d) the holding as agent by the Seller of any of the Nominated Security Bonds shall not create a relationship of association, partnership or joint venture between the Seller and the Purchaser.

Article 3 REPRESENTATIONS AND WARRANTIES

Section 3.1 Representations and Warranties of the Seller.

Each Seller jointly and severally hereby represents and warrants as follows to the Purchaser:

- (1) Organization. The Seller is a corporate entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation, continuation or organization and has full corporate power and authority to own and operate its property and assets, carry on its business and to enter into, and to perform its obligations under this Agreement and each of the Acquisition Agreements to which it is a party. The Seller is duly licensed, registered or qualified to carry on business and is in good standing in each jurisdiction in which the ownership of the Purchased Assets or the operation of the Operations as currently conducted makes such licensing or qualification necessary.
- (2) Authorization. The Seller has full corporate power and authority to execute, deliver and perform this Agreement and each of the Acquisition Agreements to which it is a party and to consummate the transactions contemplated thereby. The execution and delivery of this Agreement and each of the Acquisition Agreements and the consummation of the transactions contemplated thereby have been duly authorized by all necessary corporate

action on the part of the Seller, do not (or would not with the giving of notice, the passage of time or the happening of any other event or circumstance) result in a breach or a violation of, or conflict with, or allow any other Person to exercise any rights under, any of its constating documents, shareholders' agreements, by-laws or resolutions of its board of directors or shareholders and no other corporate proceedings on the part of the Seller or its securityholders are necessary to authorize this Agreement or the Acquisition Agreements has been duly executed and delivered by the Seller and, assuming due authorization, execution and delivery by the Purchaser and the Parent Guarantor, constitutes a legal, valid and binding obligation of the Seller, enforceable against it in accordance with its terms, subject to limitations with respect to enforcement imposed by law in connection with bankruptcy or similar proceedings and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they are sought.

- (3) **No Conflict with Contracts.** Except as set out in Section 3.1(3) of the Disclosure Letter or as would be immaterial, individually or in the aggregate, to the Business, and subject to Section 2.15, the execution, delivery and performance by the Seller of this Agreement and each of the Acquisition Agreements to which it is a party, do not (or would not with the giving of notice, the passage of time or the happening of any other event or circumstance):
 - (a) result in a breach or a violation of, or conflict with, any Contract binding on or affecting the Seller or the Operations; or
 - (b) result in or give any Person the right to seek, or to cause (i) the termination, cancellation, amendment or renegotiation of any material Contract binding on or affecting the Seller or the Operations, or (ii) the forfeiture or other loss, in whole or in part, of any benefit which would otherwise accrue to the Seller in connection with the Operations or the Purchased Assets.
- (4) Required Consents. There is no requirement for the Seller to making any filing, give any notice or obtain any Consent of any Person who is a party to a Contract binding on or affecting the Seller as a result of, or in connection with, or as a condition to the lawful completion of, the transactions contemplated by this Agreement or any of the Acquisition Agreements except for the filings, notifications and Consents set out in Section 3.1(4) of the Disclosure Letter.
- (5) **Execution and Binding Obligation.** This Agreement and each of the Acquisition Agreements to which the Seller is a party have been (or will be) duly executed and delivered by the Seller and constitute (or will constitute) legal, valid and binding obligations of the Seller, enforceable against it in accordance with their respective terms.

(6) SFL Holdcos.

(a) Section 2.1(15) of the Disclosure Letter sets forth a correct and complete list of all of the SFL Holdcos and the SFL Shares, together with the jurisdiction of organization of each SFL Holdco; and (ii) the percentage of each SFL Holdco's outstanding SFL Shares owned by the Seller. - 33 -

- (b) Copies of the shareholders agreements of the SFL Holdcos have been made available to the Purchaser prior to the date hereof.
- (c) The Seller is the registered and beneficial owner of the SFL Shares, and at Closing, the Seller will transfer to the Purchaser good and valid title to the SFL Shares, free and clear of all Liens other than Permitted Liens.

(7) No Violation.

- The execution, delivery and performance by the Seller of this Agreement and each of the Acquisition Agreements to (a) which it is a party do not, and the performance by the Seller of this Agreement will not, (or would not with the giving of notice, the passage of time or the happening of any other event or circumstance) conflict with, or result in any breach or a violation of, or constitute a default (with or without notice or lapse of time, or both) under, or give rise to a right of, or result by its terms in the termination, cancellation or acceleration of any obligation or the loss of a benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or create any obligation to make a payment to any other Person under, or the loss of, any of the Purchased Assets pursuant to (1) any Laws or Order, (2) any provision of the articles, bylaws or other charter documents of the Seller, or (3) any material Contract to which the Seller is a party, except, in the case of this subclause (3), as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Operations. The execution and delivery by the Seller of this Agreement and each of the Acquisition Agreements to which it is a partner do not, and the performance by the Seller of this Agreement will not, result in the creation of any Lien upon any of the Purchased Assets (excluding any Permitted Liens and except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Operations).
- (b) Except for the Regulatory Approvals, if applicable, and except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Operations, the Seller is not required to submit any notice, report or other filing to, or obtain any Authorization of, any Governmental Entity in connection with the lawful execution, delivery or performance of this Agreement, any of the Acquisition Agreements or the consummation of the transactions contemplated thereby and no waiver, consent, approval or authorization of any Governmental Entity is required to be obtained or made by the Seller in connection with its execution, delivery or performance of the transactions contemplated hereby.

(8) Books and Records.

(a) All accounting and financial Books and Records relating to the Operations and the Purchased Assets have been fully, properly and accurately kept and are complete in all material respects. Such Books and Records are not recorded, stored, maintained, operated or otherwise wholly or partly dependent upon or held by any means (including any electronic, mechanical or photographic process, whether computerized or not) which are not or will not be available to the Purchaser in the Ordinary Course after Closing.

- (b) The Seller has designed disclosure controls and procedures so that material information relating to the Operations and the Purchased Assets is made known to its management by Persons within the Seller. The Seller has, in connection with the preparation of its financial statements, evaluated its internal controls and procedures to detect any deficiencies in their design or operation which could adversely affect the Seller's ability to record, process, summarize and report financial data relating to the Operations and the Purchased Assets. No senior manager, officer or director of the Seller nor, to the knowledge of the Seller, the Seller's independent auditors has identified or been made aware of any fraud involving employees (including senior management) who prepare or review the financial statements (or any inputs to such financial statements) of the Seller or any claim or allegation regarding same.
- (9) **Financial Statements.** True, correct and complete copies of the Financial Statements have been made available to the Purchaser prior to the date hereof. The Financial Statements have been prepared from and using the Books and Records in accordance with GAAP applied on a consistent basis during the respective periods and dates set forth therein unless otherwise disclosed therein and are complete and accurate in all material respects and present fairly in all material respects:
 - (a) the assets, liabilities, sales, income, losses, retained earnings, accruals, reserves, adjustments and financial condition of the Operations;
 - (b) the results of Operations;
 - (c) the changes in financial position of the Operations; and
 - (d) all as at the dates and for the periods specified in such statements.
- (10) **Working Capital.** The amount of working capital is consistent with amounts held in accordance with the Seller's past practices.
- (11) **Customers and Suppliers.** Section 3.1(11) of the Disclosure Letter sets out a true, correct and complete list of the ten largest customers and ten largest suppliers of the Seller in relation to the Operations by dollar amount for the 12-month period ending December 31, 2020. Such list includes the value of the sales and purchases for each such customer and supplier during that time. To the Seller's knowledge, the benefits of any relationship with the customers or suppliers of the Operations will continue after the Closing Date in substantially the same manner as prior to the date of this Agreement. For greater certainty, to the knowledge of the Seller, no material complaints over the level of service or quality of goods sold has been raised by any of these customers. There has been no termination or cancellation of, and no material modification or change in, the business relationship of the Seller with those customers or suppliers since December 31, 2020.
- (12) **Liabilities.** Except for liabilities (a) set forth on the Financial Statements, (b) incurred since the date of the most recent Financial Statements in the Ordinary Course, (c)

incurred in connection with the transactions contemplated hereby, (d) arising from executory performance obligations under Contracts that are Purchased Assets, or (e) set forth in Section 3.01(12) of the Disclosure Letter, there are no material liabilities of, relating to or affecting the Operations (to the extent it relates to the Purchased Assets and the Assumed Obligations and for which Purchaser will become liable at Closing) that has or would reasonably be expected to have a Material Adverse Effect on the Operations.

(13) **Related Party Transactions.** Except as set out in Section 3.1(13) of the Disclosure Letter, all Contracts, binding upon or affecting the Seller and relating to the Operations or forming part of the Purchased Assets have been entered into on an arm's length basis (within the meaning of the Tax Act). Any amounts due and payable by the Seller to any Affiliate of the Seller in relation to such Contracts are recorded on the Books and Records at their fair market value.

(14) Tax Matters.

- (a) The Seller has duly and timely paid all Taxes, including all instalments on account of Taxes for the current year, that are due and payable by it, whether or not shown on a Tax Return.
- (b) The Seller has prepared and filed all Tax Returns required to be made or prepared by it within the prescribed period with the appropriate Governmental Entity in accordance with applicable Laws. The Seller has reported all income and all other amounts and information required by applicable Law to be reported on each such Tax Return. Each such Tax Return is true, correct and complete in all material respects.
- (c) The Seller has, in all material respects, duly and timely withheld all Taxes and other amounts required by Law to be withheld by it (including Taxes and other amounts required to be withheld by it in respect of any amount paid or credited or deemed to be paid or credited by it to or for the account or benefit of any Person, including any employees, officers or directors and any non-resident Person), and has duly, in all material respects, and timely remitted to the appropriate Governmental Entity such Taxes and other amounts required by Law to be remitted by it.
- (d) The Purchased Assets are not subject to any Liens for Taxes of the Seller, other than Permitted Liens.
- (e) RYAM GP is and will be at Closing a "Canadian partnership" for purposes of the Tax Act and Rayonier A.M. Canada Industries Inc. is not and will not be at Closing a non-resident of Canada for the purposes of the Tax Act.
- (f) The Seller is registered under Part IX of the *Excise Tax Act* (Canada) for GST purposes and under Chapter VIII of Title I of the *Act respecting Québec Sales Tax* for QST purposes, as follows: (A) RYAM GP: GST: 81671 8357 RT0001 and QST 1213978621 TQ0001; and (B) Rayonier A.M. Canada Industries Inc.: GST: 89683 9446 RT0026 and QST: 1017697281 TQ0025.

- (15) **Absence of Certain Changes.** Except as disclosed in Section 3.1(15) of the Disclosure Letter, from November 17, 2017, through the date of this Agreement, (a) the Seller has conducted the Operations only in the Ordinary Course consistent with past practice in all material respects and (b) there has not been any change, event, development, condition, occurrence or combination of changes, events, developments, conditions or occurrences that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Operations. Without limiting the generality of the foregoing, the Seller has not:
 - sold, transferred or otherwise disposed of any assets of the Operations except (i) assets which are obsolete and which, individually or in the aggregate, do not exceed \$750,000 in book value, and (ii) Inventories sold in the Ordinary Course;
 - (b) granted or suffered any Lien upon any assets of the Operations other than a Permitted Lien;
 - (c) made any capital expenditures in respect of the Operations in excess of the amount budgeted for same in the capital expenditure budget provided to the Purchaser;
 - (d) cancelled any debts or claims owed to it in connection with the Operations or amended, terminated or waived any rights of value to the Seller in connection with the Operations;
 - (e) made any bonus or profit sharing distribution or similar payment of any kind other than in accordance with the Benefit Plans;
 - (f) made any payment to an officer, director, former director or related party other than at the regular rates payable by way of salary or other remuneration for the reimbursement of expenses incurred in the Ordinary Course;
 - (g) suffered any extraordinary loss, damage or destruction in respect of the Operations or any of its assets, whether or not covered by insurance;
 - (h) terminated or suffered the termination of, any Material Contract other than due to its expiration in accordance with its terms and not as a result of the potential completion of the transactions contemplated by the Agreement;
 - (i) written down the value of any assets owned or used by the Seller in the Operations, including inventory and capital lease assets, except on account of normal depreciation and amortization;
 - (j) suffered any material shortage or any cessation or material interruption of inventory shipments, supplies or ordinary services to, for or in respect of the Operations;
 - (k) made any forward for the Operations either in excess of the requirements of the Operations for normal operating purposes or at prices higher than the current market prices;

- (I) compromised or settled any litigation or governmental action relating to assets owned or used by the Seller in connection with the Operations;
- (m) cancelled or reduced any insurance coverage on the Operations or its property and assets;
- (n) made any change in any method of accounting or auditing practice;
- (o) made any change in the method of billing or the credit terms made available to the customers of the Operations; or
- (p) authorized, agreed or otherwise committed, whether or not in writing, to do any of the foregoing.
- (16)Litigation; Orders. Except as disclosed in Section 3.1(16) of the Disclosure Letter, there is no claim, action, suit, arbitration, litigation, dispute, mediation, proceeding, audit, investigation, hearing or inquiry, whether civil, criminal or administrative, at law or in equity, before or by any Governmental Entity or arbitrator ("Legal Proceeding") current or pending or, to the knowledge of the Seller, threatened against or affecting the Operations, the Purchased Assets or any other property or assets used by the Seller in connection with the Operations that, if adversely determined, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Operations. Neither the Seller nor the Operations is subject to any Order of any Governmental Entity that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Operations. No event has occurred or circumstance exists which could reasonably be expected to give rise to or serve as a valid basis for the commencement of any action, suit, proceeding, arbitration or investigation by or against the Seller, the Operations, any of the Purchased Assets or any other property or assets used by the Seller in connection with the Operations including the Leased Real Property. Except as set out in Section 3.1(16) of the Disclosure Letter, since November 17, 2017, the Seller has not been subject to any judgment, order or decree, entered in any lawsuit or proceeding relating to the Operations or any of the Purchased Assets nor has the Seller settled any claim relating to the Operations or any of the Purchased Assets prior to being sued or prosecuted or a judgment being given in respect of it and which, in each instance, exceeds \$250,000.
- (17) Permits. Section 3.1(17) of the Disclosure Letters set out a list of all the material Permits held by the Seller in connection with the Operations or the Purchased Assets. The Seller holds all material Permits that are required for the Operations as currently conducted and for the ownership and use of the Purchased Assets and the other property and assets used by the Seller in connection with the Operations. Each Permit is valid, subsisting and in good standing, the Seller is not in material default or breach of any Permit and no proceedings are pending or, to the knowledge of the Seller, threatened to revoke or limit any Permit. Except as set out in Section 3.1(17) of the Disclosure Letter, all such material Permits are assignable to the Purchaser and renewable by their terms or in the Ordinary Course without the need for the Seller or the Purchaser to comply with any special rules or procedures, agree to any materially different terms or conditions or pay any amounts other than routine filing fees.

- (19) Environmental Matters. Except as disclosed in Section 3.1(19) of the Disclosure Letter:
 - (a) the Seller, the Operations, the Purchased Assets, the Owned Real Property, and the Leased Real Property are in compliance in all material respects with Environmental Laws, including any required Permits;
 - (b) the Seller has received no written notice and has no knowledge of any proceeding, Order or decision (judicial or administrative) pending or threatened by any Governmental Entity to revoke, modify, or limit any Permits required by Environmental Laws, and all such Permits are listed in Section 3.1(19) of the Disclosure Letter;
 - (c) the Seller has not received any written notice and has no knowledge of any issued, outstanding, pending, or threatened Orders, complaints, claims, actions, prosecutions, administrative monetary penalties, or investigations, with respect to any Environmental Liabilities of the Operations or the Purchased Assets which are unresolved as of the date hereof, or with respect to any alleged breach, default, or liability under Environmental Laws which is unresolved as of the date hereof, or with respect to any Hazardous Substances on, at, under, migrating from, migrating to, or originating from the Operations, the Owned Real Property, the Leased Real Property and/or the Purchased Assets;
 - (d) the Seller, the Operations, the Purchased Assets, the Owned Real Property, and the Leased Real Property are in material compliance with all financial assurance requirements under Environmental Laws or any Orders or Permits issued pursuant to Environmental Laws, and Seller has received no written notice and has no knowledge of any proceeding or decision (judicial or administrative), pending or threatened, by any Governmental Entity to modify or increase any applicable financial assurance requirements, and all such financial assurance requirements applicable to the Operations are listed in Section 3.1(19) of the Disclosure Letter;
 - (e) to the knowledge of the Seller, there are no Hazardous Substances in the Environment originating from the Operations, and migrating from the Owned Real Property, the Leased Real Property and/or the Purchased Assets in violation of any Environmental Laws or above concentration standards and/or limit values as set out in any Environmental Laws or the *Soil, groundwater and sediment standards for use under Part XV.1 of the Environmental Protection Act* (in the case of lands in Ontario) and/or the *Guide d'intervention Protection des sols et*

réhabilitation des terrains contaminés (in the case of lands in Québec), as in force on the Closing Date;

- (f) to the knowledge of the Seller, all Hazardous Substances disposed of, treated or stored by the Seller on the site of the Operations, the Owned Real Property, the Leased Real Property, or the Purchased Assets have been disposed of, treated and stored in material compliance with all Environmental Laws.
- (g) the Seller has disclosed to the Purchaser the most recent environmental reports, and any other report (most recent or otherwise) that discloses the actual presence of any Hazardous Substances at the Owned Real Property and/or the Leased Real Property, including site assessments, sampling results, environmental audits, and/or compliance audits in the possession of the Seller, that relate to the Operations, the Purchased Assets, the Owned Real Property, and/or the Leased Real Property, including reports relating to any offsite migration of Hazardous Substances, and all such reports are listed in Section 3.1(19) of the Disclosure Letter;
- (h) to the knowledge of the Seller, the Seller has not changed the use or ceased activities on the Owned Real Property or the Leased Real Property which would require environmental reports to be prepared under Environmental Laws;
- (i) no material unbudgeted works or other expenditure is required or planned in relation to the Operations, the Purchased Assets, the Owned Real Property or the Leased Real Property to ensure material compliance with Environmental Laws or Permits issued pursuant to Environmental Laws; and
- (j) to the knowledge of the Seller, the Operations, the Purchased Assets, the Owned Real Property, and the Leased Real Property currently meet the applicable greenhouse gas emissions standards under Environmental Laws and no cost or expenditure, including the payment of charges or fees, will be required for the 2019 and 2020 compliance periods other than as disclosed in Section 3.1(19) of the Disclosure Letter.
- (20) **First Nations.** Except as disclosed in Section 3.1(20) of the Disclosure Letter, the Seller has not received written notice that the Operations or Purchased Assets are subject to any, and to the knowledge of the Seller, there are no, material current or pending unresolved First Nations' Claims that would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Operations.

(21) Employee Benefits.

(a) Section 3.1(21)(a) of the Disclosure Letter contains a true and complete list of each Seller Benefit Plan. The Seller has made available to the Purchaser copies of (i) each Seller Benefit Plan (or, with respect to any unwritten Seller Benefit Plan, a written description thereof) other than copies of multi-employer plans for which contribution obligations are set out in collective agreements made available to the Purchaser; and (ii) to the extent applicable to each Assumed Benefit Plan, (A) the most recent annual report on file and all schedules thereto filed with a Governmental Entity with respect to such Assumed Benefit Plan,

(B) each current trust agreement, insurance contract or policy, group annuity contract and any other funding arrangement relating to such Assumed Benefit Plan, (C) the most recent actuarial report, financial statement or valuation report filed by the Seller with a Governmental Entity, (D) the most recent summary plan description, if any, required to be provided by Seller under applicable Laws with respect to such Assumed Benefit Plan, and (E) all material correspondence to or from any Governmental Entity relating to such Assumed Benefit Plan.

- (b) Each Seller Benefit Plan has been established, operated and administered in all material respects in compliance with its terms and applicable Laws and all material obligations required to be performed by the Seller pursuant to the terms of any Seller Benefit Plan or by applicable Laws have been performed in a timely fashion and there are no outstanding defaults or violations with regard to same. All contributions or other amounts payable by the Seller with respect to each Seller Benefit Plan in respect of Employees have been timely paid or accrued in accordance with applicable accounting standards and applicable Laws.
- (c) There are no actions, suits, audits or investigations by any Governmental Entity, termination proceedings or other claims (except routine claims for benefits payable) under the Assumed Benefit Plans, and as regards the Employees under any Seller Benefit Plan, pending or, to the knowledge of the Seller, threatened, that would not reasonably be expected to have, individually or in the aggregate Material Adverse Effect on the Operations.
- (d) The Seller has no material obligations in respect of any Benefit Plan that is a multiemployer pension plan or a multiemployer benefit plan, except contribution obligations as are set out in collective agreements made available to the Purchaser. The name of each such multi-employer plan is disclosed in Section 3.1(21)(d) of the Disclosure Letter.
- (e) No Seller Benefit Plan provides health insurance, life insurance or death benefits to current or former employees of the Seller beyond their retirement or other termination of service, other than as disclosed in Section 3.1(21)(d) of the Disclosure Letter.
- (f) Except as disclosed in Section 3.1(21)(f) of the Disclosure Letter, the Assumed Benefit Plans that are registered pension plans are fully funded on both a going concern and a termination basis.
- (g) Since the date of the latest Financial Statements, no material promises or commitments have been made by the Seller to amend any Seller Benefit Plan, to provide increased benefits thereunder or to establish any new benefit plan, except as required by applicable Laws or as disclosed in Section 3.1(21)(g) of the Disclosure Letter.
- (h) The liabilities of the Seller under any unfunded or underfunded Seller Benefit Plan are properly accrued and reflected in the Financial Statements as required by GAAP.

- (i) Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (either alone or in conjunction with any other event): (i) entitle any current or former employee, officer, director or independent contractor of the Seller to any payment or benefit (or result in the immediate and full funding of any such payment or benefit) under any Assumed Benefit Plan; (ii) result in the acceleration of the time of payment, funding or vesting of any compensation, equity award or other benefits under any Assumed Benefit Plan; (iii) increase any amount payable under any Assumed Benefit Plan, whether due to a change of control provision or otherwise; or (iv) result in the acceleration of any material obligation pursuant to any Benefit Plan.
- (j) Except as permitted by the Assumed Benefit Plans, their applicable funding agreements and applicable Laws, there has been no withdrawal of assets or any other amounts from any of the Benefit Plans other than proper payments of benefits to eligible beneficiaries, refunds of overcontributions to plan members and permitted payments of reasonable expenses incurred by or in respect of such Assumed Benefit Plans.
- (k) To the knowledge of the Seller, Section 3.1(21)(k) of the Disclosure Letter lists all the registered pension plans under which the Employees have accrued benefits and which have been wound up during the last three years or are in the process of being wound up ("Wound Up Pension Plans"). The Seller shall indemnify and hold the Purchaser harmless from and against any Damages that are in any way related to the Wound Up Pension Plans.
- (I) Any purchase of annuities under any Seller Benefit Plan that is a registered pension plan has been in the form of a "buy-out" and has resulted in a complete and valid discharge of the administrator of any such plan.

(22) Labour and Employee Matters.

- (a) Section 3.1(22) of the Disclosure Letter sets forth a true and complete list of each agreement, collective bargaining agreement, work rule or letter of agreement with any labour union, labour organization, works council, employee association, group of employees or employee and relating to the Operations (each, a "Collective Bargaining Agreement"). Copies of the material Collective Bargaining Agreements have been made available to the Purchaser.
- (b) With respect to the Operations and except as disclosed in Section 3.1(22) of the Disclosure Letter: (i) there are no applications for certification or, to the knowledge of the Seller, threatened or apparent unionorganizing campaigns for employees not covered under a Collective Bargaining Agreement; (ii) there are no current or, to the knowledge of the Seller, threatened strikes, slowdowns, stoppages, walkouts, lockouts or other labourrelated disputes; (iii) there are no complaints of unfair labour practice, charges, grievances, arbitration proceedings or appeals of such matters (other than routine individual grievances), that, if adversely determined, individually or in the aggregate, would reasonably be expected to be material to the Operations; and (iv) there are no successor or related employer applications.

(c) The Seller is in compliance in all material respects with all applicable Laws with respect to employment, employment practices, terms and conditions of employment, wages and hours, vacation, overtime, labour relations, collective bargaining, unfair labour practices, human rights, health and safety, privacy, immigration and pay equity.

(23) Employees.

- (a) The Seller does not have any written employment contracts relating to the Operations with any Person whomsoever, except as disclosed in Section 3.1(23)(b) of the Disclosure Letter.
- (b) Section 3.1(23)(b) of the Disclosure Letter sets out:
 - (i) the names of all Employees (redacted to comply with Privacy Laws, as needed);
 - (ii) their position or title;
 - (iii) their status (such as full time, part time, temporary, casual, seasonal, coop student);
 - (iv) their total annual remuneration, including a breakdown of salary and bonus, profit sharing plan, short-term incentive plan or other incentive compensation, if any;
 - (v) whether the Employee is a Unionized Employee or a Non-Unionized Employee;
 - (vi) their total length of employment including any prior employment that would affect calculation of years of service for any purpose, including statutory entitlements, contractual entitlements (express or implied) benefit entitlement and/or pension entitlement; and
 - (vii) whether any Employees are on any approved or statutory leave of absence and, if so, the reason for such absence and the expected date of return.
- (24) **Privacy.** The Seller has conducted and is conducting the Operations in compliance in all material respects with all applicable Privacy Laws including in connection with its collection, use and disclosure of Personal Information. The Seller has not received any written complaint or notice of any breach or violation of any such Privacy Laws. All Personal Information of the Seller relating to the Operations:
 - (a) has been collected, used or disclosed with the consent of each individual to which such Personal Information relates (if such consent is required under applicable Privacy Laws);

- (b) has been used only for the purposes for which the Personal Information was initially collected or for a subsequent purpose for which consent was subsequently obtained; or
- (c) has been collected, used or disclosed for a purpose in respect of which consent may, under applicable Privacy Laws, be implied.

(25) Intellectual Property.

- (a) Except as would not be material to the Operations, the Seller has good title to, or with respect to items not owned by the Seller, sufficient rights to use all Intellectual Property Rights that is used by the Seller in the conduct of the Operations (collectively, the "Intellectual Property"). Section 3.1(25) of the Disclosure Letter sets forth a true, correct and complete list of the material trademarks and trade names owned by the Seller. To the knowledge of the Seller, there is no infringement or misappropriation by others of Intellectual Property Rights owned or used by the Seller in a manner that would reasonably be expected to be material to the Operations. To the knowledge of the Seller, the conduct of the Operations does not infringe on or misappropriate any Intellectual Property Rights of others in a manner that would reasonably be expected to be material to the Operations. The consummation of the transactions contemplated hereby will not impair any rights of the Seller, to or under any Intellectual Property. The Seller beneficially owns the Intellectual Property free and clear of all Liens. The Seller is not a party to or bound by any Contract or other obligation whatsoever that limits or impairs its ability to sell, transfer, assign or convey the Intellectual Property. No Person has been granted by the Seller any interest in or right to use all or any portion of the Intellectual Property owned by the Seller.
- (b) Except as would not be material to the Operations, no claims with respect to Intellectual Property are pending or, to the knowledge of the Seller, threatened in writing by any Person (i) to the effect that the manufacture, performance, sale or use of any product, process or service as now used or offered or proposed for use or sale by the Seller in the course of their conduct of the Operations infringes on any Intellectual Property Rights of any Person, (ii) against the use by the Seller of any Intellectual Property in the course of their conduct of the Operations or (iii) challenging the ownership, validity, enforceability or effectiveness of any Intellectual Property used in the conduct of the Operations.
- (c) No royalty or other fee is required to be paid by the Seller to any other Person in respect of the use of any of the Intellectual Property and there are no restrictions on the ability of the Seller or any successor to, or assignee from, the Seller to use and exploit all rights in the Intellectual Property.
- (26) **Information Technology.** Section 3.1(26) of the Disclosure Letter sets out a true, correct and complete list of all material Information Technology owned, licensed, used or held for use in connection with the Operations and all Contracts relating to the maintenance and support, security, disaster recovery management and utilization of such Information Technology.

(27) Material Contracts.

- (a) Section 3.1(27) of the Disclosure Letter contains a complete and accurate list of all Material Contracts as of the date of this Agreement. The Seller has delivered to the Purchaser true, complete and correct copies of each Material Contract. For purposes of this Agreement, "Material Contract" means each Contract entered into with respect to the Operations and to which the Seller is a party or by which any of the Purchased Assets is bound that:
 - (i) involves or would reasonably be expected to involve aggregate payments by or to the Seller in excess of \$500,000 over the remaining term of the Contract, except for (A) sales orders and purchase orders for products and raw materials under existing master purchase agreements to operate the Operations entered into in the Ordinary Course of business consistent with past practice, and (B) any Contract which is terminable on 90 days or less notice without payment or continuing obligation;
 - (ii) provides for or otherwise relates to joint venture, partnership, strategic alliance or similar arrangements in excess of \$3,000,000;
 - (iii) constitutes or provides for indentures, mortgages, promissory notes, loan agreements, guarantees, letter of credit or other agreements or instruments of the Seller or commitments for the borrowing or the lending by the Seller with an outstanding payable amount or available commitment, as applicable, in excess of \$1,000,000;
 - (iv) provides for reserve production capacity or a call or option in production;
 - (v) provides for the guarantee of obligations of any Person other than the Seller (other than Ordinary Course endorsements for collection) in excess of \$1,000,000;
 - (vi) is a Contract with respect to Intellectual Property Rights (other than the license of commercially available software) that is material to the Operations;
 - (vii) requires the Seller to make any advance, extension of credit or capital contribution to, or other investment in, any Person, other than Contracts providing for shortterm extensions of credit to customers in the Ordinary Course of business consistent with past practice;
 - (viii) has been entered into in the past 12 months or in respect of which the applicable transaction has not yet been consummated and relating to the purchase or acquisition of a business or operating assets to be used in the conduct of the Operations by the Seller for an aggregate consideration of more than \$1,000,000;
 - (ix) has been entered into in the past 12 months or in respect of which the applicable transaction has not yet been consummated and relating to the sale, transfer or other disposition of a business or operating assets or

real, immovable, movable or personal property used in the conduct of the Operations by the Seller for an aggregate consideration of more than \$1,000,000;

- (x) is a financial risk management Contract, such as a currency, commodity, interest or equity related hedge or derivative Contract;
- (xi) relating to the Tenures;
- (xii) is a Contract that if terminated or if it ceased to be in effect, would reasonably be expected to have a Material Adverse Effect on the Operations.
- (b) Each Material Contract is valid and binding on the Seller and, to the knowledge of the Seller, each other party thereto, and is in full force and effect and unamended. The Seller is entitled to all benefits under, and is not alleged to be in material breach or default under any Material Contract, nor does the Seller have knowledge of any condition that with the passage of time or the giving of notice or both would result in such a breach or default. To the knowledge of the Seller, there is no material breach or default under, nor does there exist any condition that with the passage of time or the giving of notice or both would result in such a breach or default. To the knowledge of time or the giving of notice or both would result in such a breach or default by any other party thereto. The Seller has not received any written or, to the knowledge of the Seller, other notice that any party to a Material Contract intends to cancel, terminate or otherwise modify such Contract, or otherwise terminate its relationship with the Seller, and, to the knowledge of the Seller, no such action has been threatened.
- (c) Except as set out in Section 3.1(27) of the Disclosure Letter, no consent or notice is required for a valid assignment to the Purchaser of any Material Contract.
- (28) **No Breach of Other Contracts.** The Seller has not received written notice of any alleged violation or breach in any material respect of any of the terms or conditions of any Contract (excluding Material Contracts) forming part of the Purchased Assets and, to the knowledge of the Seller, all material covenants to be performed and material obligations to be fulfilled have been fully performed and fulfilled in all material respects.
- (29) **Only Operations.** No operations, other than the Operations are conducted or have been conducted at the Lumber Mills and the Newsprint Mill since November 17, 2017.
- (30) Sufficiency of Purchased Assets. The Purchased Assets include all rights, assets and property necessary for the conduct of the Operations after the Closing substantially in the same manner as it was conducted prior to the Closing. Except for the Excluded Assets, the property and assets included in the Purchased Assets (a) constitute all the assets used by the Seller in carrying on the Operations and (b) include all of the assets set forth or reflected on the Financial Statements other than assets acquired since December 31, 2020 or disposed of in accordance with this Agreement since December 31, 2020.
- (31) **Properties and Assets**.

- (a) The Seller is the legal and beneficial owner of the Purchased Assets (other than the Owned Real Property) with good title thereto, free and clear of all Liens other than Permitted Liens. No other Person owns any assets which are being used in the Operations except for the Leased Real Properties and personal or moveable property leased to the Seller pursuant to the Material Contracts.
- (b) Section 3.1(31)(b) of the Disclosure Letter identifies for each Owned Real Property its registered and beneficial owner and except as would not be material to the Operations, such registered and beneficial owners own the Owned Real Property in such capacity, free and clear of all Liens other than the Permitted Liens.
- (c) The interests in each lease, sublease, license and other agreement under which the Seller uses or occupies or has the right to use or occupy the real or immovable property are set forth in Section 3.1(31)(c) of the Disclosure Letter (such property subject to a lease, sublease or other agreement, the "Leased Real Property" and such leases, subleases and other agreements are, collectively, the "Real Property Leases"). The Real Property Leases are valid and binding on the Seller and, to the knowledge of the Seller, on each other party thereto, and in full force and effect and other than as set forth in Section 3.1(31)(c) of the Disclosure Letter, there are no consents required in order to assign the Seller's right, title and interest in any of the Real Property Leases. The Seller is not in material breach or default under any Real Property Lease, nor does the Seller have knowledge of any condition that with the passage of time or the giving of notice or both would result in such a material breach or default by any other party thereto.
- (d) Except as would not be material to the Operations, there are no leases, subleases, licenses, rights or other agreements affecting the Owned Real Property or, other than the Real Property Leases, the Leased Real Property. The Seller is currently not leasing, subleasing, licensing or otherwise granting any person the right to use or occupy any portion of an Owned Real Property or a Leased Real Property. There are no outstanding options or rights of first refusal in favour of any other party to purchase any Owned Real Property or any portion thereof or interest therein.
- (e) Since November 17, 2017, there has been no off-title transfer of the ownership of the Owned Real Property located in Québec by the Seller which would have triggered the payment of any Transfer Taxes.
- (f) All municipal taxes, rates, levies and assessments of every nature or kind in respect of the Owned Real Property and the Leased Real Property for the calendar year 2019 and all preceding calendar years have been paid in full or will be reflected in the Statement of Adjustments, and the Seller does not have any overdue obligation to pay moneys to any Governmental Entity in connection with services, utilities or the like, relating to the Owned Real Property or the Leased Real Property.

- (33) **Machinery and Equipment.** Section 3.1(33) of the Disclosure Letter sets forth the machinery and equipment (which for greater certainty excludes movable property (personal property) incorporated with an immovable (real property)), each item of which has a book value greater than \$200,000, used by the Seller in the Operations.
- (34) Location of Assets. With the exception of (i) log Inventories that (A) have been felled and bucked in cut blocks or road rights of way on the Tenures but have not yet been loaded onto trucks or railway for transport or (B) are en route to the Lumber Mills, (ii) machinery and equipment that is in the field in the Ordinary Course and (iii) finished lumber en route from the Lumber Mills, all of the Purchased Assets are situate in, on or at the Owned Real Property, the Leased Real Property, the Woodland Operations, or on consignment.
- (35) Expropriation. The Seller has not received written notice of, and the Seller has no knowledge of, any intention of any Governmental Entity to expropriate all or any part of the Owned Real Property or the Leased Real Property or any written notice of any intention to alter its zoning by-law or official community plan or its road or traffic plans so as to affect or potentially affect the Owned Real Property or the Leased Real Property or the operation thereof or access to or egress from the Owned Real Property or the Leased Real Property.
- (36) **Product Liability and Warranty Claims.** The Seller has not provided or agreed to provide any express warranties to its customers other than as may be specified in the Contracts.

(37) Tenures.

- (a) The Seller has not received any written notice of breach of the Tenures, any of the timber cutting rights or permits or operating or development plans issued or filed pursuant to any of the Tenures, other than breaches which individually or in the aggregate, would not be material to the Operations.
- (b) The Seller has not received any written notice from any Governmental Entity with respect to any of the Tenures, the Ancillary Forestry Authorizations or the Tenure Related Assets concerning any matter or breach which would have the effect of amending, varying, reducing, impairing, suspending or terminating such Tenures, Ancillary Forestry Authorizations, Tenure Related Assets or any rights or privileges attached thereto, and they are not otherwise aware of any impeding notice or of any material breach of any of the foregoing or any forest management, operating, reforestation or other plans issued or filed pursuant to any of the Tenures or Ancillary Forestry Authorizations, which has not been remedied or abandoned.
- (c) No written notice or other written statement, or, to the best of the Seller's knowledge, verbal notice or other verbal statement, which is currently

outstanding, has been received by the Seller, advising or threatening the cancellation, amendment, non-renewal or non-extension of any of the Tenures or the Ancillary Forestry Authorizations.

- (d) Section 3.1(37) of the Disclosure Letter contains (i) a list of the Tenure and (ii) a list of any active road permits and cutting permits issued in connection with the Tenures (the "Ancillary Forestry Authorizations"). The declarations made by the Seller to the relevant Governmental Entity in order for such Governmental Entity to publish official information on its website with respect to the Tenures were, at the time they were made, true, correct and complete in all material respects. True, correct and complete copies of each of the Tenures and the Ancillary Forestry Authorizations, and all amendments applicable thereto, have been made available to the Purchaser. Except as disclosed in Section 3.1(37) of the Disclosure Letter or pursuant to our rights under the SFL Holdcos, the Seller has the exclusive right to harvest timber under the Tenures, which are free and clear of all Liens, except Permitted Liens.
- (e) Each of the Tenures and the Ancillary Forestry Authorizations is now recorded on the records of the applicable Governmental Entity in the name of the Seller, and all rental, stumpage, royalty and scale accounts and any other Taxes, assessments, fees and costs arising thereunder that are due and payable now have been paid in full. No notice of suspension, cancellation, reduction in term or realization on any security deposited by the Seller in respect of any of the Tenures or Ancillary Forestry Authorizations is outstanding, and the Seller is not aware of any fact or event that has occurred and could reasonably be expected to result in any such suspension, cancellation, reduction in term or realization.
- (f) The Seller has observed and performed in all material respects, commensurate with good forest industry practice prevailing in Ontario and Quebec as at the date hereof, all covenants, agreements and obligations on their respective parts to be observed or performed under the provisions of each of the Tenures, the Forestry Laws and all Authorizations and other applicable Laws.
- (g) All Tenure Deposits, prepaids, credits and other amounts under the Tenures and, if applicable, Ancillary Forestry Authorizations, are in amounts and in the forms required by the Tenures, Ancillary Forestry Authorizations and Forestry Laws and are listed in Section 3.1(37) of the Disclosure Letter.
- (h) Except as expressly described in the Tenures, there are no privately owned lands within the Tenures.
- (38) Anti-Corruption. Except as would not, individually or in the aggregate, reasonably be expected to be material to the Operations, since November 17, 2017, (a) the Seller has been and is in compliance with all applicable anticorruption Laws, including (i) the United States Foreign Corrupt Practices Act of 1977, as amended, (ii) the Corruption of Foreign Public Officials Act (United States), as amended, (iii) the Corruption of Foreign Public Officials Act (Canada), as amended, and (iv) the Criminal Code (Canada), as amended, and (b) Neither the Seller nor, to the knowledge of the Seller, any director, officer, employee or agent of the Seller has, directly or indirectly, (i) used any funds of the

Seller for unlawful contributions, unlawful gifts, unlawful entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment, loan or benefit of any kind to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns from funds of the Seller, (iii) made any unlawful payments or gifts to any governmental officials out of funds of the Seller (but excluding payments to governmental agencies in amounts legally due and owing by the Seller), (iv) established or maintained any unlawful fund of monies or other assets of the Seller, (v) made any fraudulent entry on the books or records of the Seller or (vi) made any unlawful bribe, unlawful rebate, unlawful payoff, unlawful influence payment, unlawful kickback or other unlawful payment, to obtain favorable treatment in securing business for the Seller, to obtain special concessions for the Seller or to pay for favorable treatment for business secured or to pay for special concessions already obtained for the Seller. Except as would not, individually or in the aggregate, reasonably be expected to be material to the Operations, since November 17, 2017, neither the Seller nor, to the knowledge of the Seller, any director, officer, employee or agent of the Seller has violated or operated in non-compliance with any applicable export restrictions, applicable antiboycott regulations or applicable embargo regulations (including any trade, economic or financial sanctions law or regulations).

- (39) **Insurance.** True, correct and complete copies of all insurance policies maintained by the Seller which relate to the Operations (collectively, the "**Insurance Policies**") as of the date hereof have been made available to the Purchaser as well as the most recent inspection reports received from insurance underwriters. Each Insurance Policy is in full force and effect and in good standing. The Purchased Assets and all other property and assets used in the Operations are insured against loss or damage by all insurable hazards and risks. Section 3.1(39) of the Disclosure Letter is a list of insurance policies which are maintained by or on behalf of the Seller in connection with the Operations setting out, in respect of each policy, a description of the type of policy, the name of insurer, the coverage, the expiration date, the annual premium and any pending claims. The Seller has not failed to give any notice or to present any claim under any insurance policy in a due and timely fashion. Except as set out in Section 3.1(39) of the Disclosure Letter, in the past three years, there has been no change in the relationship of the Seller with its insurers, the availability of coverage or the premiums payable pursuant to the policies. Section 3.1(39) of the Disclosure Letter is a list setting out all claims, with reasonable particulars, made under any policies of insurance maintained by or for the benefit of the Seller in the past three years.
- (40) Investment Canada Act. The Operations are not a cultural business within the meaning of the Investment Canada Act.
- (41) **No Options.** Except for the Purchaser under this Agreement, no Person has any written or oral agreement, option, understanding or commitment, or any right or privilege (whether by law, contractual or otherwise) capable of becoming such for the purchase or other acquisition from the Seller of any of the property and assets forming part of the Purchased Assets other than pursuant to purchase orders for Inventories sold in the Ordinary Course.

- (42) **No Finder's Fee.** The Seller has not taken, and agrees that the Seller will not take, any action that would cause the Purchaser to become liable to any claim or demand for a brokerage commission, finder's fee or other similar payment.
- (43) **Full Disclosure.** Neither this Agreement nor any Acquisition Agreement to which the Seller is a party (a) contains or will contain any untrue statement of fact, or (b) omits or will omit any fact necessary in order to make the statements contained herein or therein not misleading.

Section 3.2 Disclaimer of Other Representations and Warranties

- (1) The representations and warranties set forth in this Article 3 are the only representations and warranties made by the Seller with respect to the Purchased Assets, the Seller, the Operations or any other matter relating to the transactions contemplated by this Agreement. Except as specifically set forth in Article 3, the Seller makes no other representation or warranty concerning (A) the Purchased Assets, the Seller, the Operations or any other matter relating to the transactions contemplated by this Agreement, or (B) the probable success or profitability of the Operations after the Closing. In connection with the transactions contemplated hereby, the Purchaser has been represented by, and consulted with, legal counsel of its choice and the Purchaser and such counsel have read this Agreement and have been given time to consider this Agreement, understand this Agreement, and, after such consideration and with such understanding, the Purchaser has knowingly, freely and without coercion entered into this Agreement and, in particular, this Section 3.2 and Section 10.9. Notwithstanding any other provisions of this Agreement, the representations and warranties listed in this Section 3.2(1) constitute the sole and exclusive representations and warranties of the Seller related to each Seller Benefit Plan.
- (2) Except for the representations and warranties set forth in this Article 3, the Purchaser hereby acknowledges and agrees that the Owned Property are being sold and purchased "as-is, where-is", at the Purchaser's own risk and peril from a non-professional vendor, and without any other representation or warranty, legal or conventional. All warranties, expressed or implied, otherwise provided for by the *Civil Code of Québec* with respect to the Owned Property are hereby waived by the Purchaser, to the maximum extent permitted by law. With respect to the Owned Property, the Purchaser shall rely solely upon its own findings resulting from its due diligence and not upon any information, documentation, statement or opinion, written or oral, provided by the Seller or any agent of the Seller other than the representations and warranties set out in Section 3.1.

Article 4 REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

Section 4.1 Representations and Warranties of the Purchaser.

The Purchaser represents and warrants as follows to the Seller and acknowledges and confirms that the Seller is relying on such representations and warranties in connection with the sale of the Purchased Assets:

- (1) **Organization.** The Purchaser is a Person duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation, continuation or organization and has full power and authority to enter into, and to perform its obligations under this Agreement and the Acquisition Agreements to which it is a party. The Purchaser is a wholly-owned subsidiary of the Parent Guarantor.
- (2) Authorization. The Purchaser has full corporate power and authority to execute and deliver this Agreement and each of the Acquisition Agreements to which it is a party and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Acquisition Agreements to which it is a party and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action on the part of the Purchaser, and no other corporate proceedings on the part of the Purchaser or its securityholders are necessary to authorize this Agreement and to consummate the transactions contemplated hereby. This Agreement has been duly executed and delivered by the Purchaser and, assuming due authorization, execution and delivery by the Seller and Parent Guarantor, constitutes a legal, valid and binding obligation of the Purchaser, enforceable against it in accordance with its terms, subject to limitations with respect to enforcement imposed by law in connection with bankruptcy or similar proceedings and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they are sought.

(3) No Violation.

- (a) The execution and delivery by the Purchaser of this Agreement and the Acquisition Agreements to which it is a party do not, and the performance by the Purchaser of this Agreement and the Acquisition Agreements to which it is a party will not, conflict with, or result in any violation of, or constitute a default (with or without notice or lapse of time, or both) under, or give rise to a right of, or result by its terms in the, termination, cancellation or acceleration of any obligation or the loss of a benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or create any obligation to make a payment to any other Person under, or the loss of, any properties or assets of the Purchaser pursuant to (1) any Laws or Order, (2) any provision of the articles, bylaws or other charter documents of the Purchaser, or (3) any material Contract to which the Purchaser is a party, except, in the case of this sub-clause (3), as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Purchaser.
- (b) Except for Regulatory Approvals, if applicable, the Purchaser is not required to submit any notice, report or other filing with any Governmental Entity in connection with the execution, delivery or performance of this Agreement and the Acquisition Agreements or the consummation of the transactions contemplated hereby and thereby and no waiver, consent, approval or authorization of any Governmental Entity is required to be obtained or made by the Purchaser in connection with its execution, delivery or performance of this Agreement or the consummation of the transactions contemplated hereby.

- (4) Debt Financing. The Purchaser has delivered to the Seller a true and complete, fully-executed copy of the Debt Commitment Letter, including all exhibits and schedules thereto, from the Debt Financing Sources (redacted as to rates, fees, monetary thresholds or other economic terms and excluding any fee letters) evidencing the availability of the Debt Financing. The commitments described in the Debt Commitment Letter are not subject to any condition precedent other than the conditions expressly set forth therein.
- (5) **Other Financing** The Purchaser has delivered to the Seller a true and complete, fully-executed copy of commitment letter or backstop agreement from Financing Sources (other than the Debt Financing Sources) evidencing the availability of the Financing (other than the Debt Financing). The commitments described in the commitment letter or backstop agreement are not subject to any condition precedent other than the conditions expressly set forth therein.
- (6) **Authorized Capital of the Purchaser.** Immediately after the Closing but prior to the Share Exchange, the authorized capital of the Purchaser shall be an unlimited number of common shares, of which the only equity that will be outstanding at such time (including the Consideration Shares) will be the Consideration Shares and such number of common shares held by the Parent Guarantor, all of which (including the Consideration Shares) shall have been duly and validly authorized and issued as fully paid and non-assessable equity of the Purchaser in accordance with applicable Laws.
- (7) Litigation. There are no Legal Proceedings pending or, to the knowledge of the Purchaser, threatened against the Purchaser or any of its respective officers or directors (in such capacity) or affecting its business or assets that, if adversely determined, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Purchaser. None of the Purchaser, its business or assets is subject to any Order of any Governmental Entity that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Purchaser.
- (8) Tax Matters. At or prior to Closing, the Purchaser will be registered under Part IX of the Excise Tax Act (Canada) for GST purposes and under Chapter VIII of Title I of the Act respecting Québec sales tax for QST purposes and will provide its registration numbers to Sellers.
- (9) **No Finder's Fee.** The Purchaser has not taken, and agrees that the Purchaser will not take, any action that would cause the Seller to become liable to any claim or demand for a brokerage commission, finder's fee or other similar payment.
- (10) Investment Canada Act. The Purchaser is a Canadian within the meaning of the Investment Canada Act.

Section 4.2 Disclaimer of Other Representations and Warranties.

The representations and warranties set forth in this Article 4 are the only representations and warranties made by the Purchaser relating to the transactions contemplated by this Agreement. Except as specifically set forth in Article 4, the Purchaser makes no other representation or warranty relating to the transactions contemplated by this Agreement. In connection with the

- 53 -

transactions contemplated hereby, the Seller has been represented by, and consulted with, legal counsel of its choice and the Seller and such counsel have read this Agreement and have been given time to consider this Agreement, understand this Agreement, and, after such consideration and with such understanding, the Seller has knowingly, freely and without coercion entered into this Agreement and, in particular, this Section 4.2 and Section 10.9.

Article 5 REPRESENTATIONS AND WARRANTIES OF THE PARENT GUARANTOR

Section 5.1 Representations and Warranties of the Parent Guarantor.

The Parent Guarantor represents and warrants as follows to the Seller and acknowledges and confirms that the Seller is relying on such representations and warranties in connection with the sale of the Purchased Assets:

- (1) **Organization.** The Parent Guarantor is a Person duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation, continuation or organization and has full power and authority to enter into, and to perform its obligations under this Agreement and the Acquisition Agreement to which it is a party.
- (2) Authorization. Other than consents, approvals and authorizations as are required by the TSXV in connection with the transactions contemplated herein, the Parent Guarantor has full corporate power and authority to execute and deliver this Agreement and each of the Acquisition Agreements to which it is a party and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and each of the Acquisition Agreements to which it is a party and thereby have been duly authorized by all necessary corporate action on the part of the Parent Guarantor, and no other corporate proceedings on the part of the Parent Guarantor or its securityholders are necessary to authorize this Agreement and to consummate the transactions contemplated hereby. This Agreement has been duly executed and delivered by the Parent Guarantor and, assuming due authorization, execution and delivery by the Seller and the Purchaser, constitutes a legal, valid and binding obligation of the Parent Guarantor, enforceable against it in accordance with its terms, subject to limitations with respect to enforcement imposed by law in connection with bankruptcy or similar proceedings and to the extent that equitable remedies such as specific performance and injunction are in the discretion of the court from which they are sought.

(3) No Violation.

(a) Except for the consents, approvals and authorizations as are required to be obtained by the TSXV in connection with the transactions contemplated herein, the execution and delivery by the Parent Guarantor of this Agreement and each of the Acquisition Agreements to which it is a party do not, and the performance by the Parent Guarantor of this Agreement and each of the Acquisition Agreements to which it is a party will not, conflict with, or result in any violation of, or constitute a default (with or without notice or lapse of time, or both) under, or give rise to a right of, or result by its terms in the, termination, cancellation or acceleration of any obligation or the loss of a benefit under, or to increased, - 54 -

additional, accelerated or guaranteed rights or entitlements of any Person under, or create any obligation to make a payment to any other Person under, or the loss of, any properties or assets of the Parent Guarantor pursuant to (1) any Laws or Order, (2) any provision of the articles, bylaws or other charter documents of the Parent Guarantor, or (3) any material Contract to which the Parent Guarantor is a party, except, in the case of this subclause (3), as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Parent Guarantor.

- (b) Except for Regulatory Approvals, if applicable, and the consents, approvals and authorizations as are required to be obtained by the TSXV in connection with the transactions contemplated herein, the Parent Guarantor is not required to submit any notice, report or other filing with any Governmental Entity in connection with the execution, delivery or performance of this Agreement or the consummation of the transactions contemplated hereby and no waiver, consent, approval or authorization of any Governmental Entity is required to be obtained or made by the Parent Guarantor in connection with its execution, delivery or performance of this Agreement or the consummation of the transactions contemplated hereby and no waiver, consent, approval or authorization of any Governmental Entity is required to be obtained or made by the Parent Guarantor in connection with its execution, delivery or performance of this Agreement or the consummation of the transactions contemplated hereby.
- (4) Litigation. There are no Legal Proceedings pending or, to the knowledge of the Parent Guarantor, threatened against the Parent Guarantor or any of its respective officers or directors (in such capacity) or affecting its business or assets that, if adversely determined, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Parent Guarantor. None of the Parent Guarantor, its business or assets is subject to any Order of any Governmental Entity that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect on the Parent Guarantor.

(5) Parent Shares.

- (a) The Parent Shares issued in exchange of the Purchaser Shares will have been duly and validly authorized and issued as fully paid and non-assessable equity of the Parent in accordance with applicable laws. The issuance of the Parent Shares (if and when issued) to the Seller will, subject to applicable law, be exempt from the registration and prospectus requirements of applicable securities Laws.
- (b) The Parent Guarantor is a "reporting issuer" or equivalent in each of British Columbia, Alberta, Ontario and Quebec within the meaning of applicable securities Laws and is not in default of any requirement of applicable securities Laws in any material respect.
- (c) The issued and outstanding Parent Shares are listed and posted for trading on the TSXV and the Parent Guarantor is in compliance with the rules and regulations of the TSXV in all material respects.
- (d) The Parent Guarantor's authorized share capital consist of an unlimited number of common shares, of which 23,810,626 are issued and outstanding as at the

date hereof, and 100,000,000 preferred shares without par value, of which none are issued and outstanding as at the date hereof.

(6) Corporate Matters.

- (a) Other than for the its ownership in the Purchaser, the Parent Guarantor does not own securities in any Persons.
- (b) The Parent Guarantor is not, directly or indirectly through the Subsidiaries, a member, partner or participant in any partnership, joint venture or syndicate where the joint liability that could arise from such membership, partnership or participation would reasonably be expected to have a Material Adverse Effect.
- (c) Except as contemplated in this Agreement and as otherwise disclosed in the Public Record, no Person or firm, as of the date hereof, has any agreement or option with the Parent Guarantor, or any right or privilege (whether preemptive or contractual) capable of becoming an agreement or option with the Parent Guarantor, for the purchase, subscription or issuance of any Parent Shares.
- (d) Computershare Investor Services Inc., at its principal office located in the city of Toronto, has been duly appointed as registrar and transfer agent for the Parent Shares in Canada.
- (e) Neither the Parent Guarantor nor any of the Subsidiaries is in violation of its constating documents; and neither the Parent Guarantor nor the any of the Subsidiaries is in default in the performance or observation of any obligation, agreement, covenant, or condition contained in any contract, indenture, mortgage, loan agreement, note or other instrument to which it is a party or by which it may be bound or to which any of its properties or assets is subject which default, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect.

(7) Securities Matters.

- (a) In particular, without limiting the foregoing, the Parent Guarantor is in compliance, in all material respects, with all of its applicable continuous disclosure obligations under applicable securities Laws and no material change relating to the Parent Guarantor has occurred with respect to which the requisite material change report has not been filed on a non-confidential basis with all relevant securities regulatory authorities (unless originally filed on a confidential basis and subsequently made non-confidential). The information and statements set forth in the Public Record, as of the respective applicable dates of such information and statements, did not contain any misrepresentation.
- (b) No securities commission or similar regulatory authority or the TSXV has issued any order which is currently outstanding preventing or suspending trading in any securities of the Parent Guarantor, and no such proceeding is, to the knowledge of the Parent Guarantor, pending, contemplated or threatened.
- (8) **Financial and Tax Matters.**

- (a) The Parent Financial Statements have been prepared in accordance with International Financial Reporting Standards ("IFRS") (and on a basis consistent throughout the periods indicated), and present fairly in all material respects, the financial position of the Parent Guarantor as of the dates indicated and the results of operations and the changes in financial position of the Parent Guarantor for the periods specified.
- (b) The Parent Guarantor had a reasonable basis for disclosing all forward-looking information contained in the Public Record.
- (c) Other than as disclosed in the Financial Information, there are no material off-balance sheet transactions, arrangements, obligations (including contingent obligations) or other relationships of the Parent Guarantor or the Subsidiaries with unconsolidated entities or other persons that would reasonably be expected to have a Material Adverse Effect.
- (d) None of the Parent Guarantor or any of the Subsidiaries has, or on the Closing Date will have, incurred any material liabilities or obligations, whether accrued, contingent or otherwise, of the type required to be reflected on a balance sheet prepared in accordance with IFRS, except for liabilities or obligations: (i) that occurred in the Ordinary Course of business, (ii) reflected in or reserved against in the Parent Financial Statements, or (iii) disclosed in the Public Record.
- (e) The Parent Guarantor maintains a system of internal control over financial reporting which is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of Parent Financial Statements for external purposes in accordance with IFRS.
- (f) The Parent Guarantor maintains a system of internal accounting controls sufficient to provide reasonable assurances that: (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary to permit preparation of Parent Financial Statements in conformity with IFRS and to maintain accountability for assets; (iii) access to its assets is permitted only in accordance with management's general or specific authorization; (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to differences; and (v) material information relating to it and the Subsidiaries (taken as a whole) is made known to those within the Parent Guarantor or the Subsidiaries responsible for the preparation of the Parent Financial Statements during the period in which the Parent Financial Statements have been prepared and that such material information is disclosed to the public within the time periods required by applicable securities Laws; except as disclosed in the Public Record, (A) the Parent Guarantor is not aware of any material weakness in the Parent Guarantor's internal control over financial reporting which would be required to be disclosed in a certificate issued pursuant to NI-52-109, and (B) since the date of the most recent Parent Financial Statements, there has been no change in the Parent Guarantor's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Parent Guarantor's internal control over financial reporting.

- (g) Except for the transactions contemplated in this Agreement or as disclosed in the Public Record, the Parent Guarantor (i) has not made any acquisition that is a "significant acquisition" within the meaning of NI 51-102 in its current financial year, and (ii) does not currently propose to make an acquisition that has progressed to a state where a reasonable person would believe that the likelihood of the acquisition being completed is high, and that would currently be a "significant acquisition" within the meaning of NI 51-102.
- (h) The Parent Guarantor and the Subsidiaries have duly filed all Tax returns required to be filed by it and has paid all material Taxes, assessments, re-assessments and all governmental charges, penalties, interest and other fines related thereto due and payable by it, and has properly withheld or collected and remitted all amounts required to be withheld or collected and remitted by it in respect of any governmental charges, and adequate provision has been made for material Taxes payable for any completed fiscal period for which Tax returns are not yet required and there are no agreements, waivers or other arrangements providing for an extension of time with respect to the filing of any Tax return or payment of any Tax, governmental charge or deficiency by the Parent Guarantor or the Subsidiaries, and to the knowledge of the Parent Guarantor, there are no actions, suits, proceedings, investigations or claims threatened or pending against the Parent Guarantor or the Subsidiaries in respect of Taxes, governmental charges, assessments or any matters under discussion with any Governmental Entity relating to Taxes, governmental charges, assessments or reassessments or reassessments asserted by any such authority, in each case, with such exceptions as would not result in a Material Adverse Effect.
- (i) There is no Tax deficiency which has been asserted against the Parent Guarantor or the Subsidiaries which would have a Material Adverse Effect, and all material Tax liabilities are adequately provided for in accordance with IFRS in the Parent Financial Statements for all periods up to December 31, 2020.

(9) Legal and Compliance Matters.

- (a) Except as otherwise disclosed in the Public Record, to the knowledge of the Parent Guarantor, each of the Parent Guarantor and the Subsidiaries has conducted and is conducting its business in compliance with all applicable laws, rules and regulations of each jurisdiction in which its business is carried on, except for non-compliance that would not individually or in the aggregate reasonably be expected to have a Material Adverse Effect.
- (b) None of the Parent Guarantor, the Subsidiaries or, to the knowledge of the Parent Guarantor, any director, officer, agent, employee or affiliate of the Parent Guarantor or any of the Subsidiaries is currently the subject or the target of any sanctions administered or enforced by Governmental Entity (collectively, Sanctions), nor is the Parent Guarantor located, organized or resident in a country or territory that is the subject of Sanctions.
- (c) The operations of the Parent Guarantor and the Subsidiaries are and have been conducted at all times in material compliance with the requirements of applicable

anti-money laundering laws, including, applicable financial recordkeeping and reporting requirements of the Proceeds of Crime (Money Laundering) and Terrorist Financing Act (Canada), Part II.1 of the Criminal Code (Canada) and, in each case, the rules and regulations promulgated thereunder, and the anti-money laundering laws of various jurisdictions where the Parent Guarantor and the Subsidiaries conduct business (collectively, the "Anti-Money Laundering Laws") and no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Parent Guarantor or the Subsidiaries with respect to the Anti-Money Laundering Laws is pending or, to the knowledge of the Parent Guarantor, threatened.

- (10) **No Finder's Fee.** The Parent Guarantor has not taken, and agrees that the Parent Guarantor will not take, any action that would cause the Seller to become liable to any claim or demand for a brokerage commission, finder's fee or other similar payment.
- (11) Investment Canada Act. The Parent Guarantor is a Canadian within the meaning of the Investment Canada Act.

Section 5.2 Disclaimer of Other Representations and Warranties.

The representations and warranties set forth in this Article 5 are the only representations and warranties made by the Parent Guarantor relating to the transactions contemplated by this Agreement. Except as specifically set forth in this Article 5, the Parent Guarantor makes no other representation or warranty relating to the transactions contemplated by this Agreement. In connection with the transactions contemplated hereby, the Seller has been represented by, and consulted with, legal counsel of its choice and the Seller and such counsel have read this Agreement and have been given time to consider this Agreement, understand this Agreement, and, after such consideration and with such understanding, the Seller has knowingly, freely and without coercion entered into this Agreement and, in particular, this Section 5.2 and Section 10.9.

Article 6 COVENANTS

Section 6.1 Conduct of Operations Prior to Closing.

- (1) During the Interim Period, and except (i) with the prior written consent of the Purchaser, such consent not to be unreasonably withheld, delayed or conditioned, or (ii) as otherwise expressly set forth in this Agreement, the Seller will:
 - (a) manage the Purchased Assets, conduct the Operations and maintain the Books and Records in the Ordinary Course;
 - (b) use all reasonable commercial efforts to, in the Ordinary Course, maintain Inventory levels of the Operations until the Closing Date;
 - (c) promptly advise the Purchaser of any Material Adverse Effect;
 - (d) promptly advise the Purchaser of any Environmental Liabilities that arise or become known during the Interim Period;

- (e) promptly advise the Purchaser of:
 - (i) its intention to make any or of any material change in the present business organization of the Operations;
 - (ii) its intention to make any or of any material change in any of its relationships with customers, suppliers and others having business dealings with it;
 - (iii) its intention to make any or of any impairment of the Operations in any material respect;
 - (iv) any instance of a customer of the Operations terminating, or communicating to the Seller the intention or threat to terminate, its relationship with the Seller or the Operations, or the intention to reduce substantially the quantity of products or services it purchases from the Seller or the Operations, or its dissatisfaction with the products or services sold by the Seller or the Operations, except in the case of customers whose purchases from the Operations are not, in the aggregate, material to the Operations;
 - (v) any shift reduction, temporary or permanent shutdown, termination, discontinuance, closure or disposal of any Lumber Mill, the Newsprint Mill or any material portion of the Woodland Operations; and
- (f) continue, in full force and effect, all Insurance Policies and coverage currently maintained on the Purchased Assets to and including until the Closing Date.
- (2) Except as contemplated by this Agreement, or otherwise consented to in writing by the Purchaser, the Seller covenants not to do or permit to be done any of the following, from the date of this Agreement until Closing:
 - (a) except as required by applicable Law, pursuant to any Collective Bargaining Agreements in accordance with their terms as at the date of this Agreement, in the Ordinary Course: (i) grant or commit to any employee any increase in compensation (other than routine increases in compensation consistent with past practice with respect to Non-Unionized Employees not exceeding \$500,000 in aggregate); (ii) grant or commit to any Employee any increase in severance or termination payments, or enter into new, or amend existing agreements respecting employment, compensation or termination (including Benefit Plans, bonuses, incentive compensation, non-pension employee benefits, deferred compensation, insurance, profit sharing, and other forms of compensation or benefits) with such a Person, including pursuant to any amendment, replacement or extension of the Collective Bargaining Agreements, provided that nothing in this Agreement shall preclude the Seller from terminating an Employee, from paying annual bonuses and incentive compensation to Non-Unionized Employees in the Ordinary Course;
 - (b) sell, transfer or otherwise dispose of, or agree to sell, transfer, pledge, lease, encumber or otherwise dispose of, any Purchased Assets other than obsolete or

surplus assets or Inventory which Inventory may be disposed of in the Ordinary Course;

- (c) enter into any agreement which obligates the Seller to pay more than \$350,000 in the aggregate and that cannot be terminated without cost or liability to the Operations on not more than 90 days' notice;
- (d) relinquish any rights of material value to the Operations with respect to any Contracts, Real Property Leases or Permits or terminate or make a material modification or amendment to any of the Contracts (including the Collective Bargaining Agreements), Real Property Leases or any of the Permits to the extent such Contracts, Real Property Leases or Permits are of material value to the Operations;
- (e) make any changes to the existing accounting practices related to the Operations, except as required by applicable Law, a change in GAAP or as otherwise provided herein; and
- (f) directly or indirectly solicit, initiate, or encourage the submission of any proposal or offer from any Person relating to the acquisition of any portion of the Purchased Assets (other than sales of Inventory in the Ordinary Course) or directly or indirectly participate in any discussions or negotiations regarding, furnish any information with respect to, assist or participate in, or facilitate in any other manner any effort or attempt by any Person to do or seek any of the foregoing. The Seller will notify the Purchaser immediately if any Person makes any proposal, offer, inquiry, or contact with respect to any of the foregoing.

Section 6.2 Actions to Satisfy Closing Conditions.

- (1) Subject to applicable Laws and the specific obligations in Section 6.4, Section 6.5 and Section 6.6, the Seller shall take all reasonable actions that are within its power and control and shall use its commercially reasonable efforts to cause other actions to be taken which are not within its power and control, so as to ensure compliance with all of the conditions set forth in Section 7.1.
- (2) Subject to applicable Laws and the specific obligations set out in Section 6.5, the Purchaser shall take all reasonable actions that are within its power and control and shall use its commercially reasonable efforts to cause other actions to be taken which are not within its power and control, so as to ensure compliance with all of the conditions set forth in Section 7.2.

Section 6.3 Confidentiality.

The Parent Guarantor acknowledges that certain directors and officers thereof have entered into a confidentiality with Rayonier Advanced Materials Inc. dated the 14th day of August, 2020 (the "**Confidentiality Agreement**"). The Parent Guarantor, the Purchaser and Seller agree that the Confidentiality Agreement shall apply to them as if they were original signatories thereto and each are bound by its terms. Upon Closing, the Confidentiality Agreement shall terminate with respect to the Operations, but shall remain in full force and effect to the extent of any confidential information and other obligations that do not relate to the Operations. If the Closing

does not occur, the Confidentiality Agreement shall remain in effect in accordance with and subject to its terms.

Section 6.4 Request for Consents.

Subject to applicable Laws and Section 6.5, reasonably promptly following the execution of this Agreement, the Seller and the Purchaser shall use their commercially reasonable efforts to obtain or cause to be obtained, prior to Closing:

- (1) all consents, approvals and waivers described in Section 7.1(5) of the Disclosure Letter;
- (2) with respect to Tenures in Ontario, either (i) the consent of the Ontario Ministry of Natural Resources to the transfer of the Tenures granted by the Ontario Government, or (ii) written comfort acceptable to the Purchaser, acting reasonably, from the Ontario Ministry of Natural Resources of its intention to issue a consent to the transfer of the Tenures granted by the Ontario Government. For the purposes of this Section 6.4(2), "Tenures" shall mean all or substantially all of the Tenures, for which purpose "substantially all of the Tenures" shall mean all Tenures, other than Tenures with respect to which the failure to receive such consent or written comfort referred to in clauses (i) and (ii) above would not have the effect of reducing, impairing, suspending or terminating (other than in a *de minimis* manner) any rights or privileges enjoyed by the Seller with respect to the Operations as of the date of this Agreement; and
- (3) with respect to the Supply Guarantees, either (i) the consent of the *Ministère des Forêts, de la Faune et des Parcs* of Québec to the transfer or reissuance of the Supply Guarantees or (ii) written comfort acceptable to the Purchaser, acting reasonably, from the *Ministère des Forêts, de la Faune et des Parcs* of Québec of its intention to issue a consent to the transfer of the Supply Guarantees.

Section 6.5 Authorizations and Regulatory Approvals.

- (1) The Purchaser shall file promptly (and in any event, within ten (10) days) after the date of this Agreement a request for an Advance Ruling Certificate and, in the alternative, a No-Action Letter with the Commissioner of Competition and, unless external legal counsel to the Parties otherwise agree on behalf of the Parties, the Parties shall concurrently file premerger notification filings under Part IX of the Competition Act with the Commissioner of Competition. The Parties shall cooperate and use their commercially reasonable efforts in connection with the foregoing.
- (2) Each Party shall make an appropriate filing, if necessary, pursuant to the HSR Act with respect to the transactions contemplated by this Agreement promptly (and in any event, within ten (10) days) after the date of this Agreement (which filing shall request "early termination" of any applicable waiting period).
- (3) Each Party shall respond as promptly as practicable in the event that a Governmental Antitrust Entity issues a request for additional information and documentary material that may be requested pursuant to the Competition Act or pursuant to the HSR Act.
- (4) Without limiting the foregoing, each of the Seller and the Purchaser, as promptly as practicable after the execution of this Agreement, shall take, and cause to be taken, all

commercially reasonable actions and efforts and do, and cause to be done, all commercially reasonable things and promptly execute and file, or join in the execution and filing of, any application, filing or other document that may be necessary, proper or advisable to permit and diligently pursue the completion of the transactions contemplated by this Agreement and the performance of their obligations hereunder in accordance with the terms hereof, including obtaining the Regulatory Approvals and all Authorizations, consents and Orders which are required by Law from all Governmental Entities or other Persons, or which the Seller or the Purchaser may reasonably request, in connection with the consummation of the transactions contemplated by this Agreement and the performance of its obligations hereunder, and shall cooperate with each other in connection therewith and provide such other information to any Governmental Entity or Person as such Governmental Entity or Person may reasonably request in connection therewith.

- (5) All filing fees (including any Taxes thereon) in respect of any filing made to any Governmental Entity in respect of any Regulatory Approvals shall be shared equally between the Seller and the Purchaser.
- (6) With respect to obtaining the Regulatory Approvals, each of the Seller and the Purchaser shall:
 - (a) not extend or consent to any extension of any applicable waiting or review period or enter into any agreement with a Governmental Entity to not consummate the transactions contemplated by this Agreement, except upon the prior consent of the other Party;
 - (b) promptly notify the other Party of written or oral communications of any substantive nature from a Governmental Entity relating to any Regulatory Approval and provide the other Party with copies thereof and provide copies of all notices and correspondence received from any Governmental Entity;
 - (c) respond as promptly as reasonably possible to any inquiries or requests received from a Governmental Entity in respect of any Regulatory Approval;
 - (d) use their commercially reasonable efforts to cooperate with and assist the other Party in the preparation and making of all such filings, applications and submissions; and
 - (e) not participate in any substantive meeting or discussion (whether in person, by phone or otherwise) with a Governmental Entity in respect of any Regulatory Approval unless it consults with the other Party in advance and gives the other Party the reasonable opportunity to attend and participate to the extent permitted by such Governmental Entity.
- (7) A Party will not be required to provide information in its possession to the other Party, even if that information may be required in connection with the filings referred to in this Section 6.5, if that Party, acting reasonably, determines that the information is material and commercially or competitively sensitive or that providing the information could reasonably be expected to have a Material Adverse Effect on that Party if the transactions contemplated by this Agreement are not completed. In any such case, the

- 63 -

Parties shall provide the information to external counsel for the other Party on an "external counsel only" basis. Such information shall be provided only to the external counsel of the other Party and shall not be disclosed by such external counsel to directors, officers, employees or other agents of that Party, unless express prior written permission is obtained from the source of the materials.

- (8) If any objections are asserted with respect to the transactions contemplated hereby under any applicable Laws or if any suit, application, challenge, injunction, action or Proceeding, is commenced by a Governmental Entity in connection with the transactions contemplated hereby, each Party shall use commercially reasonable efforts to: (i) oppose or defend against any objections, suit, application, challenge, action or Proceeding to prevent or enjoin consummation of this Agreement (and the transactions contemplated herein), and/or (ii) take such action as reasonably necessary to overturn any regulatory action by any Governmental Entity to block consummation of this Agreement (and the transactions contemplated herein), application, challenge, action or Proceeding brought by any Governmental Entity in order to avoid the entry of, or to have vacated, overturned or terminated, including by appeal if necessary, in order to resolve any such objections or challenge as such Governmental Entity may have to the transactions contemplated by this Agreement so as to permit consummation of such transactions.
- (9) The Purchaser shall not, and shall cause its Affiliates and ultimate parent entities not to, acquire or agree to acquire, by merging with or into or consolidating with, or by purchasing a portion of the assets of or equity in, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof, or otherwise acquire or agree to acquire any assets or equity interests, if the entering into of a definitive agreement relating to, or the consummation of such acquisition, merger or consolidation would reasonably be expected to (i) impose any delay in the obtaining of, or increase the risk of not obtaining, any consents of any Governmental Entity necessary to consummate the transactions contemplated by this Agreement or the expiration or termination of any applicable waiting period; (ii) increase the risk of any Governmental Entity seeking or entering an order prohibiting the consummation of the transactions contemplated by this Agreement; (iii) increase the risk of not being able to remove any such order on appeal or otherwise; or (iv) delay or prevent the consummation of the transactions contemplated by this Agreement.
- (10) The Purchaser shall not make the notification filing under the Investment Canada Act until after the Closing Date.

Section 6.6 Financing.

The Seller shall use commercially reasonable efforts to cause its Affiliates, shareholders, directors, officers, employees and advisors (including financial advisors and legal counsel) to provide to the Purchaser or Parent Guarantor any co-operation reasonably requested by the Purchaser or the Parent Guarantor in connection with the Financing, including:

 (a) furnishing the Purchaser or the Parent Guarantor and the Financing Sources, as promptly as reasonably practicable, with financial and other pertinent information regarding the Operations as may reasonably be requested in writing by the Purchaser or the Parent Guarantor;
 (b) upon reasonable notice and in reasonably

- 64 -

convenient locations, making senior management of the Operations available to participate in a reasonable number of meetings, presentations and other sessions with prospective lenders, investors and rating agencies in connection with the Financing; (c) assisting with the preparation of customary materials for rating agency presentations (and assisting in the obtaining of corporate, credit and facility ratings from ratings agencies), offering documents, private placement memoranda, bank information memoranda, prospectuses and all other materials to be used in connection with the Financing (including customary authorization and management representation letters) and all documentation and other information required in connection with applicable "know your customer" and anti-money laundering rules and regulations; (d) taking all corporate actions, subject to and only effective upon the occurrence of the Closing, required to permit the consummation of the Financing and to permit the proceeds thereof to be made available to the Purchaser or the Parent Guarantor, and (e) otherwise taking actions within its control to co-operate in satisfying the conditions precedent set forth in any definitive document related to the Financing.

- (2) Notwithstanding anything to the contrary contained in this Agreement, the Seller hereby waives any rights or claims against any Financing Source in connection with this Agreement, the Financing or in respect of any other document or theory of law or equity (whether in tort, contract or otherwise) or in respect of any oral representations made or alleged to be made in connection herewith or therewith and the Seller agrees not to commence any action or proceeding against any Financing Source in connection with this Agreement, the Financing or in respect of any other document or theory of law or equity in connection therewith and agrees to cause any such action or proceeding asserted by the Seller in violation of the prohibition on commencing actions or proceedings contained in this Section 6.6(2) and in connection with this Agreement, the Financing or in respect of any other document or theory of law or equity in connection therewith against any Financing Source to be dismissed or otherwise terminated.
- Seller shall, with the assistance of its auditors, assist the Parent Guarantor and Purchaser in the preparation of such (3)prospectus (including any required annexes, schedules or exhibits thereto). Without limiting the generality of the foregoing, the Seller shall: (i) cause its Affiliates, directors, officers, employees and advisors (including auditors, financial advisors and legal counsel) to provide the Parent Guarantor and Purchaser with all information regarding the Operations, Purchased Assets, Assumed Obligations required to be included in the prospectus, (ii) respond, and assist in the response, to requests from the applicable securities regulator in connection with the prospectus and (iii) prepare the Carve-out Financial Statements for inclusion in the prospectus. Seller shall obtain any necessary consents from any of its auditors and other advisors to the use of any financial, technical or other expert information required to be included in the prospectus (including the Carve-out Financial Statements). Seller shall ensure that information included in the prospectus and Carve-out Financial Statements is complete and accurate in all material respects, complies in all material respects with applicable Laws and, without limiting the generality of the foregoing, does not include any misrepresentation concerning the Seller, the Purchased Assets and the Seller's financial condition, the material terms of the Purchased Assets, or omit a fact necessary to make a statement not misleading. The Seller shall promptly notify the Seller if it becomes aware of any misrepresentation (as defined under applicable securities Laws) contained in the prospectus.
Section 6.7 Access.

- (1) Subject to Section 6.8, to applicable Laws, and to any confidentiality restrictions binding the Seller, the Seller shall, during the Interim Period, make available to the Purchaser and its authorized representatives (including the Debt Financing Sources) all title documents, Contracts, Real Property Leases, financial statements, policies, reports, licences, orders, permits, books of account, Tax Returns, accounting records and other documents, information and data relating exclusively to the Operations in its possession. During the Interim Period the Seller shall permit the Purchaser and its representatives to have access to Owned Real Property, Leased Real Property and the management team or such other personnel as may be reasonably by the Purchaser. At the request of the Purchaser, the Seller will execute or cause to be executed, such consents, authorizations and directions as may be necessary to enable the Purchaser and its representatives to obtain access to all files and records maintained by Governmental Entities or other Persons in respect of the Purchased Assets and the Operations.
- (2) The exercise of any rights by or on behalf of the Purchaser under this Section 6.7 shall be carried out (i) strictly for purposes of transitional planning, (ii) during normal business hours in such manner as not to interfere unduly with the normal operation of the Operations and upon the reasonable prior written request to the Seller, (iii) under the supervision of the Seller's personnel and (iv) in compliance with the Seller's internal standards and policies. All requests by the Purchaser for access pursuant to this Section 6.7 shall be submitted or directed exclusively to either [**] or such other individuals as the Seller may designate in writing from time to time.

Section 6.8 Contact with Customers, Suppliers, Employees and Other Operations Relations.

During the Interim Period, the Purchaser shall not contact any customers, suppliers, employees, Governmental Entities (other than a Governmental Antitrust Entity) or any other business relations of the Seller without the prior written consent of the Seller.

Section 6.9 Notice of Untrue Representation or Warranty.

During the Interim Period, the Seller shall promptly notify the Purchaser, and the Purchaser shall promptly notify the Seller, upon any representation or warranty made by it and contained in this Agreement becoming untrue or incorrect in material respect. Any such notification must set out particulars of the untrue or incorrect representation or warranty in material respect and details of any actions taken to rectify the matters.

Section 6.10 Transfer Taxes.

(1) The Purchaser will be liable for and will pay, or cause to be paid, if and as applicable, all Transfer Taxes payable under any applicable Law on or with respect to the purchase by the Purchaser of the Purchased Assets under this Agreement no later than five (5) days prior to the time such Transfer Taxes must be remitted by the Seller to the appropriate Governmental Entity, as applicable. All amounts referred to in this Agreement are exclusive of any applicable Transfer Taxes which shall be payable in addition to the amounts referred to herein, unless otherwise specifically referred to.

- (2) Notwithstanding the foregoing, the Purchaser and the Seller shall apply to use Revenu Québec's GST/QST "Fast Track" program, and shall file, jointly and separately, as applicable, all Tax returns, forms and documents as required by Revenu Québec, for purposes of obtaining the expedited GST and QST refunds of the Purchaser in connection with the supply of the Purchased Assets (other than the Owned Real Property and the Leased Real Property for which the GST and QST treatment is further set out in Section 6.10(4)). To the extent that Revenu Québec agrees and is authorized to pay or credit the refund owing to the Purchaser directly to the Seller on account of GST and QST payable in connection with the supply of the Purchased Assets and in accordance with the terms of the "Fast Track" program, the Purchaser shall be relieved of its obligation to pay GST and QST to the Seller under this Section 6.10. Notwithstanding the foregoing, the Purchaser shall be responsible to remit and pay to the Seller all GST and QST otherwise due which will not have been directly credited or paid by Revenu Québec to the Seller as a result of the reduction of the expected net GST and QST refund of the Purchaser. For greater certainty, the Purchaser shall remain liable and responsible for any claim or refusal of input tax credits or input tax refunds by Revenu Québec or a similar Governmental Entity in connection with the supply of the Purchased Assets contemplated under this Agreement and of any implications or impacts of such claim or refusal on the "Fast Track" program and process described in this Section 6.10(1).
- (3) If, despite the application of the Purchaser and the Seller to use Revenu Québec's GST/QST "Fast Track" program, the Seller is notified that GST or QST, or any interest or penalties in respect thereof, are required to be paid by and are imposed upon the Seller with respect to the purchase by the Purchaser of the Purchased Assets, the Purchaser will reimburse, or cause to be reimbursed, to the Seller, such GST or QST and any applicable interest and penalties within ten (10) Business Days of the payment thereof by the Seller, except to the extent that such GST, QST, interest or penalties arose solely on account of any gross negligence or wilful misconduct of the Seller.
- (4) Notwithstanding Section 6.10(1) of this Agreement, the Seller shall not collect GST and QST in connection with the transfer, sale, conveyance and assignment of rights of the Owned Real Property and to the Leased Real Property contemplated under this Agreement if the Purchaser is and remains registered for purposes of the GST and QST as of the Closing and duly provides and delivers a GST and QST certificate and indemnity in the form and substance attached hereto as Exhibit E and in accordance with Section 7.2 of this Agreement (the "GST and QST Certificate").

Section 6.11 Cooperation on Tax Matters.

The Seller and the Purchaser will furnish or cause to be furnished to each other, each at its own expense, as promptly as practicable, such reasonable information and assistance, and provide additional information and explanations of any material provided, relating to the Purchased Assets as is reasonably necessary for the filing of any Tax Returns, for the preparation of any audit, for the prosecution or defence of any Claim relating to any adjustment or proposed adjustment with respect to Taxes, with respect to the application to use Revenu Québec's GST/QST "Fast Track" program, and to obtain or benefit from any Tax refund, credit or exemption.

Section 6.12 TSXV Required Information.

- (1) Parent Guarantor and Purchaser shall use commercially reasonable efforts to prepare, as promptly as practicable after the date of this Agreement, any documents required to be prepared or filed under applicable securities Laws and the policies of the TSXV in connection with the transactions contemplated herein.
- (2) Seller shall use its commercially reasonable efforts to assist the Parent Guarantor with any filing or other disclosure required by the TSXV.

Section 6.13 Treatment of Purchaser Shares and Parent Shares

RYAM GP hereby agrees not to sell, transfer, assign or dispose of any Purchaser Shares or any Parent Shares underlying such Purchaser Shares for a period of six (6) months following the Closing. Notwithstanding the previous sentence, the Parent Shares shall be freely tradable (subject to any restrictions under applicable securities law) prior to the end of such six month period if any of the following event is completed by the Parent Guarantor, the Purchaser and/or one or more of its Affiliates: (i) any plan of arrangement, merger, amalgamation, consolidation, share exchange, business combination, reorganization, recapitalization, liquidation, dissolution, winding up or exclusive license involving the Operations, the Parent Guarantor, the Purchaser or any of its Affiliates which own directly or indirectly the Operations (ii) any direct or indirect sale or disposition in a single transaction or a series of transactions, of assets representing 50% or more of the assets comprising the Operations (or rights or interests therein or thereto); (iii) any direct or indirect take-over bid, tender offer, exchange offer, treasury issuance or other transaction that, if consummated, would result in such person or group of persons beneficially owning more than 50% of any class of voting or equity securities of the Parent Guarantor, the Purchaser or any of its Affiliates which own directly or indirect the Operations (or securities convertible into or exchangeable or exercisable for such voting or equity securities assuming, if applicable, the conversion, exchange or exercise of such securities convertible into or exchangeable or exercisable for such voting or equity securities) provided that in the case of a direct or indirect take-over bid, tender offer or exchange offer until such direct or indirect take-over bid, tender offer or exchange offer, is consummated, the release from the restrictions provide in this Section 6.13 shall solely be for the purpose of tendering to such take-over bid, tender offer or exchange offer; or (iv) any other similar transaction or series of transactions involving the Parent Guarantor, the Purchaser or any of its Affiliates which own directly or indirectly the Operations.

Section 6.14 Commitment to Financing

The Parent Guarantor and the Purchaser covenant to deploy their best efforts, or to cause to be completed, the Financing as soon as practicable following the execution of this Agreement.

Article 7 CONDITIONS OF CLOSING

Section 7.1 Conditions for the Benefit of the Purchaser.

The obligation of the Purchaser to complete the purchase of the Purchased Assets is subject to the following conditions being satisfied at or prior to Closing, which conditions are for the

exclusive benefit of the Purchaser and may be waived, in whole or in part, by the Purchaser in its sole discretion:

- (1) Truth of Representations and Warranties. (i) The representations and warranties contained in Section 3.1(1), Section 3.1(2), Section 3.1(7)(a)(2), and Section 3.1(31)(a) and (b) (to the extent it relates to the Operations, Owned Real Property, or the Leased Real Property) (the "Fundamental Representations"), of this Agreement shall be true and correct in all respects as of the Closing Date (except to the extent of any *de minimis* inaccuracy), and (ii) the other representations and warranties of the Seller contained in this Agreement shall be true and correct as of the Closing Date, in all cases, except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on the Operations, provided, however, that (A) if a representation and warranty is qualified by a materiality or Material Adverse Effect qualification, such qualification shall be disregarded for the purposes of this Section 7.1(1)(ii), and (B) if a representation and warranty speaks only as of a specific date, it only needs to be true and correct as of that date and the Seller will have delivered a certificate executed by a senior officer to that effect.
- (2) **Performance of Covenants**. The Seller shall have fulfilled or complied, in all material respects, with all covenants contained in this Agreement to be fulfilled or complied with by it at or prior to the Closing, and the Seller will have delivered a certificate executed by a senior officer to that effect.
- (3) **Regulatory Approvals**. The Competition Act Clearance, if required, the HSR Act Clearance shall have been obtained.
- (4) TSXV Approvals. All consents, approvals and authorizations as are required to be obtained from the TSXV in connection with the transactions contemplated herein or necessary to complete the transactions contemplated herein, including with respect to the issuance of the Consideration Shares, shall have been obtained, subject only to ordinary conditions typical for the issuance and listing of common shares.
- (5) **Consents.** The Seller shall have delivered to the Purchaser consents or waivers set forth in Section 7.1(5) of the Disclosure Letter with respect to the transactions contemplated hereby, in each case in form and substance reasonably acceptable to the Purchaser; provided that, if such consent shall not be given solely as a result of a change in Law after the date hereof, this condition shall be deemed satisfied.
- (6) **Material Adverse Effect**. There shall have been no Material Adverse Effect in respect of the Operations between the date of this Agreement and the Closing Date.
- (7) **Allocation of Purchase Price**. The final allocation of the Purchase Price shall have been agreed to by the Seller and the Purchaser in accordance with Section 2.4(1).
- (8) **Deliveries**. The Seller shall deliver or cause to be delivered to the Purchaser the following in form and substance satisfactory to the Purchaser acting reasonably:
 - (a) certified copies of (i) the articles and by-laws of the Seller, and (ii) certified resolutions of the board of directors of the Seller approving, as applicable, the

entering into and completion of the transactions contemplated by this Agreement (including, without limitation, the transfer of the Purchased Assets to the Purchaser);

- (b) all keys, entry devices and passcodes with respect to the Purchased Assets (including the Owned Real Property and the Leased Real Property), including combinations to any locks or vaults;
- (c) all plans and specifications in the Seller's possession or under its control relating to the plant, buildings, structures, erections, improvements, appurtenances and fixtures situate on or forming part of the Owned Real Property, including all such electrical, mechanical and structural drawings related thereto as are in the possession or under the control of the Seller;
- (d) all data bases recorded or stored by means of any device, including in electronic form (but in such case, only to the extent technically possible), title documents, abstracts of title, deeds, surveys, leases, certificates of trademarks and copyrights, Contracts and commitments in the possession or under the control of the Seller relating to the Operations;
- (e) all records of past sales, customer lists, supplier lists, payroll records, Inventory data, Inventory master records data in the possession or under the control of the Seller relating to the Operations;
- (f) all deeds, conveyances, bills of sale, discharges, assurances, transfers, assignments required to transfer the Purchased Assets to the Purchaser in accordance with this Agreement, and in the case of the Owned Real Property (i) located in Ontario, with the statements relating to the *Planning Act* (Ontario) completed by the Seller and Seller Counsel and (ii) located in Quebec, a deed of sale to be registered at the applicable land registry office;
- (g) a duly executed copy of each Acquisition Agreement, effective as of the Closing;
- (h) a certification (without personal liability) of an executive officer of the Seller to the effect that each of the conditions specified in Section 7.1(1) and Section 7.1(2) is satisfied in all respects;
- share certificates representing the SFL Shares endorsed in blank for transfer or accompanied by irrevocable stock transfer powers of attorney executed in blank, in either case, by the holders of record, together with evidence satisfactory to the Purchaser that the Purchaser has been entered in the corporate records of the SFL Holdcos as the holder of record of the SFL Shares;
- (j) an assignment, assumption and general conveyance agreement in the form to be agreed by the Seller and the Purchaser, each acting reasonably;
- (k) the Intellectual Property assignment agreements with Owned Intellectual Property in the form to be agreed by the Seller and the Purchaser, each acting reasonably;

- the original vehicle permit/ownership registration for each vehicle included in the Purchased Assets duly endorsed by Seller for transfer and all required safety and emissions certificates;
- (m) copies of the Books and Records of the Operations;
- (n) evidence, satisfactory to the Purchaser, of the release and discharge of the Liens specified in Section 7.1(8)(m) of the Disclosure Letter, or in respect of the Owned Real Property, (i) a personal undertaking of Seller Counsel to discharge same within a reasonable period of time following Closing so long as such undertaking conforms to generally accepted legal practice in the province in which such Owned Real Property is located, (ii) a pay-out letter from the holder of any such Lien, which pay-out letter is to be satisfactory to the Purchaser in form and substance, acting reasonably, and (iii) a payment direction by the Seller, satisfactory to the Purchaser in form and substance, acting reasonably, directing payment to the holder of such Lien from the Purchase Price;
- (o) a purchase certificate issued by the Ontario Workplace Safety and Insurance Board and the Commission des normes, de l'équité, de la santé et de la sécurité du travail in respect of Seller and the Operations confirming that as at the Closing Date, the relevant Governmental Entity has no Claim against the Seller in respect of any amounts payable under the relevant workers' compensation Law in respect of the Operations or any of the Purchased Assets; and
- (p) such other documents, agreements or instruments as reasonably necessary to give effect to the assignment of the Purchased Assets.
- (9) **No Legal Action**. No action, proceeding or Law is in effect that prohibits or enjoins the Purchaser or the Seller from consummating the transactions contemplated by this Agreement.

Section 7.2 Conditions for the Benefit of the Seller.

The obligation of the Seller to complete the sale of the Purchased Assets is subject to the following conditions being satisfied at or prior to Closing, which conditions are for the exclusive benefit of the Seller and may be waived, in whole or in part, by the Seller in its sole discretion:

(1) Truth of Representations and Warranties. (i) The representations and warranties contained in Section 4.1(1), Section 4.1(2), Section 4.1(3)(a)(2), Section 5.1(1), Section 5.1(2) and Section 5.1(3)(a) of this Agreement shall be true and correct in all respects as of the Closing Date (except to the extent of any de minimis inaccuracy), and (ii) the other representations and warranties of the Purchaser contained in this Agreement shall be true and correct as of the Closing Date, in all cases, except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on the Operations, provided, however, that (A) if a representation and warranty is qualified by a materiality or material adverse effect qualification, such qualification shall be disregarded for the purposes of this Section 7.2(1)(ii), and (B) if a representation and warranty speaks only as of a specific date, it only needs to be true and correct as of that date and the Purchaser will have delivered a certificate executed by a senior officer to that effect.

- (3) **Regulatory Approvals**. The Competition Act Clearance, if required, the HSR Act Clearance shall have been obtained.
- (4) TSXV Approvals. The conditional approval of the TSXV as are required to be obtained from the TSXV in connection with the transactions contemplated herein or necessary to complete the transactions contemplated herein shall have been obtained.
- (5) **Deliveries**. The Purchaser shall deliver or cause to be delivered to the Seller the following in form and substance satisfactory to the Seller, acting reasonably:
 - (a) certified copies of (i) the resolutions of the board of directors of the Purchaser approving the entering into and completion of the transactions contemplated by this Agreement, including the issuance of the Consideration Shares, and (ii) a list of the officers and directors of the Purchaser authorized to sign agreements together with their specimen signatures;
 - (b) a certificate of status, compliance, good standing or like certificate with respect to the Purchaser issued by the appropriate government official of the jurisdiction of its formation or incorporation, as the case may be;
 - (c) a duly executed copy of each Acquisition Agreement, effective as of the Closing;
 - (d) a certification (without personal liability) of an officer of the Purchaser to the effect that each of the conditions specified in Section 7.2(1) and Section 7.2(2) is satisfied in all respects;
 - (e) evidence of the consents, approvals or authorizations of the TSXV with respect to the transactions of the Parties contemplated herein or necessary to complete the transactions contemplated herein;
 - (f) a duly executed GST and QST Certificate in the form and substance attached hereto as Exhibit F;
 - (g) the share certificates, representing the Consideration Shares and the Parent Shares (or evidence thereof); and
 - (h) the Chip Purchase Setoff Note.
- (6) **No Legal Action**. No action, proceeding or Law is in effect that prohibits or enjoins the Purchaser or the Seller from consummating the transactions contemplated by this Agreement.

- 72 -

Article 8 CLOSING

Section 8.1 Date, Time and Place of Closing.

The completion of the transactions contemplated by this Agreement shall take place at the offices of McCarthy Tétrault LLP, 1000 De La Gauchetière Street West, Suite 2500, Montréal, Québec, at 10:00 AM (Montreal time) on the Closing Date or at such other place, on such other date and at such other time or solely by way of electronic means as may be agreed upon in writing between the Seller and the Purchaser, provided that such means conform to generally accepted legal practice in the province in which such Owned Real Property is located.

The Seller and Purchaser covenant and agree to cause their respective solicitors to enter into a document registration agreement (the "**DRA**") to govern the electronic submission of the transfer/deeds for the Owned Real Property (the "**Registration Documents**") to the applicable Land Registry Office. The DRA shall outline or establish, without limitation, the following:

- (1) the procedures and timing for completing all registrations in Ontario and in Québec electronically;
- (2) provide for all closing documents and closing funds to be held in escrow pending the submission of the Registration Documents to the Land Registry Office in Ontario and Québec and their acceptance by virtue of each Registration Document being assigned a registration number;
- (3) with respect to the Owned Real Property situated in Quebec, the Seller and Purchaser shall enter into a Deed of Sale in registrable and customary form and registration of the Deed of Sale shall have occurred upon confirmation of registration of the Deed of Sale on title to the Owned Real Property situated in Québec without adverse entries;
- (4) if there is a problem with the Teraview electronic registration system which does not allow the parties to electronically register all Registration Documents on Closing, the Closing Date shall be deemed to be extended until the next day when the said system is accessible and operating for the applicable Land Registry Office in Ontario applicable to the Owned Real Properties; and
- (5) upon the registration of the Registration Documents without adverse entries and in accordance with the DRA, the Purchase Price shall be immediately released to the Seller.

Section 8.2 Risk of Loss.

Until the Closing, the Purchased Assets will remain at the risk of the Seller. If, prior to the Closing, all or any part of the Purchased Assets which are necessary to carry on the Operations as currently conducted are destroyed or damaged by fire or any other casualty or shall be appropriated, expropriated or seized by a Governmental Entity, the Purchaser shall complete the purchase without reduction of the Purchase Price provided that the value of the Purchased Assets destroyed or damaged by fire or any other casualty or expropriated or seized by a Governmental Entity is of \$10,000,000 or less. All proceeds of insurance or compensation for expropriation or seizure, if any, shall be paid to the Purchaser on the Closing Date and all rights

and claims of the Seller to any such amounts not paid by the Closing Date shall be assigned on the Closing Date to the Purchaser.

Section 8.3 Single Transaction.

The Seller and Purchaser acknowledge and agree that, save as otherwise expressly provided in this Agreement, the transactions provided for in this Agreement require the Purchaser to purchase and the Seller to sell all of the Purchased Assets and, unless the Seller and Purchaser otherwise agree in writing, the Purchaser shall have no right and waives any right it may otherwise have to require the conveyance to it and the Seller shall have no right and waives any right it may otherwise have to require the purchase of less than all of the Purchased Assets *provided that* the Parties hereto agree not to hinder, delay or prevent Closing should assets immaterial in value or to the overall operation of the Operations be unable to be conveyed or purchased at Closing.

Article 9 TERMINATION

Section 9.1 Termination Rights.

- (1) This Agreement may be terminated by notice in writing:
 - (a) by mutual written consent of the Seller and the Purchaser;
 - (b) by either the Seller or the Purchaser if a Law shall have been enacted, entered or promulgated prohibiting the consummation of the transactions contemplated hereby substantially on the terms contemplated hereby; provided, however, that the right to terminate this Agreement pursuant to this Section 9.1(1)(b) shall not be available to any Party if such Law was primarily due to the failure of such Party to perform any of its obligations under this Agreement;
 - (c) by either the Seller or the Purchaser if the Closing has not occurred on or prior to the Outside Date, provided that the right to terminate this Agreement shall not be available to any Party whose failure to fulfill any of its obligations or breach of any of its representations, warranties or covenants under this Agreement has been the primary cause of, or resulted in, the failure of the Closing to occur by the Outside Date;
 - (d) by the Seller if a breach of any representation or warranty or failure to perform any covenant or agreement on the part of the Purchaser under this Agreement occurs that would cause any condition in Section 7.2 not to be satisfied, and such breach or failure is incapable of being cured by the Outside Date or is not cured by the Purchaser within thirty (30) days of being so notified by the Seller; provided that the Seller is not then in breach of this Agreement so as to cause any condition in Section 7.1 not to be satisfied; and
 - (e) by the Purchaser if a breach of any representation or warranty or failure to perform any covenant or agreement on the part of the Seller under this Agreement occurs that would cause any condition in Section 7.1 not to be satisfied, and such breach or failure is incapable of being cured by the Outside

Date or is not cured by the Seller within thirty (30) days of being so notified by the Purchaser; provided that the Purchaser is not then in breach of this Agreement so as to cause any condition in Section 7.2 not to be satisfied.

Section 9.2 Effect of Termination.

- (1) Each Party's right of termination under this Article 9 is in addition to any other rights each Party or a Party may have under this Agreement, and the exercise of a right of termination will not be an election of remedies. Nothing in this Article 9 limits or affects any other rights or causes of action any Party may have with respect to the representations, warranties, covenants and indemnities in its favour contained in this Agreement. If a Party waives compliance with any of the conditions, obligations or covenants contained in this Agreement, the waiver will be without prejudice to any of its rights of termination in the event of nonfulfilment, nonobservance or nonperformance of any other condition, obligation or covenant in whole or in part.
- (2) If this Agreement is terminated pursuant to Section 9.1, all obligations of the Parties under this Agreement will terminate, except that:
 - (a) the provisions of Article 1, Section 6.3 and Article 12 will survive; and
 - (b) if this Agreement is terminated by the Purchaser pursuant to Section 9.1(1)(e) or the Seller pursuant to Section 9.1(1)(d), the terminating Party's right to pursue all legal remedies will survive such termination unimpaired and the other Party shall remain fully liable for all Damages.
- (3) In the event this Agreement is terminated, the Seller accepts that it shall have no remedy against the Debt Financing Sources or their respective former, current or future general or limited partners, stockholders, managers, members, directors, officers, Affiliates, employees, representatives agents (for the purposes of this Section 9.2(3), the "Debt Financing Related Parties"), for any loss or damages suffered hereunder, and the Debt Financing Related Parties, shall not have any liability or obligation to the Seller relating to or arising out of, or in connection with, this Agreement, the Debt Commitment Letter or the transactions contemplated hereby or thereby.

Article 10 INDEMNIFICATION

Section 10.1 Survival.

(1) Subject to Section 10.1(2), the representations and warranties contained in this Agreement will survive the Closing and continue in full force and effect for a period of eighteen (18) months after the Closing Date, except that: (a) the Fundamental Representations and the representations and warranties set out in Section 4.1(1), Section 4.1(2), Section 4.1(2), Section 4.1(3)(a), Section 4.1(6), Section 5.1(1), Section 5.1(2), Section 5.1(3)(a) and Section 5.1(5)(a) will survive the Closing and continue in full force and effect indefinitely; (b) the representations and warranties set out in Section 3.1(19) will survive the Closing and continue in full force and effect for a period of three (3) years; and (c) the representations and warranties contained in Section 3.1(9) will survive the Closing and continue in full force and effect for a period of three (3) years; and effect for a period of 60 days after the

expiration of the last applicable limitation period under any Tax legislation subsequent to the expiration of which an assessment, reassessment or other form or recognized document assessing liability for Tax with respect to any taxation period to which these matters relate cannot be issued.

- (2) The time limitation in Section 10.1(1) will not apply in respect of any claims arising out of, resulting from, related to, or involving fraud, negligence, intentional misrepresentation or wilful misconduct by the Seller or the Purchaser.
- (3) All covenants contained in this Agreement which, by their express terms, are to be performed prior to the Closing shall terminate on (and shall not survive) the Closing Date. The covenants contained in this Agreement which are to be performed at or after the Closing shall terminate once such covenants are fully performed.
- (4) No Party has any obligation or liability with respect to any representation, warranty or covenant in this Agreement after the end of the applicable time period specified in Section 10.1(1) and Section 10.1(3), except for claims relating to the representations, warranties or covenants that the Party has been notified of prior to the end of the applicable time period.

Section 10.2 Indemnification in Favour of the Purchaser.

- (1) The Seller will indemnify, defend and save harmless the Purchaser and its directors, officers, employees, Affiliates, shareholders and representatives (collectively, the "**Purchaser Indemnitees**") from any Damages suffered by, imposed upon or asserted against any of the Purchaser Indemnitees as a result of, in respect of, connected with, or arising out of, under, or pursuant to:
 - (a) any breach, default or violation or any representation or warranty of the Seller contained in this Agreement;
 - (b) any failure of the Seller to perform or fulfill any covenant of the Seller contained in this Agreement;
 - (c) any Excluded Asset or Excluded Obligation; and/or the Seller's failure to discharge, perform, or fulfil any of the Excluded Obligations;
 - (d) any Taxes relating to the Operations or the Purchased Assets with respect to any taxable period (or portion thereof) ending on or before the Closing;
 - (e) all Damages related to the investigation, and the events, facts and circumstances giving rise to such investigation, by the Independent Electricity System Operator (Market Assessment and Compliance Division) ("MACD") [**] (collectively, the "Ongoing MACD Investigations") or in any similar investigations, and the events, facts and circumstances giving rise to such investigation, by MACD initiated prior to Closing regarding acts or omissions of Seller or its Affiliates with respect to the Operations that are of substantially the same nature as those alleged in the Ongoing MACD Investigations; and
 - (f) any Excluded Environmental Liabilities.

Section 10.3 Indemnification in Favour of the Seller.

The Purchaser will indemnify, defend and save harmless the Seller and its directors, officers, employees, Affiliates, shareholders and representatives (collectively, the "Seller Indemnitees") harmless from, any Damages suffered by, imposed upon or asserted against any of the Purchaser Indemnitees as a result of, in respect of, connected with, or arising out of, under, or pursuant to:

- (1) any breach, default or violation or any representation or warranty of the Purchaser contained in this Agreement;
- (2) any failure of the Purchaser to perform or fulfill any covenant of the Purchaser contained in this Agreement;
- (3) any Assumed Obligations or Purchased Assets and the Purchaser's failure to discharge, perform or fulfill any of the Assumed Obligations;
- (4) any Damages referred to in Section 2.16(3); and
- (5) any Taxes relating to the Operations or the Purchased Assets with respect to any taxable period (or portion thereof) after the Closing.

For greater certainty, should any Transfer Taxes be deemed included in any amount payable under this Section 10.3, such amount shall be grossed-up accordingly in order for the Purchaser to pay to the Seller (i) the full indemnity amount provided for under this Section 10.3 and (ii) an additional amount equivalent to any Transfer Taxes deemed included in such payment under applicable Law.

Section 10.4 Limitations.

- (1) The Purchaser Indemnitees will not be entitled to recover under this Article 10 until the total of all Damages with respect to all matters exceeds the Deductible, in which event the Indemnifying Party shall only be required to pay or be liable for Damages in excess of the Deductible.
- (2) The Purchaser Indemnitees shall not be entitled to recover any Damages for any Claim (or series of Claims arising from similar events or facts) where the Damages relating to each such Claim or series of Claims are less than the Per Claim Threshold, and such Damages shall not be counted toward the Deductible.
- (3) The Purchaser Indemnitees will not be entitled to recover under this Article 10 an aggregate amount in excess of the Cap other than in respect of a breach of the Fundamental Representations which shall be capped at the Purchase Price.
- (4) The Seller Indemnitees will not be entitled to recover under this Article 10 an aggregate amount in excess of the Purchase Price.
- (5) Notwithstanding any other provisions of this Agreement:

- (a) the limitations set forth in Section 10.4(1), Section 10.4(2) and Section 10.4(3) shall not apply to any claim involving fraud, negligence, intentional misrepresentation or wilful misconduct by the Seller or the indemnitees contemplated by Section 10.2(1)(c), Section 10.2(1)(d), Section 10.2(1)(e), or Section 10.2(1)(f); and
- (b) the limitations set forth in Section 10.4(4) shall not apply to any claim involving fraud, negligence, intentional misrepresentation or wilful misconduct by the Purchaser, the Parent Guarantor or the indemnitees contemplated by Section 10.3(2) to Section 10.3(5).

Section 10.5 Notice of Claim.

- (1) Any Indemnified Party may be entitled to make a claim for indemnification (a "Claim") under this Agreement, by written notification to the Indemnifying Party of such Claim (a "Notice of Claim") promptly upon becoming aware of the Claim, but in no event later than the relevant date, if any, as specified in Section 10.1. The Notice of Claim shall specify whether the Claim arises as a result of a Third Party Claim or a Direct Claim, and shall also specify with reasonable particularity (to the extent that the information is available), the factual basis for the Claim, the amount of the Claim and the identity of the Person making the Claim if it is a Third Party Claim.
- (2) If an Indemnified Party fails to provide the Indemnifying Party with a Notice of Claim promptly as required by Section 10.5(1), the Indemnifying Party shall be relieved of the obligation to pay damages to the extent it can show that it was prejudiced in its defence of the Claim or in proceeding against a third party who would have been liable to it by the fact of the delay, but the failure to provide such Notice of Claim promptly shall not otherwise release the Indemnifying Party from its obligations under this Section 10.5.
- (3) If the date by which a Notice of Claim must be given in respect of a breach of representation and warranty has passed without any Notice of Claim having been given to the Indemnifying Party, then the related Claim shall be forever extinguished, notwithstanding that by the date specified in Section 10.5(1) the Indemnified Party did not know, and in the exercise of reasonable care could not have known, of the existence of the Claim.

Section 10.6 Direct Claims.

- (1) With respect to any Direct Claim, following receipt of notice from the Indemnified Party of the Claim, the Indemnifying Party shall have forty-five (45) days to make such investigation of the Claim as is considered necessary or desirable. For the purpose of such investigation, the Indemnified Party shall make available to the Indemnifying Party the information relied upon by the Indemnified Party to substantiate the Claim, together with all such other information, books and records and access to employees as the Indemnifying Party may reasonably request in all cases except to the extent it is confidential or subject to privilege.
- (2) If the Indemnifying Party disputes the validity or amount of the Direct Claim, the Indemnifying Party shall provide written notice of the dispute to the Indemnified Party within the 45day period specified in Section 10.6(1). The dispute notice will describe in

reasonable detail the nature of the Indemnifying Party's dispute. During the 45day period immediately following receipt of a dispute notice by the Indemnified Party, the Indemnifying Party and the Indemnified Party shall attempt in good faith to resolve the dispute. If the Indemnifying Party and the Indemnified Party fail to resolve the dispute within that 45day time period, the Indemnifying Party's dispute shall be submitted to the Chief Executive Officer of each of the Indemnifying Party and the Indemnified Party fail to resolve the dispute. If the Chief Executive Officer of each of the Indemnifying Party and the Indemnified Party fail to resolve the dispute. If the Chief Executive Officer of each of the Indemnifying Party and the Indemnified Party fail to resolve the dispute within a 10day time period, the Indemnified Party is free to pursue all rights and remedies available to it, subject only to this Agreement. If the Indemnifying Party fails to respond in writing to the Direct Claim within the 45day period specified in Section 10.6(1), the Indemnifying Party is deemed to have rejected the Direct Claim in which event the Indemnifying Party is free

Section 10.7 Procedure for Third Party Claims.

to pursue all rights and remedies available to it, subject to this Agreement.

- (1) Subject to the terms of this Section 10.7, upon receiving notice of a Third Party Claim, the Indemnifying Party may, at its cost and expense, participate in the investigation and defence of the Third Party Claim or may also elect to assume the negotiation, investigation and defence of the Third Party Claim.
- (2) The Indemnifying Party may not, without the prior written consent of the Indemnified Party, assume the investigation and defence of a Third Party Claim if:
 - (a) the Indemnifying Party fails to acknowledge in writing its indemnity obligation in respect of such Third Party Claim, subject to the terms and conditions of this Agreement;
 - (b) the Third Party Claim relates to or arises in connection with any criminal proceeding, action, indictment, allegation or investigation;
 - (c) settlement of, or an adverse judgment with respect to, the Third Party Claim is, in the judgment of the Indemnified Party, likely to establish a precedent, custom or practice adverse to the continuing business interests of the Indemnified Party;
 - (d) the Third Party Claim is filed by a current material customer or supplier of the Operations;
 - (e) the Third Party Claim seeks any injunctive or other relief and not only monetary damages; and
 - (f) the named parties in any Third Party Claim include both the Indemnifying Party and the Indemnified Party and representation by the same counsel would, in the opinion of counsel to the Indemnified Party, be inappropriate (including as a result of the availability of different defences).
- (3) In order to assume the investigation and defence of a Third Party Claim, the Indemnifying Party must give the Indemnified Party written notice of its election within thirty (30) days of Indemnifying Party's receipt of notice of the Third Party Claim.

- (4) If the Indemnifying Party assumes the investigation and defence of a Third Party Claim, the Indemnifying Party will actively and diligently proceed with the defense and pay for all costs and expenses of the investigation and defence of the Third Party Claim except that the Indemnifying Party will not, so long as it diligently conducts such defence, be liable to the Indemnified Party for any fees of other counsel or any other expenses with respect to the defence of the Third Party Claim incurred by the Indemnified Party after the date the Indemnifying Party validly exercised its right to assume the investigation and defence of the Third Party Claim.
- (5) If the Indemnifying Party is not entitled to assume the investigation and defence of a Third Party Claim as aforesaid, does not elect to assume the investigation and defence of a Third Party Claim, or assumes the investigation and defence of a Third Party Claim but fails to diligently pursue such investigation and defence, the Indemnified Party has the right (but not the obligation) to undertake the investigation and defence of the Third Party Claim at the Indemnifying Party's cost and expense.
- (6) If the Indemnified Party undertakes the investigation and defence of a Third Party Claim, the Indemnifying Party will not be bound by any compromise or settlement of the Third Party Claim effected without the consent of the Indemnifying Party (which consent may not be unreasonably withheld, conditioned or delayed).
- (7) The Indemnifying Party will not be permitted to compromise and settle or remedy or to cause a compromise and settlement or remedy of a Third Party Claim without the prior written consent of the Indemnified Party, which consent may not be unreasonably withheld or delayed, unless:
 - the terms of the compromise and settlement or remedy require only the payment of money for which the Indemnified Party is entitled to full indemnification under this Agreement and not otherwise likely to prejudice or create precedent adverse to the Indemnified Party;
 - (b) the terms of the compromise and settlement do not require the Indemnified Party to admit any wrongdoing, take or refrain from taking any action, acknowledge any rights of the Person making the Third Party Claim or waive any rights that the Indemnified Party may have against the Person making the Third Party Claim; and
 - (c) the Indemnified Party receives, as part of the compromise and settlement or remedy, a legally binding and enforceable unconditional release, which is in form and substance satisfactory to the Indemnified Party, acting reasonably, from any and all obligations or liabilities it may have with respect to the Third Party Claim.
- (8) The Indemnified Party and the Indemnifying Party agree to keep each other fully informed of the status of any Third Party Claim and any related Proceedings. If the Indemnifying Party or Indemnified Party assumes the investigation and defence of a Third Party Claim, the other Party will act in such manner as not to interfere with the investigation and defence of the Third Party Claim. The Indemnified Party and the Indemnifying Party shall each make available, on a timely basis, those employees whose assistance, testimony or presence is reasonably necessary to assist in investigating and

- 80 -

defending the Third Party Claim. The Indemnified Party and the Indemnifying Party shall each, at the request and expense of the Indemnifying Party, make available, on a timely basis all documents, records and other materials in the possession, control or power, reasonably required solely for defending any Third Party Claim subject at all times to applicable confidentiality or privilege. The Indemnified Party shall cooperate on a timely basis with the Indemnifying Party in the defence of any Third Party Claim.

Section 10.8 Manner of Payment.

Subject to, and in accordance with, the provisions of this Article 10, any indemnification of the Indemnified Party pursuant to this Article 10 shall be delivered by the Indemnifying Party to the applicable Indemnified Party by wire transfer of immediately available funds to an account designated by the Indemnified Party within ten (10) days after the final, non-appealable determination thereof.

Section 10.9 Exclusion of Other Remedies.

Except as provided in this Section 10.9 and in Section 2.9 or in the case of gross negligence, intentional misrepresentation or wilful misconduct, the indemnities provided in Section 10.2 and Section 10.3 constitute the only remedy of the Purchaser, on the one hand, or the Seller, on the other hand, against a Party in the event of any breach of a representation or warranty of such Party contained in this Agreement. The Parties may exercise their right to payment of an adjustment in Section 2.9 and their rights of termination in Section 9.1. The Parties acknowledge that the failure to comply with a covenant or obligation contained in this Agreement may give rise to irreparable injury to a Party inadequately compensable in Damages. Accordingly, a Party may seek to enforce the performance of this Agreement by injunction or specific performance upon application to a court of competent jurisdiction without proof of actual damage (and without the requirement of posting a bond or other security), provided that in no event shall the Seller be entitled to seek the remedy of specific performance against any Debt Financing Sources (which shall not limit any remedies of the Purchaser pursuant to the Debt Commitment Letter). The Purchaser and the Seller expressly waive and renounce any other remedies whatsoever, whether at law or in equity, which it would otherwise be entitled to as against any other Party. The provisions in this Agreement relating to indemnification, and the limits imposed on the remedies with respect to this Agreement and the transactions contemplated hereby were specifically bargained for between sophisticated parties and were specifically taken into account in the determination of the amounts to be paid hereunder. No Party may avoid the limitations on liability set forth in this Article 10 by seeking damages for breach of contract, tort or pursuant to any other theory of liability.

Section 10.10 Duty to Mitigate.

Nothing in this Agreement in any way restricts or limits the general obligation pursuant to applicable Laws of an Indemnified Party to mitigate any loss which it may suffer or incur by reason of the breach by an Indemnifying Party of any representation, warranty, covenant or obligation of the Indemnifying Party under this Agreement. If any Claim can be reduced by any recovery any, settlement or otherwise under or pursuant to any insurance coverage, or pursuant to any Claim, recovery, settlement or payment by or against any other Person, the Indemnified Party shall, at the expense of the Indemnifying Party, take all appropriate and commercially reasonable steps to enforce such recovery, settlement or payment and the amount of any

Damages of the Indemnified Party shall be reduced by the amount of insurance proceeds actually received by the Indemnified Party (net of any deductible or other costs or expenses of recovery including Taxes and any actual increase in insurance premiums as a result of such Claim). Any Claim shall also be reduced by any Tax benefit actually realized by the Indemnified Party on account of such Damages in the taxable periods beginning on or prior to the first anniversary of the first day of the taxable period such Damages were incurred. If the Indemnified Party receives a Tax benefit after an indemnification payment is made to it, the Indemnified Party shall promptly pay to the Indemnifying Party, the amount of such Tax benefit at such time or times as and to the extent that such Tax benefit is realized by the Indemnified Party, less all costs, fees and expenses (including Taxes) incurred in seeking and collecting such recovery and realization.

Section 10.11 One Recovery.

- (1) Any Indemnified Party is not entitled to double recovery for any Claim even though they may have resulted from the breach of more than one of the representations, warranties, covenants and obligations of the Indemnifying Party in this Agreement.
- (2) Any amount required to be paid by an Indemnifying Party to an Indemnified Party under this Article 10 shall bear interest at an annual rate equal to the Bank of Canada's Target Overnight Rate plus 600 basis points accruing on a daily basis from the date on which a demand for payment is made until payment in full.

Section 10.12 Adjustments to Purchase Price.

Any payment made to the Purchaser Indemnitees under this Article 10 is a dollar-for-dollar decrease in the Purchase Price. Any payment made to the Seller Indemnitees under this Article 10 is a dollarfordollar increase in the Purchase Price.

Section 10.13 Guarantee by Parent Guarantor

- (1) The Parent Guarantor shall cause the Purchaser to timely perform and fulfill its obligations under this Agreement and hereby irrevocably guarantees the full, complete and timely performance by the Purchaser of each and every obligation of the Purchaser under this Agreement, including the payment of all present and future amounts. If any default shall be made by the Purchaser in the performance of any such obligations, then the Parent Guarantor shall perform or cause to be performed such obligation promptly following notice from the Seller specifying the default. The Seller may proceed to enforce its rights against the Parent Guarantor from time to time contemporaneously with or after any enforcement against the Purchaser. This guarantee is absolute and unconditional and the Parent Guarantor waives any and all defenses specifically available only to a guarantor other than performance in full by the Purchaser. The guarantee set forth in this Section 10.13 is a guarantee of payment and not merely of collection and shall be deemed a continuing guarantee and shall remain in full force and effect until the satisfaction in full of all obligations of the Purchaser under this Agreement.
- (2) The liabilities and obligations of the Parent Guarantor under this Section 10.13 are subject to the terms of this Agreement and shall not exceed any liability or obligation of the Purchaser to the Seller under this Agreement. The Parent Guarantor is entitled to all rights, privileges and defences available to the Purchaser with respect to any obligation

or liability, including all provisions of this Agreement relating to limitation of liability and the resolution of disputes.

Article 11 POSTCLOSING COVENANTS AND OTHER MATTERS

Section 11.1 Employees.

- (1) The Purchaser, subject to Closing:
 - (a) will, effective on and after the Closing Date, employ:
 - (i) all Non-Unionized Employees employed by the Seller on the close of business on the date immediately preceding the Closing Date, on substantially the same terms and conditions of employment in the aggregate as are in effect on the date hereof, including without limitation, recognizing length of service with the Seller and any predecessors of the Operations for all purposes, except for benefit accrual under the Purchaser's benefit plans; and
 - (ii) all Unionized Employees employed by the Seller on the close of business on the date immediately preceding the Closing Date
 - (collectively, the "Transferred Employees"); and
 - (b) will be the successor to the Seller under the Collective Bargaining Agreements pursuant to the provisions of applicable labour legislation and on and after the Closing Date will be bound by and observe all of the terms, conditions, rights and obligations of the Seller under the Collective Bargaining Agreements to the extent that they apply to the Unionized Employees.
- (2) For greater certainty, the Employees listed in Section 11.1 of the Disclosure Letter shall continue their employment with the Seller and shall not be transferred to the Purchaser under the terms herein (the "Excluded Employees"). The Seller will be responsible for and will discharge all obligations and liabilities regarding the conditions of employment of the Excluded Employees and will indemnify the Purchaser in case of administrative or legal proceedings, claim or complaint filed by an Excluded Employee involving the Purchaser.
- (3) The Seller will be responsible for and will discharge all obligations and liabilities for wages, severance pay, termination pay, notice of termination of employment or pay in lieu of such notice, damages for wrongful dismissal or other employee benefits or Claims, including vacation pay and short-term incentive plan payments, accrued up to the close of business on the date immediately preceding the Closing Date in respect of all Transferred Employees, provided that, for greater certainty:
 - (a) all obligations and liabilities of the Seller for severance pay, termination pay, notice of termination of employment or pay in lieu of such notice, damages for wrongful dismissal shall only relate to Employees whose employment was

terminated prior to the Closing Date or to Employees who file claims for alleged events which occurred prior to the Closing Date; and

- (b) the Purchaser assumes and will discharge all such obligations and liabilities accruing after the close of business on the day immediately preceding the Closing Date (including all such obligations and liabilities resulting from the transactions contemplated by this Agreement) in respect of all Transferred Employees.
- (4) All items in respect of Transferred Employees that require adjustment including premiums for unemployment insurance, Canada Pension Plan, Quebec Pension Plan, Quebec Parental Insurance Plan, employer health tax, applicable statutory hospitalization insurance, workers' compensation assessments, accrued wages, salaries and commissions and employee benefit plan payments or any statutory payments or taxes will be appropriately adjusted to the close of business on the day immediately preceding the Closing Date.

Section 11.2 Pension Plans and Other Benefit Plans.

- (1) Except as may be specifically provided herein, the Purchaser shall not assume any liability or obligation with respect to any Seller Benefit Plans that are not Assumed Benefit Plans (the "Excluded Benefit Plans").
- (2) Effective as of the Closing Date, and subject to any third party consents or waivers and other regulatory approvals necessary to permit the transfer or assignment, the Seller shall transfer or assign and the Purchaser shall assume all of the rights, responsibilities, duties, obligations and liabilities of the Seller with respect to (i) the Retirement Plan for Organized Employees of Tembec Enterprises Inc.- Kapuskasing Operations (Reg. No. 0342352), (ii) the Régime de rentes pour les employés de Tembec Industries Inc. (Taschereau et employés de bureau de La Sarre (Retraite Québec no. 30723 - CRA no. 969535) and (iii) Régime de retraite des employés syndigués de Tembec Industries Inc. - Division La Sarre N (operations forestières, garage, sciage et rabotage) (Retraite Québec no. 30729 - CRA no. 691865) (collectively, the "Assumed Benefit Plans"). Upon obtaining the required consent, waiver and approvals of such transfer or assignment and assumption, the Purchaser shall become the successor sponsor or policyholder, as applicable, of the Assumed Benefit Plans. In order to effect such assignment and assumption, the Seller and the Purchaser shall take such steps, prepare and execute such documents and seek such approvals of the appropriate regulatory authorities as may be necessary or desirable, including the execution of the Pension Plan Assignment and Assumption Agreement, attached hereto as Exhibit F. The Purchaser shall enter into new agreements as necessary to allow the efficient transfer or assignment and assumption of the Assumed Benefit Plans. Upon the completion of the transfer or assignment of the Assumed Benefit Plans to the Purchaser, and except as otherwise provided herein, the Seller shall no longer be responsible for the pension benefits accrued by the Employees and the former employees under the Assumed Benefit Plans in accordance with the terms of such Assumed Benefit Plans, the Collective Bargaining Agreements and applicable Laws. Each of the Seller and Purchaser shall be solely responsible for its respective expenses attributable to the assignment of the rights, duties, obligations and liabilities under the Assumed Benefit

Plans to the Purchaser. The Seller shall also collaborate with the Purchaser with regard to the administration of the Assumed Benefit Plans, pending their intended transfer, if it does not occur as of the Closing Date, including, among others, with respect to the replacement of any member of the pension committee for such plan. For the avoidance of doubt, and subject to any necessary pension plan committee consents, the Seller shall also collaborate with the Purchaser to allow the Purchaser to establish its own investment structure for the Assumed Benefit Plans.

- (3) The Seller shall provide, prior to Closing, all true, complete and correct data, records and correspondence in its possession that is necessary or useful to administer the Assumed Benefit Plans, in a form which is sufficient for the proper administration of any such Assumed Benefit Plan in accordance with their terms and all applicable Laws. Prior to Closing, the Purchaser shall have the right to reasonably request, and shall be provided with, all data, records or other information that is reasonably necessary or useful to allow the Purchaser to establish its own Benefit Plans.
- (4) For greater certainty, it is acknowledged that the Purchaser shall assume and by fully responsible for all liabilities and obligations with respect to the following: (i) the Ontario Bridge Benefit Plan, (ii) all post-retirement benefits and post-employment benefits of the Unionized Employees and retirees of the Operations as provided for in the relevant Collective Bargaining Agreements and the Non-Unionized Employees and retirees of the Operations, and (iii) the DC SERP Benefits.
- (5) For greater certainty, the Purchaser shall be responsible only for the following claims or benefit payments with respect to group insurance benefits of the Transferred Employees:
 - (a) with respect to death or dismemberment claims, those in respect of which the event occurred after the Closing Date;
 - (b) with respect to health claims, those in respect of which the services were provided or the supplies were purchased after the Closing Date; and
 - (c) with respect to short term and/or long term disability claims, for those claims resulting from events that occurred after the Closing Date.

Section 11.3 Pension Plan Arrangements.

(1) Subject to applicable Law and the requirements for any third party consents, for the purposes of facilitating the assignment and assumption of the Retirement Plan for Organized Employees of Tembec Enterprises Inc.- Kapuskasing Operations, the Seller shall cause a written notice to be issued to the Custodial Trustee prior to the Closing Date seeking the segregation and transfer of those underlying assets owned by, or attributed to, the Retirement Plan for Organized Employees of Tembec Enterprises Inc.- Kapuskasing Operations to the terms of the Master Trust Agreements. Such proportionate share of the underlying assets of the funds established pursuant to the Master Trust Agreements, shall, no earlier than the Closing Date, be segregated in favour of the Organized Employees of Tembec Enterprises Inc.-Kapuskasing Operations and shall be transferred to the pension fund of the Organized Employees of Tembec Enterprises Inc.-Kapuskasing Operations.

- (2) Any costs and expenses attributable to the segregation and transfer of the underlying assets of the funds established pursuant to the Master Trust Agreements shall be shared equally by the Seller and Purchaser. During the period from the execution of the Agreement up to and including the Closing Date, the Seller shall ensure that no notice of redemption is issued by any Person in respect of the Master Trust Agreements, other than consistent with past practice, and that the assets of the Master Trust Agreements are invested consistent with past practice.
- (3) The Parties agree that the value of the assets to be segregated and transferred from the funds established pursuant to the Master Trust Agreements to the pension fund of the Retirement Plan for Organized Employees of Tembec Enterprises Inc. – Kapuskasing Operations shall include the return obtained by the relevant investment or investment structure up to the date of the transfer of the relevant assets.

Section 11.4 Access to Purchaser Books and Records.

For a period of six (6) years from the Closing Date or for such longer period as may be required by applicable Laws, the Purchaser shall retain all original Books and Records relating to the Operations for the period prior to the Closing Date. So long as any such Books and Records are retained by the Purchaser pursuant to this Agreement, the Seller shall have the reasonable right to inspect and to make copies (at its own expense) of them at any time upon reasonable request during normal business hours and upon reasonable notice for any proper purpose and without undue interference to the business operations of the Purchaser and its Subsidiaries. The Purchaser shall have the right to have its representatives present during any such inspection. The Purchaser agrees to provide timely access to the Books and Records, financial statements, and any other resources and personnel of the Purchaser to the Seller upon the reasonable request of the Seller for purposes of complying with any applicable tax, financial reporting, regulatory requirements or any other proper purpose.

Section 11.5 Access to Seller Books and Records.

- (1) As regards written or electronic documents other than those transferred to the Purchaser as Books and Records, the Purchaser shall be entitled to request additional written or electronic records by specifying them in a written request made to the Seller within one (1) year from the Closing Date if and to the extent the request involves documents fulfilling the following criteria:
 - (a) the documents are identified by the Purchaser with such accuracy that the Seller can reasonably identify and search for such documents;
 - (b) the documents contain information which was used for the Operations before Closing and is necessary to conduct the Operations in the Ordinary Course after Closing;
 - (c) the documents do not contain information subject to confidentiality obligations owed to any third parties or, alternatively, such information can be reasonably redacted from the documents; and

(d) the documents do not contain sensitive information about any operations of the Seller other than the Operations or, alternatively, such information can be reasonably redacted from the documents,

Section 11.6 Duties.

- (1) The Purchaser shall cause any and all Duties refunds (including any credits for Duties received in lieu of a refund) received by or for the benefit of the Purchaser or any of its Subsidiaries, net of any reasonable expenses that the Purchaser or any of its Subsidiaries incur with respect to such refund and related interest, to be immediately paid to the Seller to the extent that such Duties refund relate to Duties that were paid by the Seller or for which the Seller was liable before the Closing. The Purchaser and any of its Subsidiaries shall, if the Seller so request, (i) reasonably cooperate with the Seller in filing any documentation or taking any steps, as may be necessary to obtain any Duties refunds to which the Seller is entitled or may be entitled under this Section 11.6(1), including but not limited to the filing of any customs adjustment form or request for refund as necessary, and (ii) provide to the Seller any information on or document relating to the progress made in respect of, or the status of, any Duties refunds. For greater certainty, the Purchaser and any of its Subsidiaries shall use all commercially reasonable efforts to obtain any Duties refund to which the Seller is entitled or may be entitled under this Section 11.6(1).
- (2) The Seller shall cause any Duties payable by the Purchaser and any of its Subsidiaries to the U.S. Customs and Border Protection, the Canada Border Services Agency or other foreign customs authority, in addition to any reasonable expenses that the Purchaser or any of its Subsidiaries incur with respect to such payment and related interest, to be immediately paid to the Purchaser to the extent that such Duties relate to Duties that should have been paid by the Seller before the Closing. The Seller shall not pay any amount in respect of such Duties as a result of an adjustment filed following the Closing unless the Seller, acting commercially reasonably, agrees that an error was made in the original declaration. The Purchaser and any of its Subsidiaries shall (i) provide the Seller with prior notice of its intent to make any adjustments to a customs declaration that may result in additional duties owing prior to making such an adjustment and allow the Seller an opportunity to comment or dispute the adjustment for any shipments subject to Section 11.6(2) and (ii) provide to the Seller any information on or document relating to the progress made in respect of, or the status of, any Duties payable.

Section 11.7 Insurance Policies.

From the Closing Date, the Purchaser shall maintain for itself and any of its Subsidiaries insurance policies in full force and effect and in good standing and in amounts and in respect of such risks as are normal and usual for companies engaged in similar businesses and subject to the same or similar perils or hazards.

Section 11.8 Further Assurances.

From time to time after the Closing Date, each Party shall, at the request of any other Party, execute and deliver such additional conveyances, transfers and other assurances as may be

reasonably required to effectively transfer the Purchased Assets to the Purchaser and carry out the intent of this Agreement.

Article 12 MISCELLANEOUS

Section 12.1 Notices.

Any notice, request for consent, direction or other communication (each a "**Notice**") given regarding the matters contemplated by this Agreement must be in writing (regardless of the fact that a specific provision of this Agreement specifies or fails to specify that a particular notice will be in writing), sent by personal delivery, courier, facsimile or electronic mail addressed:

(1) to the Seller at:

Rayonier Advanced Materials Inc. 1301 Riverplace Boulevard Suite 2300 Jacksonville, Florida 32207 United States of America

Attention: Chief Financial Officer with a copy to: General Counsel

with a copy to:

McCarthy Tétrault LLP 1000, De La Gauchetière Street West Suite 2500 Montréal, Québec H3B 0A2

Attention: Clemens Mayr and Chrystelle Chevalier-Gagnon Email: <u>cmayr@mccarthy.ca</u> and <u>cchevaliergagnon@mccarthy.ca</u>

(2) to the Purchaser or the Parent Guarantor at:

GreenFirst Forest Products Inc. 1800 - 510 West Georgia Street Vancouver, British Columbia V6B 0M3

Attention: Chief Financial Officer

with a copy to:

Norton Rose Fulbright Canada LLP Suite 3000 222 Bay St Suite 3000, Toronto, Ontario M5K 1E7

Attention: Walied Soliman and Trevor Zeyl Email: wailed.soliman@nortonrosefulbright.com and trevor.zeyl@nortonrosefulbright.com

A Notice is deemed to be delivered and received on the date of delivery if it is a Business Day and the delivery was made prior to 4:00 p.m. (local time in place of receipt) and otherwise on the next Business Day. A Party may change its address for service from time to time by providing a Notice in accordance with the foregoing. Any subsequent Notice must be sent to the Party at its changed address. Any element of a Party's address that is not specifically changed in a Notice shall be assumed not to be changed. Sending a copy of a Notice to a Party's legal counsel as contemplated above is for information purposes only and does not constitute delivery of the Notice to that Party. The failure to send a copy of a Notice to legal counsel does not invalidate delivery of that Notice to a Party.

Section 12.2 Time of the Essence.

Time is of the essence in this Agreement.

Section 12.3 Announcements and Public Communications.

- (1) Immediately after the execution of this Agreement, the Parent Guarantor shall issue a public announcement, announcing the entering into of this Agreement, which announcement shall address all matters required by TSXV policies and shall be in form and substance acceptable to each Party, each acting in a commercially reasonable manner.
- (2) A Party who wishes to make a release, public announcement or publicity with respect to the transactions contemplated in this Agreement shall obtain approval of the other Party to the form, nature and extent of such disclosure, which approval shall not be unreasonably withheld; provided, however, that to the extent the disclosure is required by Law or by a recognized securities exchange, the disclosing Party shall give prior notice to the other Party and a reasonable opportunity for the other Party to review or comment on the disclosure. Notwithstanding the foregoing, the disclosing party shall have the right to make the announcement or statement without the consent of the other Party if, after consultation, or attempted consultation, with the other Party, the disclosing party has been advised by its legal counsel that the information must be disclosed under applicable law or that changes to the proposed disclosure suggested by any other Party would make the disclosure misleading or otherwise not responsive to applicable law.
- (3) The Parties consent to, and nothing in this Section 12.3 or elsewhere in this Agreement shall prohibit, the use and disclosure by the Seller or its Affiliates of information with respect to the Operations, the Purchased Assets or the transactions contemplated in this Agreement necessary to comply with disclosure obligations under applicable Canadian and U.S. securities law and rules of the New York Stock Exchange, including the public filing of this Agreement on the Canadian Securities Administrators' System for Electronic Document Analysis and Retrieval (SEDAR) at www.sedar.com or on the United States Securities and Exchange Commission's Electronic Data Gathering, Analysis and Retrieval (EDGAR) system at www.sec.gov.

Section 12.4 Third Party Beneficiaries.

Except as otherwise provided in Section 10.2, Section 10.3 and Section 12.14, the Parties intend that this Agreement will not benefit or create any right or cause of action in favour of any Person, other than the Parties; <u>provided</u> that, notwithstanding the foregoing, each Debt Financing Sources is an express third party beneficiary of Section 6.6, Section 6.7, Section 10.9, Section 12.6, Section 12.7, Section 12.3(3) and this Section 12.4. Except for the Indemnified Party and (to the extent set forth in the immediately preceding sentence) the Debt Financing Sources, no Person, other than the Parties, shall be entitled to rely on the provisions of this Agreement in any action, suit, proceeding, hearing or other forum. The Parties reserve their right to vary or rescind the rights at any time and in any way whatsoever, if any, granted by or under this Agreement to any Person who is not a Party, without notice to or consent of that Person, including any Indemnified Party.

Section 12.5 Expenses.

- (1) Except as otherwise expressly provided herein or otherwise agreed by the Parties, the Purchaser will pay for its own costs and expenses and the Seller will pay for its own costs and expenses which are, in each case, incurred in connection with the negotiation, preparation, execution and performance of this Agreement and the transactions contemplated by them including the fees and expenses of legal counsel, investment advisors and accountants.
- (2) Notwithstanding Section 12.5(1), the Purchaser will, up to a maximum amount of US\$500,000, reimburse the Seller on the Closing Date for all reasonable and documented third party costs and expenses incurred by the Seller primarily in connection with (a) the preparation of the Carve-out Financial Statements, (b) any assistance of the Seller and its advisors pursuant to Section 6.6, and (c) any work or assistance of the Seller and its advisors to obtain the consents, approval and authorizations as are required to be obtained from the TSXV in connection with the transactions contemplated herein provided that, any expenses of the Seller's legal counsel shall be billed separately from any legal fees incurred in connection with the negotiation and preparation of this Agreement or other matters undertaken in furtherance of the transactions contemplated hereby (such amount to be reimbursed, the "Reimbursement Amount"). For greater certainty, (a) should any Transfer Taxes be applicable to the Reimbursement Amount, the Purchaser shall be liable for and shall pay any such Transfer Taxes and (b) should any Transfer Taxes be deemed included in the Reimbursement Amount under applicable Law, the Reimbursement Amount shall be grossed-up accordingly in order for the Purchaser to pay to the Seller (i) the net Reimbursement Amount and (ii) an additional amount equivalent to any Transfer Taxes deemed included in such payment under applicable Law.

Section 12.6 Amendments.

This Agreement may only be amended, supplemented or otherwise modified by written agreement signed by the Parties; provided that, Section 6.6, Section 6.7, Section 10.9, Section 12.4, Section 12.7, Section 12.3(3) and this Section 12.6 (and in each case any related definitions to the extent a modification of such definitions would modify the substance of any of the foregoing provisions) may not be amended, supplemented or otherwise modified in a

manner that is adverse in any material respect to the Debt Financing Source, without the prior written consent of the Debt Financing Source.

Section 12.7 Waiver.

No waiver of any of the provisions of this Agreement will constitute a waiver of any other provision (whether or not similar). No waiver will be binding unless executed in writing by the Party to be bound by the waiver; provided that, Section 6.6, Section 6.7, Section 10.9, Section 12.4, Section 12.6, Section 12.3(3) and this Section 12.7 (and in each case any related definitions to the extent a waiver or termination of such definitions would modify the substance of any of the foregoing provisions) may not be waived in a manner that is adverse in any material respect to the Debt Financing Sources, without the prior written consent of the Debt Financing Sources. A Party's failure or delay in exercising any right under this Agreement will not operate as a waiver of that right. A single or partial exercise of any right will not preclude a Party from any other or further exercise of that right or the exercise of any other right.

Section 12.8 NonMerger.

Except as otherwise expressly provided in this Agreement, the covenants, representations and warranties shall not merge on and shall survive the Closing as provided herein.

Section 12.9 Entire Agreement.

This Agreement constitutes the entire agreement between the Parties with respect to the transactions contemplated by this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the Parties. There are no representations, warranties, covenants, conditions or other agreements, express or implied, collateral, statutory or otherwise, between the Parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement. The Parties have not relied and are not relying on any other information, discussion or understanding in entering into and completing the transactions contemplated by this Agreement.

Section 12.10 Successors and Assigns.

- (1) This Agreement becomes effective only when executed by all of the Parties. After that time, it will be binding upon and enure to the benefit of each of the Parties and their respective successors and permitted assigns.
- (2) Neither this Agreement nor any of the rights or obligations under this Agreement are assignable or transferable by any Party without the prior written consent of the other Parties. Notwithstanding the foregoing, (x) the Seller shall not assign or transfer any of its rights under this Agreement, whether before or after Closing, to any Person that is a non-resident of Canada and (y) the Purchaser will have the right to assign all or any portion of its rights and obligations pursuant to this Agreement from and after the consummation of the Closing on the Closing Date to the Debt Financing Sources pursuant to the terms of the Debt Financing for purposes of creating a security interest herein or otherwise assigning as collateral in respect of the Debt Financing.

If any provision of this Agreement is determined to be illegal, invalid or unenforceable by an arbitrator or any court of competent jurisdiction, that provision will be severed from this Agreement and the remaining provisions shall remain in full force and effect.

Section 12.12 Governing Law; Waiver of Jury Trial.

- (1) This Agreement will be governed by and interpreted and enforced in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- (2) Each Party irrevocably attorns and submits to the exclusive jurisdiction of the Ontario courts situated in the City of Toronto and waives objection to the venue of any Proceeding in such court or that such court provides an inconvenient forum.
- (3) Notwithstanding anything in this Section 12.12 to the contrary, if, despite the waivers set forth in Section 6.6(2), any Proceeding is brought against the Debt Financing Sources, such Proceeding will be governed by and construed in accordance with, the laws of the State of New York, without regard to principles of conflicts of law to the extent that application of the law of another jurisdiction would be required thereby, and each of the Parties agrees that it will not bring or support any such Proceeding (whether at law, in equity, in contract, in tort or otherwise) against the Debt Financing Sources in any way relating to this Agreement or any of the transactions contemplated by this Agreement, the Debt Commitment Letter or the performance thereof, in any forum other than any New York State court or Federal court of the United States of America sitting in New York County, and any appellate court from any thereof. Each of the Parties hereby irrevocably waives any and all right to trial by jury in any Proceeding (whether based on contract, tort or otherwise) arising out of or relating to the Debt Commitment Letter, the Debt Financing or the actions of the Debt Financing Sources in the negotiation, performance or enforcement thereof.

Section 12.13 Counterparts.

This Agreement may be executed in any number of counterparts (including counterparts by facsimile or PDF) and all such counterparts taken together shall be deemed to constitute one and the same instrument.

Section 12.14 Legal Representation.

In any proceeding by or against the Purchaser wherein the Purchaser asserts or prosecutes any claim under, or otherwise seeks to enforce, this Agreement, the Purchaser agrees in connection with such proceeding (a) to waive any right the Purchaser may have to assert the attorney-client privilege against Seller Counsel or the Seller or any of their Affiliates with respect to any communication or information contained in Seller Counsel's possession or files without the prior written consent of the Seller and (b) to consent to the representation of the Seller and its Affiliates by Seller Counsel, notwithstanding that Seller Counsel has or may have represented the Seller or any of its Affiliates as counsel in connection with any matter, including any transaction (including the transactions contemplated by this Agreement), negotiation, investigation, proceeding or action, prior to the Closing. This consent and waiver extends to Seller Counsel representing the Seller against the Purchaser or its Subsidiaries in litigation,

arbitration or mediation in connection with this Agreement or the transactions contemplated by this Agreement. In addition, all communications exclusively between the Seller, on the one hand, and Seller Counsel, on the other hand, related to this sale of the Purchased Assets, this Agreement or the transactions contemplated hereby shall be deemed to be attorney-client confidences that belong solely to the Seller and its Affiliates (the "Seller Pre-Closing Communications"). Accordingly, the Purchaser shall not have access to any such Seller Pre-Closing Communications or to the files of Seller Counsel relating to such engagement from and after the Closing, and all books, records and other materials in any medium (including electronic copies) containing or reflecting any of the Seller Pre-Closing Communications or the work product of legal counsel with respect thereto, including any related summaries, drafts or analyses, and all rights with respect to any of the foregoing, are hereby retained by, assigned and transferred to the Seller effective as of the Closing. Such material and information shall be excluded from the transfer contemplated by this Agreement. From and after the Closing, the Purchaser and its Affiliates shall maintain the confidentiality of all such material and information. From and after the Closing, none of the Purchaser, its Affiliates and representatives shall access or in any way, directly or indirectly, use or rely upon any such materials or information. Without limiting the generality of the foregoing, from and after the Closing, the Seller and its Affiliates shall be the sole holders of the attorney-client privilege with respect to such engagement and the Seller Pre-Closing Communications, and none of the Purchaser and its Affiliates shall be a holder thereof. As to Seller Pre-Closing Communications, the Purchaser and the Seller, together with any of their respective Affiliates, successors or assigns, agree that no such party may use or rely on any of the Seller Pre-Closing Communications in any action or claim against or involving any of the Parties after the Closing. The Purchaser hereby acknowledges and confirms that it has had the opportunity to review and obtain adequate information regarding the significance and risks of the waivers and other terms and conditions of this Section 12.14, including the opportunity to discuss with counsel such matters and reasonable alternatives to such terms. This Section 12.14 is for the benefit of the Seller and Seller Counsel and Seller Counsel is the intended third party beneficiary of this Section 12.14. This Section 12.14 shall be irrevocable, and no term of this Section 12.14 may be amended, waived or modified, without the prior written consent of the Seller and Seller Counsel. The covenants and obligations set forth in this Section 12.14 shall survive for ten (10) years following the Closing Date.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF the Parties have executed this Asset Purchase Agreement.

PURCHASER:

9437-6001 QUÉBEC INC.

By:	(signed) "Larry Swets"
	Authorized Signatory
By:	(signed) "Michael Liggett"
	Authorized Signatory

PARENT GUARANTOR:

GREENFIRST FOREST PRODUCTS INC.

By:	(signed) "Paul Rivett"
	Authorized Signatory
By:	(signed) "Rick Doman"
	Authorized Signatory

RAYONIER A.M. CANADA G.P., herein represented by its managing partner, RAYONIER A.M. CANADA INDUSTRIES INC.

By: (signed) "Patrick Lebel" Vice President and Secretary

RAYONIER A.M. CANADA INDUSTRIES INC.

By: (signed) "Patrick Lebel" Vice President and Secretary

EXHIBIT A

CHIPS AND RESIDUAL FIBRE PURCHASE AGREEMENT

THIS AGREEMENT is dated the • day of April, 2021 (the "Agreement").

BETWEEN: RAYONIER A.M. CANADA G.P., a general partnership formed under the laws of the Province of Québec, having a place of business at 100-4 Place Ville-Marie, Montréal (Québec) H3B 2E7;

(hereinafter referred to as the "Purchaser")

SELLER:

AND: 9437-6001 QUEBEC INC., a corporation incorporated pursuant to the laws of Québec, having a place of business at 1 Place Ville-Marie, Suite 2500, Montreal (Quebec) H3B 1R1;

(hereinafter referred to as the "Seller")

(hereinafter referred to individually as a "party" and collectively as the "parties")

WHEREAS the Seller, as purchaser, GreenFirst Forest Products Inc. ("**GreenFirst**"), as parent guarantor, and Rayonier A.M. Canada Industries Inc. and the Purchaser, as sellers, entered into an asset purchase agreement dated as of April •, 2021 (the "**Purchase Agreement**");

WHEREAS pursuant to the Purchase Agreement, the Seller has agreed to sell and the Purchaser has agreed to purchase Chips (as defined below) and Residual Fibre (as defined below), as more fully set out and subject to the terms and conditions of this Agreement;

NOW THEREFORE, in consideration of the mutual covenants herein contained, the parties hereto covenant and agree as follows:

1. DEFINITIONS

1.1 As used in this Agreement, the following terms have the following meaning:

"Additional Samples" has the meaning specified in Section 7.3.1.

"Affiliates" means, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. As used in this definition, the term "control" (including, with its correlative meanings, "controlled by" and "under common control with") means the possession, directly or indirectly, of the powers to direct or cause the direction of management or policies of a Person, through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

"Agreement" has the meaning specified in the recitals.

"Applicable Region" has the meaning specified in Section 9.2.2.

"**BDMT**" means the green weight, in tonnes, of Chips or Residual Fibre less the weight of any water in such tonnage, all as determined by taking a representative sample of the wood fibre in accordance with Section 7, weighing it green, drying it in an oven at approximately 100C until the moisture is removed and then dividing the bone-dry weight of the sample by the green weight and multiplying the resulting ratio by the total green weight of the tonnage.

"Béarn Mill" means Seller's mill located at 67 rue Principale Sud or Route 391, Béarn, Québec, JOZ 1GO.

"**Business Day**" means any day of the year, other than a Saturday, Sunday or any day on which major banks are closed for business in Montreal, Québec or Toronto, Ontario (Canada).

"Business Manager" has the meaning specified in Section 10.2.

"Chapleau Mill" means Seller's mill located at 175 Planer Road, Chapleau, Ontario, POM 1K0.

"Chips" means the wood chips manufactured by the Seller at the Seller's Mills other than whole log chips.

"Cochrane Mill" means Seller's mill located at 70-17th Avenue P.O. Box 1059 Cochrane, Ontario, POL 1CO.

"Committed Chips Volume" means the aggregate of: [**].

"Committed Residual Fibre Volume" means [**].

"**Committed Volumes**" means collectively, each of (a) the Committed Chips Volume for a Contract Year and (b) the Committed Residual Fibre Volume for a Contract Year.

"**Confidential Information**" means all information disclosed by or on behalf of Disclosing Party to the Recipient in connection with or relating to this Agreement, whether tangible or intangible, and in whatever form or medium, unless such information (a) was in the possession of the Recipient at the time of such disclosure and was not subject to any other confidentiality obligations owing by the Recipient to the Disclosing Party, (b) was already known to the general public, (c) subsequently became known to the general public through no fault or omission on the part of the receiving party, or (d) was obtained from a *bona fide* third party source having the right to disclose such information to the Recipient. The specific terms and conditions of this Agreement shall be deemed to be Confidential Information of both parties hereto.

"**Contamination**" means any material that may adversely affect the quality, process or equipment or any other processing or manufacturing equipment used by the Purchaser in the manufacture of pulp or its customers in the manufacture of their end products, and includes loose sawdust, dirt, grit, soot, cinders, fly ash, burned wood, rocks, treated wood, plastic, rubber, paint, metal, nylon polypropylene, glue, petroleum products, chlorophenols, dioxins, furans and precursors of dioxins and furans and any other foreign materials.

"Contract Year" means a full year of the Term beginning on every January 1 and ending on every December 31.

"Delivery" has the meaning specified in Section 6.3.

"Delivery Schedule" has the meaning specified in Section 6.2.

"Destination" has the meaning specified in <u>Schedule A</u>.

"Disclosing Party" means any party to this Agreement that discloses Confidential Information.

"**Dispute**" has the meaning specified in Section 10.1.

"Divested Mill" has the meaning specified in Section 15.3.

"Event of Default" has the meaning specified in Section 13.1.

"Force Majeure" has the meaning specified in Section 11.1.

"FSC" has the meaning specified in Section 5.

"GMT" means the weight in green metric tonnes of the bark forming part of the Residual Fibre calculated in accordance with Section 8.

"GreenFirst" has the meaning specified in the recitals.

"Group Vice-President" has the meaning specified in Section 10.2.

"HST" means Harmonized Sales Tax.

"**Initial Contract Year**" means the period starting on the date this Agreement was executed and ending on December 31 of the year this Agreement was executed.

"Initial Term" has the meaning specified in Section 2.

"Initiator" has the meaning specified in Section 10.3.

"**Kapuskasing Chip Volume**" means the volume of Chips, measured in BDMT, produced by the Cochrane Mill during a Contract Year that the Seller designates, acting reasonably and only for use at the Kapuskasing newsprint mill, on or before November 1 of the previous Contract Year as being destined for the Kapuskasing newsprint mill.

"La Sarre Mill" means Seller's mill located at 495 Highway 111 West, La Sarre, Québec, J07 2X6.

"Losses" means all claims, liabilities, losses, damages, costs and expenses (including all costs of investigation, reasonable legal and other professional fees and disbursements, interest, penalties and amounts paid in settlement but excluding all consequential, special and indirect losses, loss of sales or profit and loss of opportunity).

"Origin" has the meaning specified in <u>Schedule A</u>.

"**PEFC**" has the meaning specified in Section 5.

"**Person**" includes an individual, a corporation, a partnership, a limited liability company, an unlimited liability company, an association, a trust or any other entity or organization.

"Price" has the meaning specified in Section 9.1.

"Price Arbitrator" has the meaning specified in Section 10.8.

"Purchase Agreement" has the meaning specified in the recitals.

"Purchaser" has the meaning specified in the recitals.

"Purchaser's Mill" means the pulp and paper mill in or about Témiscaming, Québec.

"Recipient" means any party to this Agreement that receives Confidential Information.

"Renewed Term" has the meaning specified in Section 2.

"Representative" will include any employee, officer, director, shareholder or other agent or representative of the party.

"Residual Fibre" means Wood Bark and shavings.

"Second Level Dispute Resolution" has the meaning specified in Section 10.2.

"Second Level Recipients" has the meaning specified in Section 10.2.

"Seller" has the meaning specified in the recitals.

"Seller's Mills" means collectively, the Béarn Mill, the Chapleau Mill, the Cochrane Mill and the La Sarre Mill, and any capital improvements, expansions or additions to such facilities, so long as such lumber mills and facilities are owned and operated by the Seller or its Affiliates.

"Specifications" means the Chip specifications set out in <u>Schedule B</u>, as may be amended from time to time in accordance with Section 3.

"Supply Notice" has the meaning specified in Section 4.5.

"Suspended Period" has the meaning specified in Section 11.7.

"**Term**" means collectively the Initial Term and any Renewed Term.

"Termination Notice" has the meaning specified in Section 12.1.4.

"**Wood Bark**" means wood bark and other fibre that meets the Bark Specifications set out in <u>Schedule B</u>, as may be amended from time to time in accordance with Section 3.

2. TERM

This Agreement shall have an initial term from the date of this Agreement and ending on December 31, 2040 ("**Initial Term**"), except if terminated earlier in accordance with the provisions of this Agreement. Unless one party sends the other party a notice stating its intention not to renew this Agreement at least two (2) years before the end of the Initial Term, this Agreement shall continue until such time as it is terminated pursuant to Section 12 hereof (the "**Renewed Term**").

3. SPECIFICATIONS OF THE CHIPS and residual fibre

The Chips delivered to the Purchaser during the Term must comply with the Specifications and shall be free of any Contamination. The Residual Fibre to be delivered to the Purchaser during the Term shall be free of any Contamination. The Purchaser represents to the Seller that the Chips produced by the Seller's Mills in the two years immediately preceding the date of this Agreement were in material compliance with the Specifications. If the Purchaser wishes to vary any or all of the Specifications from time to time after the date of this Agreement as it determines advisable or necessary to meet its production requirements, the Purchaser will deliver to the Seller notice of such required variation whereupon the Seller will use commercially reasonable efforts, to make such changes to ensure that it produces Chips or Residual Fibre meet the revised Specifications within a reasonable time so long as all such additional operating and capital expenditures incurred to meet such revised Specifications are the sole responsibility and cost of the Purchaser.

4. VOLUME OF CHIPS and Residual Fibre

- 4.1 <u>Obligation to Purchase and to Sell</u>. Except in case of a Force Majeure event, each Contract Year, the Purchaser shall purchase from the Seller, and the Seller shall sell to the Purchaser, the Committed Volumes.
- 4.2 <u>Commitment to Use Reasonable Efforts</u>. The Purchaser represents to the Seller that the following reflects in all material respects the historical productions at the Seller's Mills:
 - 4.2.1 [**] of Chips at the Béarn Mill;
 - 4.2.2 [**] of Chips at the Chapleau Mill;
 - 4.2.3 [**] of Chips (less the Kapuskasing Chip Volume) at the Cochrane Mill; and

4.2.4 [**] of Chips at the La Sarre Mill.

- 4.3 If, in any Contract Year and for reasons other than a Force Majeure event, the Seller anticipates that it will not produce Chips in the aggregate quantity and from the Seller's Mills set out above in Section 4.2, the Seller shall promptly inform the Purchaser and shall use all commercially reasonable efforts to find alternate sources of wood chips in order to supply to the Purchaser a total of at least [**] of wood chips (less the aggregate of (a) the Kapuskasing Chip Volume and (b) the above noted volumes for those Seller's Mills that have been permanently closed or sold) per Contract Year. The Seller shall submit a written notice setting out the proposed alternate plan for the supply of wood chips (from the Seller's Mills, its other mills or from third parties) and the costs of the wood chips. The Purchaser acknowledges that any wood chips set out in the alternate plan will not be considered substituted wood chips under Section 4.10. The Purchaser will decide, in its absolute discretion, whether it will agree in whole or in part to the proposed alternate plan for the supply of wood chips. If the Purchaser rejects the proposed alternative plan in whole or in part, the Seller has no further obligation to find alternative sources to supply wood chips to replace the shortfall addressed in the alternate plan for this occurrence or during such Contract Year, as applicable. The Purchaser shall also use all commercially reasonable efforts to find alternate sources of wood chips to replace the shortfall addressed in the alternate plan for this occurrence or during such Contract Year, as applicable.
- 4.4 Notwithstanding Section 4.3, if the Seller is not able to supply a total of at least **[**]** of wood chips (less the aggregate of (i) the Kapuskasing Chip Volume and (ii) the above noted volumes for those Seller's Mills that have been permanently closed or sold) to the Purchaser during a Contract Year, the Purchaser can request that the Seller produce whole log chips at the Seller's Mills, where permitted, to account for the shortfall of Chips. The price for the whole log chips will be based on a full cost recovery to the Seller and will take into account any capital expenditures (but only if such capital expenditures is required to produce the whole log chips and subject to the prior written approval of the Purchaser) and an appropriate overhead allocation.
- 4.5 <u>Anticipated Annual Volume</u>. Not later than 30 days before the beginning of each Contract Year, the Seller will provide to the Purchaser a schedule (a "**Supply Notice**") of the anticipated volume or weight, as the case may be, including an estimate of species, by location for Chips and Residual Fibre to be produced in each month of the Contract Year.

- 4.6 <u>Notice of Supply Change.</u> Not later than 10 Business Days prior to the beginning of a calendar quarter within any particular Contract Year, the Seller will deliver a notice to the Purchaser in the form of the Supply Notice setting out anticipated amounts of the Chips and Residual Fibre to be produced and sold for each of the remaining quarters in that Contract Year and the Supply Notice is thereby amended for the balance of the Contract Year.
- 4.7 <u>Regular Meetings.</u> Each party hereto will designate a representative for the purposes of administering this Agreement and such representatives will meet at least monthly (or less frequently if the parties agree) to review any and all variations and deviations from the Supply Notice, as amended from time to time, and to provide on a timely basis amount, pricing and other information to assist the other in administering this Agreement.
- 4.8 <u>Location of Meetings and Costs.</u> All meetings with respect to this Section 4 shall take place, unless the parties hereto otherwise agree, either in Montreal or Toronto, on an alternating basis. Each party shall bear their own costs and expenses incurred in connection with their respective activities under this Section 4.
- 4.9 <u>Right to Direct Chips.</u> The Seller acknowledges that the Purchaser may at any time and from time to time direct all or a portion of the Chips purchased by the Purchaser under this Agreement to a third party; provided that any additional costs or expenses of the Seller arising as a consequence of such direction will be assumed and paid for by the Purchaser.
- 4.10 <u>Right to Substitute Chips.</u> At any time and from time to time, despite the provisions of Sections 4.1 and 4.9, the Seller will have the right to substitute for Chips sold hereunder an equivalent volume of wood chips produced by other facilities or other Persons and the terms and conditions of this Agreement including price, Specifications and delivery terms will apply as if the substituted wood chips were Chips, except that if the freight and shipping costs of delivering such substituted chips to the Purchaser's Mill are more than the freight and shipping costs of delivering the Chips from the Seller's Mills to the Purchaser's Mill, then the Seller will pay the Purchaser the amount of the difference in such costs (in cash or by set off, as agreed with the Purchaser).

5. FOREST STEWARDSHIP COUNCIL DESIGNATION

The Seller acknowledges that, as of the date of this Agreement, its operations have been certified for Forest Stewardship Council ("**FSC**") and/or Program for the Endorsement of Forest Certification ("**PEFC**") chain-of-custody and controlled wood by a certifying body. The Seller shall maintain the certifications throughout the Term, unless the Purchaser informs the Seller in writing that it no longer requires the Seller to maintain the certifications. As long as the certifications are maintained, the Seller will provide to the Purchaser a proportional volume of FSC and PEFC credits monthly for Chips delivered to the Purchaser's mill based on the inbound volume of FSC and PEFC (coming from Sustainable Forestry Initiative certified wood) deliveries of all saw log fibre received at the Seller's Mills.

6. DELIVERY SCHEDULE, DELIVERY, TRANSPORTATION, AND TRANSFER OF TITLE AND RISKS

6.1 <u>Availability and Purchase</u>. The Seller shall make available for pickup to the Purchaser the Committed Volumes at the Seller's Mills as herein provided.

- 6.2 <u>Delivery Schedule</u>. During the Term, a delivery schedule for the Committed Chips Volume and the Committed Residual Fibre Volume will be established by the parties for the next Contract Year during the last month of the previous Contract Year, and will, insofar as possible, be distributed evenly over twelve (12) months (the "**Delivery Schedule**"). The Delivery Schedule may be modified after fourteen (14) Business Days' written notice is given and with the prior consent of the other party, which consent may not be denied without reasonable grounds. If the Seller is required to re-handle the Chips as a consequence of the failure of the Purchaser to arrange for transportation or take delivery of such Chips as per the Delivery Schedule, the Purchaser will pay the Seller all expenses and charges reasonably incurred to re-handle such Chips.
- 6.3 <u>Delivery from the Seller's Mills</u>. The Seller will load the tractor trailers provided by the Purchaser for the purposes of receiving the Chips and/or Residual Fibre, in a timely manner upon the arrival of such tractor trailers, and in accordance with standard industry practices. The Seller shall be deemed to have delivered Chips and/or Residual Fibre to Purchaser once the loading of such Chips and Residual Fibre into the Purchaser's tractor trailers at the Seller's Mills has been completed (the "**Delivery**"). The Delivery will be made in accordance with the Delivery Schedule. The Purchaser, at its costs, shall be responsible for arranging for taking Delivery and transportation of the Chips and/or Residual Fibre from the Seller's Mills.
- 6.4 <u>Transfer</u>. Transfer of title and all risks relating to the Chips and Residual Fibre from the Seller to the Purchaser will occur when the Purchaser has taken Delivery at Seller's Mill. The risk of any Losses to the Chips and Residual Fibre will shall follow the transfer of title.
- 6.5 <u>Covenants</u>. The Purchaser will:
 - 6.5.1 cause the directors, officers, employees, agents and contractors of the Purchaser and of its partners, while on the premises of the Seller's Mills, to comply with all rules and regulations of the Seller of general application and indemnify and hold harmless the Seller and its directors, officers, employees, agents and contractors from and against any Loss, to the extent attributable to a breach of such rules and regulations or to any negligence or misconduct of the Purchaser or any of such directors, officers, employees, agents or contractors;
 - 6.5.2 comply with all applicable laws, regulations, rules and governmental restrictions and maintain in good standing all licenses, permits and approvals from any and all governments, governmental commissions, boards or agencies applicable to the performance of its obligations hereunder; and
 - 6.5.3 ensure that all tractor trailers that are used to deliver the Chips or Residual Fibre purchased by the Purchaser under this Agreement are covered by comprehensive general liability insurance to cover loss or damage to Persons, property, Chips and Residual Fibre arising from actions related to the Purchaser's duties to arrange for the delivery of Chips and Residual Fibre under this Agreement with limits of coverage to be not less than \$3,000,000 (reviewed every two years and adjusted for inflation and other reasonable factors) inclusive of bodily injuries or property damage for each occurrence.
7. CHIPS and Residual Fibre QUALITY CONTROL

- 7.1 <u>General</u>. Seller warrants that the Chips and Residual Fibre sold to the Purchaser hereunder will meet the Specifications. The foregoing warranty is exclusive and is in lieu of all other warranties (whether written, oral or implied), which the Purchaser specifically disclaims.
- 7.2 <u>Analyses</u>. The Purchaser may, at its expense, perform appropriate analyses and sampling of each truckload of the Chips and Residual Fibre delivered by the Seller to determine the volume of Chips and Residual Fibre and ensure that the truckload of Chips meet the Specifications and that the truckload of Chips and Residual Fibre are free of Contamination. The Purchaser will perform and provide to the Seller such analyses and sampling within five (5) days after Delivery of the Chips and/or Residual Fibre, failing which the Purchaser will be deemed to have accepted the Delivery. Any data established by the Purchaser's analyses and sampling will be recorded in writing and will indicate (a) the volume of Chips and/or Residual Fibre delivered, (b) whether the Specifications, if applicable, have been met, and (c) whether the Chips and/or Residual Fibre delivered are free of Contamination. The Seller may at any time appoint representatives who may be present during the analyses and sampling of the Chips and/or Residual Fibre by the Purchaser, which analyses and sampling shall be conducted during normal business hours.
- 7.3 <u>Verification by the Seller</u>.
 - 7.3.1 At the reasonable request of the Seller, additional analyses will be conducted and additional samples will be taken (the "Additional Samples") in order to verify the results obtained by the Purchaser pursuant to Section 7.2. The Seller shall determine, in its sole discretion, whether the Additional Samples shall be taken and analyzed jointly by the Seller's representatives and the Purchaser's representatives, or separately by each of them. The samples used for this verification will be kept for a second verification, if necessary, after which they may be disposed of.
 - 7.3.2 When the Additional Samples are taken and analyzed separately by the parties, each of the Additional Samples must be divided into three approximately equal parts, of which one, the "control sample", will be preserved in accordance with best industry practices to ensure that the characteristics of this sample are not altered.
 - 7.3.3 If the difference between the results of the analyses performed on the Additional Samples taken by the parties, where applicable, is:
 - (i) less than 1% for each of the Specifications in respect of the Chips; or
 - (ii) equal to 0% in respect of the existence of Contamination,

the results obtained by each party will be adopted, in alternating sequence, and the control sample will be destroyed. If the difference is greater than 1% in respect of the Specifications or 0% in respect of the existence of Contamination, as the case may be, each party may revise its applicable results or request an expert examination under Section 7.4.

7.4 <u>Expert examination</u>. If a party requests an expert examination, then the Seller and the Purchaser shall jointly appoint a mutually agreeable forestry product expert who shall determine (as an expert and not an arbitrator) the characteristics in dispute using the control sample and provide a

report within five (5) Business Days. The report must specify the difference, if any, between the characteristics of the Chips in the control sample and the Specifications and/or whether the Chips and/or the Residual Fibre are free of Contamination. The expert's report will be binding on the parties and will be final and not subject to appeal. Subject to the provisions of Section 7.5, the expert's fees will be paid by the Purchaser and the Seller in equal shares.

- 7.5 <u>Adjustment</u>. In the event that an adjustment to the volume of Chips and/or Residual Fibre purchased by the Purchaser is necessary as a result of the expert's report, the adjustment must be applied to the Deliveries from which the control sample was taken. The party ultimately paying for the volume adjustment must also pay for all of the expert's fees.
- 7.6 <u>Defects</u>. In the event that the analyses or sampling performed under this Section 7 indicate that the quality of the Chips delivered does not meet the Specifications or that the Chips and/or the Residual Fibre are not free of Contamination and if, in the case of a load of Chips, use of such load would likely cause handling, production or quality problems in the manufacture of pulp at the Purchaser's Mill, the Purchaser will notify the Seller in writing, stating the alleged defect and including the statements of the analyses. After the Seller has received the notice, the Purchaser will be entitled to immediately reject all or part of any Delivery of Chips and/or Residual Fibre that the Purchaser believes, acting reasonably and following a visual inspection, to be affected by the defects described in the notice. The Purchaser reserves the right to refuse the entirety of any future Delivery that the Purchaser believes, acting reasonably and following a visual inspection 7.6.
- 7.7 <u>Variations from Specifications.</u> If the Purchaser refuses to accept a truckload of Chips or Residual Fibre for the reasons set out in Section 7.6, then Seller shall (a) take back the truckload, in which case all transportation costs related to the truckload not accepted will be for the account of the Seller, and Seller will be free to sell such truckload of Chips or Residual Fibre to any Person, and (b) make available and deliver to the Purchaser Chips or Residual Fibre, as applicable, that are free of Contamination without delay.
- 7.8 <u>Limitation</u>. Despite any other provision in this Agreement but subject to Seller's obligations set out in Section 7.7, the parties acknowledge that the Purchaser's right to reject and refuse any Delivery constitutes the sole and exclusive remedy with respect to Chips that do not meet the Specifications or Chips and/or Residual Fibre that are not free of Contamination.

8. CONTROL OF CHIPS AND RESIDUAL FIBRE QUANTITIES

- 8.1 <u>Definitions</u>. The following definitions apply to this Section 8:
 - 8.1.1 *Total Weight* The weight of the vehicle and its load of Chips or Residual Fibre, as they may be at the time of weighing;
 - 8.1.2 Vehicle Weight The weight of the empty vehicle weighed immediately after unloading; and
 - 8.1.3 Net Weight Total Weight minus Vehicle Weight.
- 8.2 <u>Net Weight</u>. Each vehicle must be weighed on a scale designated by the Purchaser before and after the Chips and the Residual Fibre are unloaded, in order to determine the Net Weight of the delivered Chips and/or Residual Fibre.

- 8.3 <u>Unit of measurement</u>. The unit of measurement for determining the quantities of Chips and Residual Fibre sold and delivered under this Agreement is one BDMT, except for Wood Bark which the unit of measurement shall be one GMT.
- 8.4 <u>Scales</u>. The Purchaser will ensure that the scales used will be subject to the standards and rules relating to weigh scales, as published and brought into force by Measurement Canada, an agency of Innovation, Science and Economic Development Canada. The Seller will be entitled, at any time, to request, by written notice to the Purchaser, that the scales be inspected by the inspectors with Measurement Canada. If the inspection establishes that the scales are in compliance with the regulations and standards, the Seller will bear the costs of the inspection. However, if the inspection establishes that the scales are not in compliance with the regulations and standards, the Purchaser must bear the costs of the inspection. Notwithstanding anything to the contrary contained herein, the Purchaser shall, at its own expense, have its scales periodically inspected as required by applicable laws and regulations and shall provide evidence thereof to the Seller upon request. In addition to the foregoing, the Seller may, upon reasonable written notice and during normal business hours, inspect and test the aforementioned measurement procedures, methods and practices and, if the Seller discovers any errors in these procedures, methods and practices, the Purchaser shall make such required modifications as may be mutually agreed upon between the parties.
- 8.5 <u>Statements</u>. Written statements stating the weight of Chips and Residual Fibre must contain the following information: the Total Weight and the Vehicle Weight of the vehicle used to carry the Chips and the Residual Fibre. Statement forms will be provided by the Purchaser and must be numbered and printed in duplicate in order for a copy to be provided to the Seller (the original will remain in the Purchaser's possession).

9. PRICE AND PAYMENT

- 9.1 <u>Price</u>. The price payable by the Purchaser to the Seller (the "**Price**") for the Chips and the Residual Fibre delivered and accepted by the Purchaser will be the prices established pursuant to Section 9.2 subject to the deductions in volume, if any, calculated in accordance with <u>Schedule B</u>. The Purchaser must also pay the Seller the applicable QST/GST or HST, as applicable, on the Price as well as any other sales taxes levied by any governmental authority on the purchase and sale transactions contemplated hereunder.
- 9.2 <u>Negotiation of the Price of Chips and the Residual Fibre</u>.
 - 9.2.1 The Price of the Chips and Residual Fibre delivered to the Purchaser during the Initial Contract Year are set forth in <u>Schedule A</u>.
 - 9.2.2 Thereafter, the Price of the Chips and Residual Fibre will be determined by negotiation in good faith by the two parties for each Contract Year during the last sixty (60) days of the Contract Year preceding the Contract Year for which the Price must be negotiated. [**]
 - 9.2.3 In the event that negotiations are delayed for any reason, any Delivery made after the first day of each Contract Year of the Agreement will be adjusted afterward to the new Price for the Chips and the Residual Fibre as the case may be, unless the parties otherwise agree.

- 9.2.4 Failure to agree on the Price will not have the effect of interrupting Deliveries and/or receipts and will rather be an incentive to continue negotiations, and any dispute in this regard must be settled pursuant to Section 10 of this Agreement.
- 9.3 <u>Payment.</u> Payment will be made as follows:
 - 9.3.1 The Seller will not be required to submit invoices. Payments will be based on the Purchaser's "Avis de Remise" computer-based reporting scale software program which provides a list of all transactions recorded for said period. The list of scaling reports will be delivered to the Seller as soon as practicable following the closing of each week of deliveries. The weekly deliveries run from Sunday to Saturday. These scaling reports will also form part of the payment summary provided to the Seller. Payment summary reports will be delivered concurrently with any payments made pursuant to Section 9.3.4.
 - 9.3.2 The Seller, upon receipt, shall cross-check the payment summary report provided with its own internal listing and the Seller shall notify the Purchaser within thirty (30) days of payment of any discrepancies of loads missed and/or loads paid incorrectly, otherwise payment will be considered accurate and complete.
 - **9.3.3** All correction listings shall be faxed or emailed to the Purchaser addressed to the attention of the person responsible as per a list of such persons provided by the Purchaser to the Seller. The correction listing shall have sufficient supporting documentation as proof of the existing errors, being as a minimum a bill of lading from the Origin site and weigh slip (computer generated or manual) from the Destination, where applicable. All loads forming part of the correction listing shall be notified by the Seller to the Purchaser within thirty (30) days following the payment; upon which the Purchaser shall issue to the Seller payment for the listed corrections on the next pay period.
 - 9.3.4 Subject to the foregoing, payments for each delivery will be made within thirty (30) days of end of the week in which that delivery is made.
- 9.4 If Purchaser fails to pay, as and when due, any amount owing hereunder, then Purchaser shall pay interest on such amount from the due date up to and including the date when such amount and all interest thereon are paid in full at the rate per annum equal to the rate of interest commonly known and referred to as the prime rate of the Royal Bank of Canada on the date of default plus 2%. Such interest shall be payable on demand.

10. DISPUTE RESOLUTION

10.1 <u>Resolution Procedure</u>. Any dispute between the parties, including (a) with respect to the interpretation of any provision of this Agreement, (b) in connection with any analysis or sampling performed under Section 7, (c) as to the Price for Chips and Residual Fibre delivered after the Initial Contract Year or (d) with respect to the performance by the parties of their obligations hereunder (a "**Dispute**"), shall first be submitted to the informal dispute resolution process as provided in this Section 10, prior to either party exercising any judicial recourses except as otherwise provided for in this Section 10.

- 10.2 <u>First Level</u>. The business manager of a party (each, a "**Business Manager**"), as designated by such party, shall notify his counterpart of any Dispute that requires resolution. The Business Managers shall diligently negotiate in good faith in an attempt to resolve the Dispute within ten (10) Business Days of such notice. Failing agreement on the resolution of the Dispute, either Business Manager may submit the Dispute to each of the parties' respective group vice-president (each, a "**Group Vice-President**") as designated by such party and their legal counsel (collectively, the "**Second Level Recipients**") for resolution (the "**Second Level Dispute Resolution**") in the manner provided for in Section 10.3.
- 10.3 <u>Draft written summary of Dispute</u>. A Business Manager (the "**Initiator**"), prior to initiating the Second Level Dispute Resolution, must provide his counterpart with a written draft notice which is intended to be sent to each of the Second Level Recipients in accordance with Section 10.2 and which shall contain a clear and concise description of (a) the Dispute together with an estimate of the dollar value or, as the case may be, the principle involved, (b) the efforts made by each party's Business Manager to come to a resolution, (c) the position taken by each party's Business Manager, (d) the proposal(s) of the Business Managers to resolve the Dispute, and (e) a request asking that the Group Vice-President of the Initiator contact his counterpart for the purpose of coming to a resolution of the Dispute in accordance with Section 10.2.
- 10.4 <u>Submission to Second Level</u>. Upon receipt of the written draft notice from the Initiator referred to in Section 10.3, the recipient Business Manager shall have five (5) Business Days to provide, in good faith, his comments in writing to the Initiator as to changes or amendments required to clarify items (a) to (d) referred to in Section 10.3. Upon receipt of the comments, the Initiator may (A) amend the draft notice to reflect the reflected changes and forward the notice in final form to each of the Second Level Recipients, or (B) forward to each of the Second Level Recipients the draft notice together with the written comments received from his counterpart.
- 10.5 <u>Second Level Dispute Resolution</u>. The Group Vice-Presidents shall negotiate in good faith in attempt to resolve the Dispute within five (5) Business Days of receipt of the notice and written comments, as the case may be, referred to in Section 10.4. Failing resolution of the Dispute by the Group Vice-Presidents within the said five (5) Business Days or any other time as the parties may mutually agree, the parties shall each be free to refer the Dispute to a single arbitrator under the *Arbitration Act* (Ontario). The award of the arbitrator will be final and binding on the parties.
- 10.6 <u>General Provisions for Informal Dispute Resolution</u>. At each stage of the informal dispute resolution process provided for in this Section 10.6, the parties agree as follows:
 - **10.6.1** The aforementioned designated representatives shall meet as often as the parties reasonably deem necessary and shall provide to the other party all non-privileged information necessary to the resolution of the Dispute, provided that such other party shall maintain the confidentiality of any Confidential Information provided to it and shall only use such Confidential Information for the purpose of resolving the Dispute.
 - **10.6.2** During the course of negotiations, all reasonable requests made by one party to another for non-privileged information, reasonably related to this Agreement, shall be honoured in order that each of the parties may be fully advised of the other party's position, provided that the Recipient shall maintain the confidentiality of

any Confidential Information provided to it and shall only use such Confidential Information for the purpose of resolving the Dispute.

- 10.6.3 The provisions of this Section 10 shall not be construed to prevent a party from instituting, and a party is authorized to institute, formal proceedings at any stage of the Dispute resolution to avoid the expiration of any applicable limitations period, to preserve a superior position with respect to other creditors or to seek immediate injunctive relief if immediate injunctive relief is determined by either party as necessary in accordance with Section 10.6.4.
- 10.6.4 The parties agree that Disputes between them shall not be subject to the informal dispute resolution process provided for in this Section 10 where a party determines, acting reasonably and in good faith, that a breach of the terms of this Agreement by the other party will cause immediate, large, severe and irreparable damages and that a temporary restraining order or other immediate injunctive relief is the only adequate remedy.
- 10.7 <u>Continuing Performance</u>. Each party shall continue performing its obligations under this Agreement while any Dispute is being resolved under the dispute resolution procedures provided for in this Section 10.
- 10.8 In the event that Seller and Purchaser are unable to agree on the Price for Chips and/or Residual Fibre, they shall use their commercially reasonable efforts to resolve any difference of view between them in accordance with the provisions of this Section 10. If the matter is referred to a single arbitrator under Section 10.5, the arbitrator ("**Price Arbitrator**") will be instructed to establish the Price for the Chips and/or the Residual Fibre for the given Contract Year under Dispute. **[**]**
- 10.9 The Price Arbitrator shall be instructed to decide within thirty (30) days of being appointed and shall be given access to all materials and information, other than privileged materials and information, requested by it for such purpose. The procedures to be followed by the Price Arbitrator shall be determined by the Price Arbitrator in his or her discretion but shall include an opportunity for each of the Seller and the Purchaser to submit evidence and argument and to respond to the other party's evidence and argument. The decision of the Price Arbitrator with respect to the Price shall be final and binding on each of Seller and Purchaser and shall not be subject to appeal by any party. The fees and disbursements charged by the Price Arbitrator shall be borne in equal parts by Seller and Purchaser.
- 10.10 During the pendency of any dispute resolution proceeding under this Section 10, the Price applicable for the Chips and/or the Residual Fibre shall be the Price for such Chips and/or Residual Fibre that was in effect at the time the notice requesting adjustment was received. Once the Price for the Chips and/or the Residual Fibre has been determined by the Price Arbitrator, any adjustment to the Price of the Chips and/or the Residual Fibre shall be retroactive to the shipments of Chips and/or Residual Fibre in the Contract Year for which the revised Price is applicable. Upon determination of the Price for the Chips and/or Residual Fibre, any monies due by one party to the other as a result of the retroactive application of the Price for the Chips and/or Residual Fibre shall be paid within ten (10) Business Days.

- 10.11 The dispute resolution procedures shall be kept confidential and the existence of the procedures and any element of it (including any documents submitted or exchanged, any submissions to the Price Arbitrator and any Price Arbitrator determination) shall not be disclosed beyond the Price Arbitrator, the parties, their respective Representatives and any Person necessary to the conduct of the procedures, except as required by applicable law.
- 10.12 Nothing in this Section 10 shall limit the right of any party to seek extraordinary recourses in courts of competent jurisdiction, including, but not limited to, specific performance, seizures before judgment and injunctions, unless the recourse sought conflicts with an resolved Dispute finally resolved in accordance with this Section 10.

11. FORCE MAJEURE

- 11.1 <u>Affecting the Purchaser</u> If, at any time, the operation of the pulp production facilities at the Purchaser's Mill is shut down or curtailed, or the Purchaser or the Purchaser's Mill is prevented from taking delivery of Chips or Residual Fibre, due to weather conditions, forest fire, fire, pandemic, sabotage, strike, civil disobedience and protest actions, riot, war, flood, explosion, laws or regulations, court order, act of any government body or agency, act of God, unexpected equipment breakage despite diligent maintenance work being carried out (each a "Force Majeure"), then the Purchaser may, by giving notice to the Seller, discontinue or curtail the purchase and acceptance of deliveries hereunder from the Seller for the period of such shutdown or curtailment without liability. For clarity, any event within the control of the Purchaser, such as voluntary reduction in workforce, strike, labour disruptions, lockout or closure because of market conditions is not and shall not be deemed a Force Majeure.
- 11.2 <u>Purchases of Chips From Third Parties.</u> Except in circumstances where the Purchaser is prevented from taking delivery and physical possession of Chips or Residual Fibre purchased hereunder but not so prevented from taking delivery of other wood chips or residual fibre from other sources, the Purchaser will not purchase or otherwise take delivery of any wood chips or residual fibre except from the Seller during a period of discontinuance or curtailment of Chips pursuant to Section 11.1.
- 11.3 <u>Curtailment On a Mill-by-Mill Basis.</u> If the Purchaser curtails its purchases pursuant to Section 11.1, the Seller has the right to determine which the Seller's Mills will be first affected.
- 11.4 <u>Co-Operation</u>. During each period of shutdown or curtailment the Purchaser will assist and co-operate with the Seller to minimize the detrimental effects of such shutdown or curtailment on the Seller's Mills except that the Purchaser will not be required to incur any cost in respect thereof.
- 11.5 <u>Affecting the Seller</u>. If, at any time, the Seller shuts down or curtails the operation of any or all of the Seller's Mills or is otherwise unable to deliver Chips or Residual Fibre, due to a Force Majeure, then the Seller may, by giving notice to the Purchaser, discontinue or curtail the production or delivery of Chips or Residual Fibre to the Purchaser without liability. For clarity, any event within the control of the Seller, such as voluntary reduction in workforce or closure because of market conditions is not and shall not be deemed a Force Majeure.
- 11.6 If the Purchaser curtails its purchases pursuant to Section 11.1 or if the Seller curtails its production pursuant to Section 11.5, then that party shall:

- 11.6.1 immediately after the occurrence of such Force Majeure give notice to the other party informing it of such Force Majeure, and, insofar as known, the probable duration and extent to which such party will be unable to perform or be delayed in performing its obligations hereunder;
- **11.6.2** take whatever action may be appropriate and/or reasonable to mitigate the extent or duration of the interruption or delay onto the other party; and

11.6.3 give prompt notice to the other party when such cause or causes have been eliminated or remedied.

- 11.7 If the Seller gives notice under Section 11.5 and the discontinuance or curtailment of the production of Chips or Residual Fibre from a particular Seller's Mill is for a continuous period in excess of sixty (60) calendar days, the Purchaser may suspend this Agreement for the remainder of the Contract Year ("**Suspended Period**") as it relates to that Seller's Mill and enter into an agreement for the supply of Chips or Residual Fibre, as the case may be, with another supplier for the Suspended Period by serving written notice to Seller. The term of the agreement with the other supplier may not exceed the end of that Contract Year. Provided the Force Majeure has been eliminated or remedied, the performance of this Agreement as it relates to that Seller's Mill shall resume at the start of the Contract Year following the Suspended Period. During the Force Majeure period and the Suspended Period, those volumes not supplied or purchased by a Seller or Purchaser shall be deducted from the Committed Chips Volume and/or the Committed Residual Fibre Volume as the case may be in such applicable Contract Year.
- 11.8 <u>Alternative Disposition.</u> If, as provided in Section 11.1, the Purchaser discontinues or curtails the purchase of Chips or Residual Fibre, the Seller may dispose of such Chips or Residual Fibre during such discontinuance or curtailment, except that:
 - 11.8.1 the Seller will, before entering into any agreement to dispose of such Chips, or Residual Fibre notify the Purchaser of its intention to do so, and the Seller will not dispose of such Chips or Residual Fibre if the Purchaser notifies the Seller, within fifteen (15) days after the date of receipt of such notice, that the Purchaser will purchase such Chips or Residual Fibre upon production on condition, at the Purchaser's option, that they are stored at a the Seller' Mills until the end of the discontinuance or curtailment, except that if Chip or Residual Fibre inventories at any the Seller's Mills at which such Chips or Residual Fibre are stored during such discontinuance or curtailment materially impair the Seller's operations, the Seller may, upon prior written notice to the Purchaser (such notice allowing reasonable time for the Purchase to arrange Delivery), notwithstanding any commitment made by the Purchaser under this Section 11.8, dispose of and deliver to a third party such volume of Chips or Residual Fibre as is necessary to relieve such material impairment (and promptly refund to the Purchaser any amounts paid by the Purchaser for such Chips or Residual Fibre);
 - 11.8.2 the Seller will not enter into any agreement concerning the disposition of Chips or Residual Fibre that would conflict with the Purchaser's right to resume the purchase of Chips or Residual Fibre under this Agreement. For greater certainty, the term of any agreement to dispose of such Chips or Residual Fibre may not exceed the end of the Contract Year; and

- 11.8.3 the Purchaser may, after the end of such discontinuance or curtailment, purchase any Chips or Residual Fibre that have not been disposed of and which are stockpiled, by delivering not less than fourteen days' notice to the Seller of the quantity of such stockpiled Chips or Residual Fibre that it wishes to purchase under this Agreement stipulating a rate of delivery of Chips that will not interfere with the efficiency of the Seller's Mills at which such Chips or Residual Fibre are located and confirming that the Purchaser will take the risk of any impairment of the quality of the Chips resulting from the stipulated rate of delivery.
- 11.9 Notwithstanding the foregoing, nothing in Section 11.1 shall relieve a party of the obligation to make any pecuniary payment in accordance with the terms of this Agreement.

12. Termination

- 12.1 This Agreement may be terminated, without prejudice to any other rights and remedies available hereunder, at law or in equity:
 - 12.1.1 by written notice from a party in the event of the bankruptcy of the other party or the making of an assignment for the benefit of creditors, or the appointment of a trustee or receiver and manager or liquidator for such party or for all or a substantial part of its property, or the commencement of bankruptcy, reorganization, arrangement, insolvency or similar proceedings by or against such party under the laws of any jurisdiction;
 - 12.1.2 by the mutual consent of the parties;
 - 12.1.3 at any time while an Event of Default exists with respect to a party, by written notice by the other party specifying, as the Termination Date, a date not earlier than the date on which such notice is given; or
 - 12.1.4 after the Initial Term, with a two year prior written notice from a party to the other (a "Termination Notice"). For clarity purposes, this Agreement shall terminate on the second anniversary of the last day of the Contract Year during which a party has delivered a Termination Notice to the other party.

13. event of DEFAULT

- 13.1 An event of default will exist with respect to a party if such party has committed a material default in the performance of its obligations hereunder (for the purposes of this Section 13.1, the failure by the Purchaser to make three payments required under this Agreement within any three month period, other than a failure arising from a *bona fide* dispute over the requirement to make such payment, will be deemed to be a material default) (each, an "**Event of Default**"). Upon the occurrence of an Event of Default, a party may send a notice to the defaulting party specifying the Event of Default and declaring the intention of the non-defaulting party to exercise its right under Section 12.1.3 if the default is not remedied within:
 - 13.1.1 ten (10) Business Days since the date that the notice is delivered to the defaulting party in respect of a failure to make three payments required under this Agreement within any three month period, other than a failure arising from a *bona fide* dispute over the requirement to make such payment; or

13.1.2 thirty (30) Business Days since the date that the notice is delivered to the defaulting party for all other Events of Default, provided that if the defaulting party, acting reasonably, is diligently working at remedying the default, this 30-day period will be extended for such appropriate time to allow the defaulting party to remedy the default.

14. Confidentiality

- 14.1 The Recipient will not use the Confidential Information except in accordance with the terms of this Agreement. The Recipient will not disclose (directly or indirectly) Confidential Information to any Person (except on a confidential basis to its Representatives and advisors who need to know and who are legally obligated to preserve the confidentiality thereof). The Recipient will employ reasonable steps to preserve the confidential information of similar nature. The Recipient undertakes and agrees to inform any person who receives Confidential Information of the confidential nature of such Confidential Information, instruct any such person to treat such Confidential Information in the same manner and to the same extent as is required of it under this Agreement and to inform such person of the existence of this Agreement. Recipient will be responsible for any breach of this Agreement due to any act or omission by its Representatives.
- 14.2 Notwithstanding Section 14.1, the Recipient may disclose such Confidential Information as required by court order or other lawful process (including pursuant to the rules of any applicable securities self-regulatory organization or stock exchange). In all such cases, the Recipient will disclose Confidential Information only to the extent required to fulfill such purpose or legal requirement. In the event the Recipient becomes legally compelled to disclose any Confidential Information, the Recipient will promptly notify the Disclosing Party of such fact so that the Disclosing Party may seek an appropriate remedy to prevent such production, and request the person demanding such production to allow the Disclosing Party a reasonable period of time to seek such remedy.
- 14.3 Upon the earlier of (a) a written request at any time by the Disclosing Party or (b) termination of this Agreement, all Confidential Information, together with any copies thereof in any media, shall be promptly returned by the Recipient to the Disclosing Party or destroyed, and the Recipient shall make no further use of the Confidential Information and shall promptly certify in writing to the Disclosing Party that it has complied with these obligations. The return of any Confidential Information will not relieve the Recipient of its obligation to maintain the confidentiality of the Confidential Information in accordance with Section 14.4. Within 10 days of the receipt by the Recipient of a written request from the Disclosing Party, the Recipient shall certify to the Disclosing Party in writing its compliance with this Section. Notwithstanding the foregoing provisions of this Section, the Recipient may retain Confidential Information on electronic back-up systems to which access within the organization of the Recipient is limited to purposes of backup and contingency planning, provided that Confidential Information so retained shall remain subject to the confidentiality obligations of this Agreement.
- 14.4 Notwithstanding the termination or expiration of this Agreement, the obligations set forth in this Section 14 shall survive for a period of three (3) years from the last date that Confidential Information is provided to Recipient by or on behalf of the Disclosing Party.

15. ASSIGNMENT

15.1 No party may assign its rights or obligations under this Agreement without the prior written consent of the other party except that either party may assign its rights and obligations under this Agreement:

15.1.1 to any of its Affiliates;

15.1.2 as a result of the granting by a party of a security interest over all or substantially all of its assets, including this Agreement; or

15.1.3 in accordance with Sections 15.2 or 15.3.

No assignment shall relieve the assigning party of its obligations under this Agreement unless such relief is expressly consented to by the other party. Any purported assignment made in contravention of this Agreement shall be null and void and have no legal effect.

- 15.2 If the Purchaser sells, transfers or otherwise disposes (other than by way of a security interest) of the Purchaser's Mill to a new owner to be used for pulp or paper purposes, then, as a condition of such transaction, the Purchaser will assign all its interest, and the new owner will assume all obligations under this Agreement and the new owner will enter into an agreement with the Seller confirming that the new owner is bound by this Agreement (with such changes as may be required to contemplate the addition or substitution of the Purchaser as a party).
- 15.3 If the Seller sells, transfers or otherwise disposes (other than by way of a security interest) of a Seller's Mill (a "**Divested Mill**") to a new owner to be used for sawmill purposes, then, as a condition of such transaction, the Seller will cause the new owner to agree to be bound by terms substantially similar to this Agreement as relating to the Divested Mill, such agreement to be evidenced by the new owner entering into such formal agreements as the Purchaser may reasonably require and, upon the completion of the sale, this Agreement will be amended to remove the Divested Mill and the volume of Chips and Residual Fibre produced by such Divested Mill.

16. Option to purchase

- 16.1 [**]
- 16.2 [**]
- 16.3 [**]
- 16.4 [**]
- 16.5 [**]

17. NOTICE

Any notice, request for consent, direction or other communication given regarding the matters contemplated by this Agreement must be in writing (regardless of the fact that a specific

provision of this Agreement specifies or fails to specify that a particular notice will be in writing), sent by personal delivery, courier or facsimile (but not by electronic mail) and addressed:

(a) to the Seller at:

9437-6001 QUEBEC INC. 1 Place Ville-Marie, Suite 2500, Montreal (Quebec) H3B 1R1

Attention: Michael Liggett Email: •

(b) to the Purchaser at:

RAYONIER A.M. CANADA G.P.
10, chemin Gatineau
C.P. 5000
Témiscaming, Québec
J0Z 1G0
Attention: Regional Fibre Manager
Email: •
With a copy to:
Attention: •

Email:

A notice is deemed to be delivered and received (a) if sent by personal delivery, on the date of delivery if it is a Business Day and the delivery was made prior to 4:00 p.m. (local time in place of receipt) and otherwise on the next Business Day, (b) if sent by same-day service courier, on the date of delivery if sent on a Business Day and delivery was made prior to 4:00 p.m. (local time in place of receipt) and otherwise on the next Business Day, or (d) if sent by email, on the date it is sent by such means. A party may change its address for service from time to time by providing a notice in accordance with the foregoing. Any subsequent notice must be sent to the party at its changed address. Any element of a party's address that is not specifically changed in a notice shall be assumed not to be changed. Sending a copy of a notice to a party's legal counsel as contemplated above is for information purposes only and does not constitute delivery of the notice to that party. The failure to send a copy of a notice to legal counsel does not invalidate delivery of that notice to a party.

18. INSPECTION

Subject to a forty-eight (48) hours prior written notice to the Seller, the Purchaser shall have the right, during the Term of this Agreement, to access the Seller's Mills during regular business hours to inspect its equipment and tools for the manufacture of Chips and Residual Fibre. The Purchaser undertakes to not to interfere with the Seller's operations during the inspections provided for in this Section 18.

19. INDEMNIFICATION

- 19.1 <u>Indemnification by Seller</u>. The Seller will indemnify, defend and save harmless the Purchaser and its Affiliates and their respective Representatives from any Losses suffered by, imposed upon or asserted against any of the Purchaser indemnitees as a result of, in respect of, connected with, or arising out of, under, or pursuant to: (a) any breach, default or violation or any representation and warranty of the Seller contained in this Agreement and (b) any failure of the Seller to perform or fulfil any covenant of the Seller contained in this Agreement.
- 19.2 <u>Indemnification by Purchaser</u>. The Purchaser will indemnify, defend and save harmless the Seller and its Affiliates and their respective Representatives from any Losses suffered by, imposed upon or asserted against any of the Seller indemnitees as a result of, in respect of, connected with, or arising out of, under, or pursuant to: (a) any breach, default or violation or any representation and warranty of the Purchaser contained in this Agreement and (b) any failure of the Purchaser to perform or fulfil any covenant of the Seller contained in this Agreement.
- 19.3 <u>No right to Set-Off</u>. Notwithstanding anything contained in this Agreement or in the Purchase Agreement to the contrary, each party shall fully pay to the other party all amounts required to be paid by it under this Agreement and shall not set-off, counterclaim or otherwise withhold any amount owed or claimed to be owed to the other party under this Agreement on account of any obligation owed or claimed to be owed by the other party, whether under the Purchase Agreement or otherwise and whether or not such obligation has been finally adjudicated, settled or otherwise agreed upon by the parties in writing.
- 19.4 <u>Enforcement of Agreement</u>. Each of the parties acknowledges and agrees that a party could be damaged irreparably in the event certain of the provisions of this Agreement are not performed in accordance with their specific terms and that any breach of this Agreement by the other party may not be adequately compensated by monetary damages. Accordingly, each party agrees that, in addition to any other right or remedy to which a party may be entitled in such circumstances, at law or in equity, it shall be entitled to enforce any provision of this Agreement by a decree of specific performance and to temporary, preliminary and permanent injunctive relief to prevent breaches or threatened breaches of the provisions of this Agreement, without posting any bond or other undertaking. The rights and remedies of the parties to this Agreement are cumulative and not alternative.

20. GENERAL PROVISIONS

- 20.1 <u>Governing Law</u>. This Agreement will be governed by and interpreted and enforced in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein.
- 20.2 <u>Independent Contractors</u>. This Agreement is an agreement to purchase Chips and Residual Fibre; it should not be interpreted in such a way as to create a relationship of agent, franchise, joint venture or partnership of any kind or nature between the parties. The parties hereto are considered independent contractors, and employees of one are not considered as employees of the other.
- 20.3 <u>Successors and Assigns</u>. This Agreement becomes effective only when executed by all of the parties. After that time, it will be binding upon and enure to the benefit of each of the parties and their respective successors and permitted assigns.

- 20.4 <u>Gender and Number</u>. Any reference in this Agreement to gender includes all genders. Words importing the singular number only shall include the plural and vice versa.
- 20.5 <u>Headings</u>. The division of this Agreement into Sections and Schedules and the insertion of headings are for convenient reference only and are not to affect its interpretation.
- 20.6 <u>Severability</u>. If any provision of this Agreement is determined to be illegal, invalid or unenforceable by an arbitrator or any court of competent jurisdiction, that provision will be severed from this Agreement and the remaining provisions shall remain in full force and effect.
- 20.7 <u>Cumulative Remedies</u>. All rights and remedies provided in this Agreement are cumulative and not exclusive, and are in addition to and without prejudice to any other rights or remedies available at law, in equity, by statute or otherwise, and the exercise by either party of any right or remedy does not preclude the exercise of any other rights or remedies that may now or subsequently be available at law, in equity, by statute, in any other agreement between the parties or otherwise.
- 20.8 <u>No Waiver</u>. No waiver of any of the provisions of this Agreement will constitute a waiver of any other provision (whether or not similar). No waiver will be binding unless executed in writing by the party to be bound by the waiver. A party's failure or delay in exercising any right under this Agreement will not operate as a waiver of that right. A single or partial exercise of any right will not preclude a party from any other or further exercise of that right or the exercise of any other right.
- 20.9 <u>Discussions and Explanations</u>. The parties acknowledge that all provisions of this Agreement have been freely discussed and negotiated between them and that they have received an independent legal opinion or adequate explanations on the nature and extent of the provisions of this Agreement.
- 20.10 <u>Entire Agreement</u>. This Agreement constitutes the entire agreement between the parties with respect to the transactions contemplated by this Agreement and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the parties. There are no representations, warranties, covenants, conditions or other agreements, express or implied, collateral, statutory or otherwise, between the parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement. The parties have not relied and are not relying on any other information, discussion or understanding in entering into and completing the transactions contemplated by this Agreement.
- 20.11 <u>Amendments</u>. This Agreement may only be amended, supplemented or otherwise modified by written agreement signed by the parties.
- 20.12 <u>Incorporation of Schedules</u>. Any schedule attached to this Agreement is an integral part of this Agreement for all purposes of this Agreement. All other schedules that could be added to this Agreement shall be subject to a written agreement signed by the parties.
- 20.13 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (including counterparts by facsimile or PDF) and all such counterparts taken together shall be deemed to constitute one and the same instrument.

[Signature page follows]

IN WITNESS WHEREOF this Agreement was executed on the date first hereinabove mentioned.

PURCHASER

SELLER

RAYONIER A.M. CANADA G.P., herein represented by its managing partner, **Rayonier A.M. Canada Industries Inc.**

9437-6001 QUEBEC INC.

	Per:	
Name:	Name:	
Title:	Title:	
	Per:	
Name:	Name:	
Title:	Title:	

Schedule A Key Terms and Conditions

[**]

Schedule B CHIP SPECIFICATIONS

Exhibit 10.1

Schedule C

Schedule D

[**]

EXHIBIT B

THIS NON-COMPETITION AGREEMENT is dated April [], 2021 and made between:

- (1) 9437-6001 QUEBEC INC., a corporation incorporated under the laws of the Province of Québec (the "Purchaser");
- (2) GREENFIRST FOREST PRODUCTS INC., a corporation incorporated under the laws of British Columbia ("Itasca");
- (3) **RAYONIER A.M. CANADA G.P.**, a general partnership formed under the laws of Quebec ("**RYAM GP**"); and
- (4) **RAYONIER A.M. CANADA INDUSTRIES INC.**, a corporation incorporated under the laws of Canada ("**RYAM Industries**"); and together with RYAM GP, the "**Seller**")

RECITALS:

- (A) Pursuant to an asset purchase agreement dated April [], 2021 (the "Purchase Agreement") between RYAM GP, RYAM Industries, the Purchaser and Itasca (as parent guarantor to the Purchaser), RYAM GP and RYAM Industries have agreed to sell to the Purchaser and the Purchaser has agreed to purchase from RYAM GP and RYAM Industries, the Purchased Assets.
- (B) The Seller has derived, and will derive, substantial economic benefits from the transactions contemplated in the Purchase Agreement and would not have sold the Purchased Assets and would not have entered into the other transactions and agreements contemplated by the Purchase Agreement without a commitment by the Purchaser to execute and deliver this Agreement.
- (C) So that the Purchaser may realize the full value associated with its purchase of the Purchased Assets pursuant to the Purchase Agreement, the Seller has agreed that it will refrain from competing with the Purchaser, all in accordance with the terms of this Agreement.
- (D) The Seller has voluntarily entered into this Agreement and agrees that the limitations and restrictions set out in this Agreement are reasonable and not oppressive and are intended to protect the Purchaser's substantial investment and legitimate business interests under the Purchase Agreement.
- (E) The Purchaser would not have acquired the Purchased Assets and would not have entered into the other transactions and agreements contemplated by the Purchase Agreement without a commitment by the Seller to execute and deliver this Agreement.

NOW THEREFORE in consideration of the mutual covenants and agreements contained in this Agreement and other good and valuable consideration (the receipt and sufficiency of which are acknowledged), the Parties agree as follows.

Article 1 Interpretation

1.1 Definitions

In this Agreement the following words have the following meanings:

"Affiliate" means, as applied to a Person, (a) any other Person that directly or indirectly controls, is controlled by, or is under common control with, that Person, or (b) any director, officer or partner of such Person. For the purposes of this definition, "control" (including with correlative meanings, the terms "controlled by", and "under common control with") as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through ownership of voting securities, by contract or otherwise.

"Agreement" means this non-competition agreement as it may be amended, restated, replaced or supplemented from time to time.

"Connected Person" means (a) an Affiliate of the Seller, or (b) a Person being or acting as employee, independent contractor, director, officer of, or for, the Seller.

"Customers" means all Persons who are at this date or were at any time during the past two years customers or clients of the Seller.

"**Operations**" means the production, distribution, marketing and sale of lumber and newsprint and the operation of sawmills as it is operated by RYAM GP and RYAM Industries on the date hereof.

"Parties" mean the Seller and the Purchaser.

"Prospective Customers" means all Persons canvassed or solicited for purposes of becoming customers or clients of the Seller at any time during the two year period prior to the date of this Agreement.

"Purchase Agreement" has the meaning specified in the Recitals.

"**Restricted Period**" means the period commencing on the date of this Agreement and ending on the date which is five years after the date of this Agreement.

"Restricted Territory" means the Provinces of Ontario and Quebec.

1.2 Other Defined Terms

Capitalized terms used in this Agreement without definition have the respective meanings specified in the Purchase Agreement.

1.3 Gender and Number

In this Agreement, unless there is something in the subject matter or context inconsistent therewith, (a) words in the singular number include the plural and vice versa, and (b) words importing the use of any gender include all genders where the context or party referred to so requires, and the rest of the sentence is to be construed as if the necessary grammatical changes had been made.

1.4 Certain Phrases

In this Agreement (a) the words "including" and "includes" mean "including (or includes) without limitation", and (b) in the computation of periods of time from a specified date to a later specified date, unless otherwise expressly stated, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding" and if the last day of any such period is not a Business Day, such period will end on the next Business Day.

1.5 Headings, Etc.

The division of this Agreement into articles and sections and the insertion of headings are for convenient reference only and are not to affect the interpretation of this Agreement.

1.6 Governing Law

This Agreement is governed by and is to be interpreted in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein. Each of the Parties irrevocably attorns and submits to the exclusive jurisdiction of the courts of Ontario.

Article 2 Non-Completion and Other Covenants

2.1 Recitals

The Purchaser and the Seller represent and warrant to each other that the recitals set forth above are true in substance and in fact and that the purpose of this Agreement is to protect the legitimate interests of the Purchaser by providing, *inter alia*, for the broadest scope, the longest duration and the widest territory permitted by law.

2.2 Non-Competition

- (a) During the Restricted Period, the Seller shall not, on its own behalf or on behalf of, or together with, any other Person, directly or indirectly, in any capacity whatsoever (including by or through a Connected Person):
 - carry on, be engaged in, have any financial or other interest in, or be otherwise commercially involved in, any endeavour, activity or business in all or any part of the Restricted Territory which is the same as, or substantially similar to, or competes with the Operations; or

- (ii) advise, assist, invest in, lend money to, guarantee the debts or obligations of any Person that carries on any endeavour, activity or business in all or any part of the Restricted Territory which is the same as, or substantially similar to, or competes with the Operations.
- (b) The Parties hereto agree and acknowledge that: (i) the restrictive covenants are undertakings of the Seller not to provide, directly or indirectly, property or services in competition with the property or services provided or to be provided by the Purchaser in the course of carrying on the Operations; (ii) for the purposes of paragraph 56.4(7) (d) of the *Income Tax Act* (Canada) and any provincial equivalent, no proceeds shall be received or receivable by the Seller for granting the restrictive covenants; and (iii) the restrictive covenants are integral to the Purchase Agreement and have been granted to maintain or preserve the fair market value of the Purchased Assets.

2.3 Non-Solicitation of Customers

Without limiting the generality of Section 2.2 of this Agreement, during the Restricted Period, the Seller shall not, on its own behalf or on behalf of, or together with, any other Person, directly or indirectly, in any capacity whatsoever (including by or through a Connected Person):

- (a) canvass, call upon, solicit the business of, or have any contact with, any Customer or Prospective Customer;
- (b) accept any business from any Customer or Prospective Customer; or
- (c) supply any goods or services to any Customer or Prospective Customer,

in each case, in connection with an activity, endeavour or business which is the same as, or substantially similar to, or competes with the Operations in the Restricted Territory.

2.4 Non-Solicitation of Employees

The Seller acknowledges that the employees of the Seller who have become employees of the Purchaser are critical to the success of the Operations and that it is necessary to protect the Purchaser from the loss of such employees. Accordingly, during the Restricted Period, the Seller shall not, on its own behalf or on behalf of, or together with, any other Person, directly or indirectly, in any capacity whatsoever (including by or through a Connected Person):

- (a) employ, offer employment to, solicit the employment of, or otherwise entice away from the employment of the Purchaser, any individual who is currently employed in the Operations; or
- (b) assist any Person to employ, offer employment to, solicit the employment of, or otherwise entice away from the Operations, any such individual.
 - 3

2.5 Non-Interference

During the Restricted Period, the Seller shall not (a) request or advise any customer, client or supplier of the Purchaser or Operations to cancel, curtail or otherwise adversely change, such customer's, client's or supplier's patronage of, or relationship with, the Purchaser or the Operations (it being acknowledged and agreed that the Seller may have other business dealings with such customers, clients and suppliers which do not complete with the Operations and will, in connection with such business dealings, correspond and communicate with such customers, clients and suppliers), or (b) make public statements or public announcements which are reasonably likely to adversely affect the public standing of the Purchaser or the Operations.

2.6 Limitations

- (a) Notwithstanding Sections 2.2 to 2.5 of this Agreement, (i) the ownership by the Seller of equity securities of the Purchaser or Itasca and (ii) a passive equity investment by the Seller in any Person the equity securities of which are listed on a recognized stock exchange, where the equity investment does not in the aggregate exceed 3% of the issued equity shares of that Person, shall not be a breach or contravention of this Agreement.
- (b) Notwithstanding Section 2.2 of the Agreement or anything to the contrary, nothing shall prohibit the Seller from (i) exercising any and all of its rights under the chip and residual fibre purchase agreement entered into among RYAM GP, the Purchaser and Itasca on the date hereof, including the Seller's option to purchase provided in Section 16 of the said agreement and (ii) operating the mill located at 495 Highway 111 West, La Sarre, Québec, J07 2X6 (the LaSarre Mill) or the mill located at 67 rue Principale Sud or Route 397, Béarn, Québec, J0Z 1G0 (the Béarn Mill) should the Seller's option to purchase mentioned in paragraph (i) be exercised.
- (c) The restrictions set forth at Section 2.4 with respect to the non-solicitation of employees will not be deemed to be violated by virtue of general advertisements, solicitations (including third party recruitment contacts), searches or other broad based hiring or engagement methods not specifically targeted or directed to employees or nonemployee consultant/contractor of the Operations.

Article 3 Remedies

3.1 Reasonableness

The Seller (a) has carefully considered with the assistance of its counsel the nature and extent of the restrictive covenants set forth in this Agreement, (b) expressly acknowledges that this Agreement is reasonable in all respects, and (c) irrevocably waives (and agrees not to raise) as a defence any issue of reasonableness (including the reasonableness of the Restricted Territory or the duration and scope of this Agreement) in any proceeding to enforce the provision of this Agreement.

3.2 Injunctive Relief

The Parties acknowledge and agree that the Purchaser will suffer irreparable harm if the Seller breaches any of its obligations set out in this Agreement and that monetary damages will be inadequate compensation. Accordingly, the Seller agrees that, in the event of a breach or threatened breach by it of any of the provisions of this Agreement, the Purchaser shall be entitled to an interim injunction, interlocutory injunction and permanent injunction for the purposes of preventing or restraining any such breach.

3.3 Other Remedies

The right of the Purchaser to injunctive relief and an accounting for profits shall be in addition to any and all other remedies available to it and shall not be construed to prevent it from pursuing, either consecutively or concurrently, any and all other legal or equitable remedies available to it including the recovery of monetary damages.

Article 4 Miscellaneous

4.1 Notices

Any notice, consent, waiver or other communication given under this Agreement must be in writing and shall be given by delivering it or sending it by facsimile or electronic mail addressed:

(a) to the Purchaser at:

Itasca Capital Ltd. 105 S. Maple St. Itasca, Illinois 60143

Attention: Facsimile: E-mail:

with a copy (which shall not constitute notice to the Purchaser) to:

the Purchaser's solicitors

Norton Rose Fulbright Canada LLP 222 Bay Street, Suite 3000, P.O. Box 53 Toronto, ON M5K 1E7

Attention:Walied Soliman and Trevor ZeylFacsimile:(416) 216-3930E-mail:walied.soliman@nortonrosefulbright.comandtrevor.zeyl@nortonrosefulbright.com

(b) to the Seller at:

Rayonier Advanced Materials Inc. 1301 Riverplace Boulevard Suite 2300 Jacksonville, Florida 32207 United States of America

Attention: Chief Financial Officer with a copy to: General Counsel

with a copy to:

McCarthy Tétrault LLP 1000, De La Gauchetière Street West Suite 2500 Montréal, Québec H3B 0A2

Attention: Clemens Mayr and Chrystelle Chevalier-Gagnon Email: <u>cmayr@mccarthy.ca</u> and <u>cchevaliergagnon@mccarthy.ca</u>

Any such communication is deemed to have been duly given (a) if delivered personally, on the day of delivery, (b) if sent by a nationally recognized courier service, on the later of (i) the first Business Day following the date of dispatch, or (ii) the scheduled day of delivery by such service, and (c) if sent by facsimile or electronic mail, on the day so sent if the day is a Business Day and it was sent prior to 5 pm (Toronto time) and otherwise on the next Business Day. A Person may change its address for service by notice given in accordance with the foregoing and any subsequent communication must be sent to such Person at its changed address.

4.2 Amendments

This Agreement may only be amended, supplemented or otherwise modified by written agreement of the Parties.

4.3 Waiver

The failure or delay by a Party in enforcing, or insisting upon strict performance of, any of the provisions of this Agreement does not constitute a waiver of such provision or in any way affect the validity or enforceability of this Agreement or deprive a Party of the right, at any time or from time to time, to enforce or insist upon strict performance of that provision or any other provision of this Agreement. Any waiver by a Party of any provision of this Agreement is effective only if in writing and signed by such Party.

4.4 Severability

The Parties agree that each of the covenants contained in Article 2 of this Agreement are separate and distinct. If any provision of this Agreement is determined by a court of competent jurisdiction to be illegal, invalid or unenforceable, that provision will be severed from this Agreement and the remaining provisions will continue in full force and effect, without amendment.

4.5 Time of the Essence

Time is of the essence of this Agreement.

4.6 Entire Agreement

This Agreement constitutes the entire agreement between the Parties with respect to the subject matter hereof and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the Parties.

4.7 Successors and Assigns

This Agreement will become effective when executed by the Parties and after that time will be binding upon and enure to the benefit of the Parties and their respective successors and permitted assigns. Neither this Agreement nor any of the rights, duties or obligations under this Agreement are assignable or transferable by a Party without the prior written consent of the other Party. Any attempt to assign any of the rights, duties or obligations in this Agreement without such written consent is void.

4.8 Further Assurances

From time to time after the completion of the transaction of purchase and sale contemplated in the Purchaser Agreement, each Party shall, at the request of the other Party, execute and deliver such additional assurances as may be reasonably required to effectively carry out the intent of this Agreement.

4.9 Counterparts and Electronic Delivery

This Agreement may be executed in any number of separate counterparts (including by electronic means) and all such signed counterparts will together constitute one and the same agreement. To evidence its execution of an original counterpart of this Agreement, a Party may send a copy of its original signature on the execution page hereof to the other Party by facsimile or other electronic means and such transmission (including in PDF form) with an acknowledgement of receipt shall constitute delivery of an executed copy of this Agreement to the receiving Party.

[Signature page follows.]

IN WITNESS WHEREOF the Parties have executed and delivered this Agreement.

By:	
	Authorized Signatory
By:	
	Authorized Signatory
GRE	ENFIRST FOREST PRODUCTS INC.
By:	
	Authorized Signatory
By:	
	Authorized Signatory
mana	DNIER A.M. CANADA G.P., herein represented by its aging partner, RAYONIER A.M. CANADA ISTRIES INC.
By:	
By:	Authorized Signatory
-	Authorized Signatory
-	Authorized Signatory Authorized Signatory
By:	
By:	Authorized Signatory ONIER A.M. CANADA INDUSTRIES INC.
By: RAY(By:	Authorized Signatory
By: RAY(By:	Authorized Signatory ONIER A.M. CANADA INDUSTRIES INC. Authorized Signatory
By: RAY(By: By:	Authorized Signatory ONIER A.M. CANADA INDUSTRIES INC. Authorized Signatory Authorized Signatory
By: RAY(By: By: RAY(Authorized Signatory ONIER A.M. CANADA INDUSTRIES INC. Authorized Signatory
By: RAY(By: By: RAY(Authorized Signatory DNIER A.M. CANADA INDUSTRIES INC. Authorized Signatory Authorized Signatory Authorized Signatory DNIER ADVANCED MATERIALS INC.
By: RAY(By: By: RAY(By:	Authorized Signatory ONIER A.M. CANADA INDUSTRIES INC. Authorized Signatory Authorized Signatory
By: By:	Authorized Signatory DNIER A.M. CANADA INDUSTRIES INC. Authorized Signatory Authorized Signatory Authorized Signatory DNIER ADVANCED MATERIALS INC.

(Signature Page for Non-Competition Agreement)

EXHIBIT C SERVICES AGREEMENT

EXHIBIT D

TRANSITION SERVICES AGREEMENT

THIS AGREEMENT (the "Agreement") is entered into on April •, 2021 between:

- (1) Rayonier A.M. Canada G.P., a general partnership formed under the laws of the Province of Québec (the "Seller"); and
- (2) 9437-6001 QUEBEC INC., a corporation incorporated under the laws of the Province of Quebec (the "Purchaser")

together, the "Parties" and each a "Party".

RECITALS

- (A) The Seller, Rayonier A.M. Canada Industries Inc., the Purchaser and GreenFirst Forest Products Inc. (as parent guarantor to the Purchaser) ("GreenFirst") entered into an asset purchase agreement dated April ●, 2021 (the "Purchase Agreement"), pursuant to which the Seller and Rayonier A.M. Canada Industries Inc. have agreed to sell, transfer and convey to the Purchaser, and the Purchaser has agreed to purchase from the Seller and Rayonier A.M. Canada Industries Inc., all of the Seller's and Rayonier A.M. Canada Industries Inc.'s right, title and interest in all of the property and assets relating to the Operations (other than the Excluded Assets) (the "Purchased Assets");
- (B) In connection with, and as a condition to, the consummation of the transactions contemplated by the Purchase Agreement, the Seller has agreed to provide to the Purchaser certain transition services to facilitate the transition of the ownership of the Purchased Assets and the Operations starting on the date hereof, pursuant to the terms and conditions set forth herein;

NOW THEREFORE, in consideration of the foregoing and the mutual agreements contained herein (the receipt and adequacy of which are acknowledged), the Parties agree as follows:

1. INTERPRETATION

1.1 Definitions

As used in this Agreement, defined terms will have the meaning given to such terms in the Purchase Agreement or as otherwise specified as follows:

"Additional Service" shall have the meaning specificed in clause 3.1(2);

"Change" shall have the meaning specified in clause 3.2(1);

"Change Request" shall have the meaning specified in clause 3.2(1);

"Confidential Information" shall have the meaning specified in clause 4.1;

"Dispute" shall have the meaning specified in clause 10;

"Fees" shall have the meaning specified in clause 3.4(1);

"Force Majeure Event" shall have the meaning specified in clause 7(4);

"GreenFirst" shall have the meaning specified in the preamble;

"Indemnities" shall have the meaning specified in clause 6(1);

"Losses" means any loss, liability, damage, cost or expense (including reasonable legal or other professional fees and expenses);

"Notice" shall have the meaning specified in clause 11.5;

"Parties" and "Party" shall have the meaning specified in the preamble;

"**Providers**" means collectively the Seller, its affiliates and the Third Party Providers that from time to time provide the Services;

"Purchase Agreement" shall have the meaning specified in the recitals;

"Purchased Assets" shall have the meaning specified in the recitals;

"Purchaser" shall have the meaning specified in the preamble;

"Reference Period" shall have the meaning specified in clause 3.1(3);

"Representatives" shall have the meaning specified in clause 4.1;

"Seller" shall have the meaning specified in the preamble;

"Services" means, collectively, the services specified in Schedule A hereto to be provided on an item by item basis;

"Service Term" means the term specified for each Service (on an item by item basis) as set out in Schedule A;

"Term" shall have the meaning specified in clause 2;

"Third Party Claim" shall have the meaning specified in clause 6(1);

"Third Party Provider" shall have the meaning specified in clause 3.1(7); and

"Transition Services Representatives" shall have the meaning specified in clause 3.3(1).

1.2 Construction

In this Agreement (a) the words "including", "includes" and "include" mean "including (or includes or include) without limitation", and (b) the phrase "the aggregate of", "the total of", "the sum of", or a phrase of similar meaning means "the aggregate (or total or sum), without duplication, of". Unless otherwise specified, the words "clause" and "Section" followed by a number mean and refer to the specified clause or Section of this Agreement. In the computation of periods of time from a specified date to a later specified date, unless otherwise expressly stated, the word "from" means "from and including" and the words "to" and "until" each mean "to but excluding".

2. TERM OF AGREEMENT

This Agreement shall take effect on the date hereof and, subject to earlier termination pursuant to clause 9.1, shall terminate automatically as of the end of the last day of the last Service Term as set out in Schedule A hereto (the "**Term**").

3. PROVISION OF THE SERVICES

3.1 The Services

- (1) Subject to the terms and conditions of this Agreement, the Seller shall provide, or cause to be provided, each Service from the date hereof until, subject to earlier termination pursuant to clause 9.1, the end of the last day of the Service Term for that Service (determined on an item by item basis). For clarity and subject to clauses 3.1(2) and 3.2, the Seller shall not be obligated to perform, for the benefit of the Purchaser, any services of any kind or nature whatsoever other than the Services. The Seller represents and warrants that fees for such Services are equal to the actual cost of such services to the Seller.
- (2) Following the date hereof, the Purchaser may request that the Seller provide additional services beyond the Services (each, an "Additional Service"), provided such Additional Services were performed by the Seller or its affiliates or third party contractors in respect of the Operations prior to the date hereof. The Seller shall provide such Additional Services for a term not to exceed the Term. The fees for such Additional Services shall be agreed upon by the parties and set out in an amendment to Schedule A. Such Additional Services shall be deemed to be Services for the purposes of this Agreement.
- (3) In providing, or causing to be provided, the Services, the Seller shall use its commercially reasonable efforts to ensure that such Services are of substantially the same nature, service level and quality, and provided with the same degree of care and diligence, and by the same Provider as the services provided for itself or its affiliates in the 12 month period immediately prior to the date hereof in the Ordinary Course of the Operations (the "Reference Period"), provided, however, that, notwithstanding anything contained herein, in providing the Services, no Provider shall be obligated to (i) hire or train additional employees, (ii) maintain the employment of any specific employee, (iii) purchase, lease or license any additional facilities, equipment or software, or (iv) otherwise deploy any commercially unreasonable efforts. The Seller shall implement prior to or on the date hereof an employee retention program for those employee of the Provider that will perform Services under this Agreement and in the event of the departure of any employee of the Provider performing Services during the Term, the Purchaser and the Seller shall work together to find a solution to replace any departing employees following the date hereof; provided that it is agreed between the Parties that such solution shall not require the Provider to hire any employee, independent contractor or Thirty Party Provider to replace such departing employee(s).
- (4) The Seller shall act in good faith in making operational or managerial decisions as to its use or management of personnel, technology and other resources used in relation to any Services. The Seller shall not be obligated to provide, or cause to be provided, any Services at a level of service which exceeds the volume, quantities or levels of the services provided in the Reference Period or, concerning the Services involving Third Party Providers, in a volume or quantity or at a level which exceeds what the

Third Party Providers have the obligation to provide pursuant to the agreements in place as of the date hereof.

- (5) The Purchaser agrees that access to and use of the Services is provided solely for the use and benefit of the Purchaser and GreenFirst and solely in support of the Operations for its orderly and timely transition, during the Term. The Purchaser shall not allow access to or use of the Services by any other Person or for any other purpose without the prior written consent of the Seller.
- (6) The Purchaser acknowledges and agrees that the Services provided hereunder are transitional in nature and are furnished solely for the purpose of effecting orderly and timely transition of the Operations from the Seller to the Purchaser and to support the Operations in the Ordinary Course during such transition period. The Purchaser acknowledges and agrees that the Seller is not in the business of providing services similar to the Services to third parties and that the Seller has no interest in continuing any Service beyond the applicable Service Term.
- (7) If the Seller is unable, using commercially reasonable efforts, to continue to provide the Services, the Seller may engage contractors, subcontractors, vendors or other third party providers (each, a "Third Party Provider") or its affiliates in connection with the provision of the Services and may delegate any of its duties to such Third Party Provider or affiliate as may be appropriate in the circumstances; provided that: (i) the Seller shall obtain prior written approval of Purchaser prior to engaging an affiliate or a Third Party Provider that is not engaged by the Seller as of the date hereof, which approval shall not be unreasonably withheld; (ii) the Seller shall continue to be responsible for the provision of the Services hereunder; (iii) such delegation or third-party retainer shall not relieve the Seller of any of its obligations under this Agreement; (iv) such affiliate or the Third Party Provider, as applicable, has the necessary knowledge, skills, qualifications and licences to provide such Services; and (v) the Purchaser shall be responsible for all fees and expenses of such Third Party Provider as approved in writing by the Purchaser in advance of Seller incurring such fees and expenses.
- (8) The Seller shall use commercially reasonable efforts to maintain in force current agreements with Third Party Providers. The Seller shall not be responsible for any inability to provide the Services in accordance with the terms hereof resulting from a termination of any agreement with a Third Party Provider by such Third Party Provider, other than due to any act or omission of the Seller. The Seller represents that it has no reason to believe that any Third Party Provider will cease to provide the Services that it currently provides for the full term that such Services are to be provided hereunder. The Seller shall provide prompt notice to the Purchaser in writing if it becomes aware of any actual or threatended termination of any agreement by a Third Party Provider, and shall reasonably cooperate with the Purchaser, at the Purchaser's sole cost and expense, to identify alternative solutions regarding Services relating to or impacted by such termination.
- (9) To the extent some of the Services are provided, as of the date hereof, or have ortherwise historically been provided, by an affiliate of the Seller, the Seller may continue to provide such Services through such affiliate without having to comply with the consent requirement set out in Section 3.1(7).

3.2 Change in the Services

- (1) Other than with respect to adding Additional Services, if the Purchaser wishes to change the nature, frequency, or the volume of the Services provided under this Agreement in any way (each, a "Change"), the Purchaser shall send to the Seller a written notice setting out in as much detail as reasonably possible the change required and the reason for requesting the change (the "Change Request"). The Seller shall diligently consider the Change Request with a view to implementing it if it can reasonably do so and respond to the Purchaser within five (5) Business Days following the date of the Change Request. If the Seller determines that it can implement the Changes, the Parties shall negotiate in good faith any related revisions to this Agreement, including any changes in fees relating to such Changes. Pending mutual agreement relating to any Changes, the Seller will continue to provide the Services in accordance with the latest agreed upon terms with respect to the Services.
- (2) The Seller may change the operational aspects of the Services or the way in which they are provided with the prior written approval of the Purchaser, which approval shall not be unreasonably withheld, conditioned or delayed; provided however, that no such modifications shall (i) materially reduce the availability, efficiency or quality of the Services or materially diminish the Services in any other way; (ii) result in an increase in Fees for the affected Services; or (iii) cause the Purchaser to incur additional material costs.

3.3 Cooperation and Access

- (1) Each Party shall cause its employees to reasonably cooperate with employees of the other Party to the extent required for effective provision of the Services. Without limiting the generality of the foregoing:
 - (a) each Party shall designate a point of contact person who shall be responsible for the daytoday implementation of this Agreement (collectively, the "Transition Services Representatives"), including attempted resolution of any issues that may arise during the performance of either Party's obligations hereunder;
 - (b) the Parties agree to correct or clarify the description of the Services specified in Schedule A in the event that such description is incomplete or unclear;
 - (c) each Party shall use commercially reasonable efforts to ensure that those of its personnel whose decisions are necessary for the performance and receipt of the Services are available at all reasonable times for consultation on any matter relating to the Services;
 - (d) each Party shall promptly provide to the other Party all information (including copies of data and documents) readily available to such Party and other assistance reasonably required by such Party relating to the Services;
 - (e) the Parties shall cooperate with each other in connection with the provision and receipt of the Services, including by developing reasonable procedures with respect to information sharing, transfer of data and similar matters; and

(f) each Party agrees that it shall, without charge, provide the other Party with such access to such Party's systems and personnel and facilities as may reasonably be required for the other Party to obtain or provide (as applicable) the Services. If a Party has access (either on-site or remotely) to any of the other Party's computer systems and/or information, such Party shall (i) comply with the other Party's applicable policies and procedures (including security); and (ii) limit such access solely to the use of such systems for purposes of receiving or providing (as applicable) the Services and shall not access or attempt to access any of the other Party's computer systems, files, software or services other than those required for its receipt or provision (as applicable) of the Services, or those that are publicly available (e.g., public websites).

3.4 Fees and Charges

- (1) In consideration for the Services provided by, or on behalf of, the Seller to the Purchaser under this Agreement, the Purchaser shall pay to the Seller the amounts set forth in Schedule A opposite each Service set forth therein (the "Fees").
- (2) The Seller shall deliver to the Purchaser no later than thirty (30) days following the last day of each month during the Term a statement of account reflecting a detailed breakdown of all Fees for all Services provided to the Purchaser in the preceding month pursuant to this Agreement, together with supporting documentation. Payment of the amounts due to the Seller pursuant to the foregoing shall be paid monthly within thirty (30) days following the date of receipt by the Purchaser of a statement of account, without set-off, defense or counterclaim.

4. CONFIDENTIALITY AND DATA PROTECTION

4.1 Confidentiality

The Parties acknowledge and agree that as a result of the relationship described hereunder, they may be privy to information regarding each other that is of a confidential or proprietary nature. As used herein, "**Confidential Information**" means all of the intellectual property, information, data and software created, maintained, used or owned by either Party and furnished or made accessible to the other, whether in oral, written, graphic or machine-readable form, which may include software code, software tool specifications, functions and features, integration and shared data block specifications, financial statements, corporate information, file layouts, marketing strategies, business, product or acquisition plans, current business relationships or strategies and customer and supplier lists; provided for the avoidance of doubt, any such intellectual property, information, data and software included in the Purchased Assets shall be the property of the Purchaser.

Each Party shall hold, and shall use reasonable efforts to cause its affiliates, officers, directors, employees, consultants, mandataries, agents, successors, representatives and advisors (collectively, the "**Representatives**") to hold, in confidence, unless compelled to disclose by judicial or administrative process or by other requirements of law or unless required to be disclosed in connection with the provision of the Services hereunder, all Confidential Information received from the disclosing Party which is now in the receiving Party's possession or which is subsequently furnished to the receiving Party by the disclosing Party, except to the extent that such Confidential Information can be shown to have been (i) in the public domain through no fault of the receiving Party or any of its Representatives, or (ii) later lawfully acquired by the receiving Party or any of its

Representatives on a non-confidential basis from sources other than the disclosing Party or any of its Representatives, or (iii) already known by the receiving Party prior to disclosure by the disclosing Party, or (iv) independently developed by the receiving Party without use or reference to Confidential Information disclosed by the disclosing Party; provided that the receiving Party may disclose such information to any of its Representatives on a need-to-know basis in connection herewith so long as such persons are contractually bound by confidentiality obligations as restrictive as set out herein. Each Party's obligation to hold any information in confidence hereunder shall be satisfied if it exercises the same care with respect to such information as it would take to preserve the confidentiality of its own similar information. If this Agreement is terminated, the confidentiality required by this clause 4.1 shall be maintained, and, except as otherwise provided herein, each Party shall, and shall use reasonable commercial efforts to cause its Representatives to, destroy or return upon request all Confidential Information, and all copies thereof, obtained from the other Party or any of its Representatives in connection herewith that are subject to such confidentiality, except copies that may be kept by the receiving Party for evidentiary or archival purposes.

4.2 Privacy and Data Protection

Without limiting the confidentiality obligations of any Party set out in clause 4.1, to the extent that a Party receives any personal information provided by another Party in connection with this Agreement, the receiving Party shall:

- (a) collect, use, disclose and otherwise handle such personal information only pursuant to the written instructions of the disclosing Party and to the extent necessary for the performance of Services and in accordance with applicable Laws; and
- (b) implement and maintain appropriate technical and organizational security measures in accordance with applicable Laws and to avoid accidental or unlawful destruction, accidental loss, alteration, unauthorized disclosure or access, and all other unlawful forms of collecting, using, disclosing or otherwise handling personal information.

4.3 Non-solicitation

None of the Purchaser nor any of its affiliates may, without the prior written consent of the Seller, directly or indirectly engage, recruit, solicit to employ or hire, offer employment to or hire (whether as an officer, employee, consultant or independent contractor) any employee of the Seller or its affiliates who is providing services under this Agreement for the term hereof and for a period of one (1) year thereafter; provided, however, that the foregoing prohibition shall not prohibit any (a) general solicitation for such person (including through the use of newspaper, trade journals, the internet, other media, employment agencies or search firms) not specifically directed at any such persons, or (b) hiring of any such person who responds to any such general solicitation without any other inducement.

5. Intellectual Property

(1) Notwithstanding anything to the contrary, the provision of the Services will in no way affect the ownership of any assets or intellectual property rights of the Seller. Except intellectual property rights relating to the Purchased Assets and the Operations, the Purchaser will not gain, by virtue of the receipt of Services, any rights of ownership in, to and under any intellectual property or other property owned by the Seller.

- (2) The Purchaser shall comply with the terms and conditions of any licenses to which the Seller or any of its affiliates are a licensee, including with respect to the term and termination of any such licenses, and which relate to the Services, to the same extent as if the Purchaser were the licensee under such licenses.
- (3) The Purchaser will be and remain the exclusive owner, as between itself and the Seller, of all data owned by the Purchaser (including data relating to the Purchased Assets and Operations) and submitted to the Seller as input and/or processed and all data and records relating to the Purchased Assets and the Operations that are generated and/or outputted in the course of providing any Service. The Purchaser hereby grants the Seller a limited, non-exclusive and non-transferable license to use such data solely for the purpose of providing the Services hereunder.

6. INDEMNIFICATION

(1) The Purchaser, on the one hand, and the Seller, on the other hand, shall each indemnify, defend and hold harmless the other Party and each of its respective affiliates, directors, officers, employees, agents and representatives (collectively, the "Indemnitees") from and against, and shall pay and reimburse each of the Indemnitees for, any and all Losses incurred or sustained by, or imposed upon, the Indemnitees to the extent arising out of claims made by third parties (a "Third Party Claim") relating to any breach of the relevant Party's obligations pursuant to this Agreement, provided, however, that no Indemnitee shall be subject to indemnification to the extent that such Losses arise out of their gross negligence or willful misconduct or breach of the confidentiality obligations hereunder.

7. LIMITATION OF Liability

- (1) In the event of any performance or non-performance under this Agreement which results in any Losses to the Purchaser, the Seller's maximum and cumulative liability to the Purchaser for such Losses shall not exceed the sums paid by the Purchaser to the Seller under this Agreement; provided, however, that such limitation shall not apply to exclude or limit the Seller's liability for fraud, gross negligence, willful misconduct or breach of confidentiality obligations hereunder. In the event of any performance or non-performance under this Agreement which results in any Losses to the Seller, the Purchaser's maximum and cumulative liability to the Seller for such Losses shall not exceed the sums paid by the Purchaser to the Seller under this Agreement, provided, however, that such limitation shall not apply to exclude or limit the Purchaser to the Seller under this Agreement, provided, however, that such limitation shall not apply to exclude or limit the Purchaser's liability for fraud, gross negligence, willful misconduct or breach of confidentiality obligations hereunder.
- (2) Neither Party shall be liable in connection with this Agreement for any Losses that are not reasonably forseeable or are indirect in nature. This limitation shall not apply in respect of such Losses that result from that Party's breach of clause 4.
- (3) Neither Party will bear any responsibility or liability for any claims, obligations, costs, expenses or other losses arising out of any delay, inability to perform or interruption of its performance of obligations under this Agreement due to any acts or omissions of the other Party or for events that are beyond its reasonable control, and is not avoidable by the exercise of reasonable care, including strikes, riots, acts of war, acts of terrorism, pandemics, epidemics, fire, earthquakes, floods or other natural disasters (each a "Force Majeure Event"), provided, however, that such Party notifies the other Party thereof without undue delay after such Force Majeure Event

has occurred. Both Parties will in any event use reasonable endeavours to avoid or mitigate the effect of a Force Majeure Event so as to recommence performance of their obligations as soon as reasonably possible and shall keep the other Party reasonably informed with respect to its activities in connection therewith.

8. DISCLAIMER OF WARRANTY

Except as otherwise provided in this Agreement, the Seller makes no warranty, express or implied, and hereby disclaims any warranties of any kind with respect to, (a) the nature, condition or quality of any Services or any product, materials, components or services obtained or provided pursuant to this Agreement, or (b) the results that will be obtained by using, receiving, or applying any such Services or product, materials, components or services, in each case including any express or implied warranty.

9. TERMINATION

9.1 Termination

- (1) The Seller may, by giving written notice to the Purchaser, terminate this Agreement if any of the following occur:
 - the Purchaser commits a material breach of this Agreement, which breach has not been cured within thirty (30) days after notice of breach is given to the Purchaser (or such longer period of time as mutually agreed by the Parties);
 - (b) the Purchaser (i) is declared insolvent or bankrupt by a court of competent jurisdiction, (ii) files a voluntary petition of bankruptcy or notice of proposal in any court of competent jurisdiction, (iii) assigns this Agreement for the benefit of creditors, or (iv) otherwise files or seeks protection from its creditors under any applicable bankruptcy or insolvency legislation.
- (2) The Purchaser may, by giving written notice to the Seller, terminate this Agreement if any of the following occur:
 - (a) the Seller commits a material breach of this Agreement, which breach has not been cured within thirty (30) days after notice of breach is given to Seller (or such longer period of time as mutually agreed by the Parties);
 - (b) the Seller (i) is declared insolvent or bankrupt by a court of competent jurisdiction, (ii) files a voluntary petition of bankruptcy or notice of proposal in any court of competent jurisdiction, (iii) assigns this Agreement for the benefit of creditors, or (iv) otherwise files or seeks protection from its creditors under any applicable bankruptcy or insolvency legislation.
- (3) Notwithstanding any other provision of this Agreement, the expiry or earlier termination of this Agreement in accordance with clause 9.1 shall not discharge, affect or otherwise modify in any manner the rights and obligations of the Parties hereto which have accrued or been incurred prior to such termination.

9.2 Effect of Termination

Upon termination or expiry of any Service pursuant to this Agreement, the Seller will have no further obligation to provide the terminated or expired Service. In connection with termination of any Service, the provisions of this Agreement not relating solely to such
terminated or expired Service will survive any such termination or expiration, and in connection with a termination or expiration of this Agreement, any terms contemplating performance subsequent to such termination or expiration will continue to survive indefinitely.

10. Dispute resolution

Any dispute or difference between the Parties arising out of or in connection with this Agreement (a "Dispute") shall be settled in accordance with the following process:

- (1) Any Dispute shall be initially referred to the Transition Services Representatives, which shall meet as soon as reasonably possible after a Dispute is referred to it, giving due regard to the nature and impact of the issue under consideration. If the Transition Services Representatives cannot resolve a Dispute referred to it within three (3) Business Days, the Dispute shall be referred to the Executives (as defined below). In the event that the Transition Services Representatives believe, acting reasonably, that there is no possibility that the Dispute will be resolved by the Transition Services Representatives, the Dispute shall be immediately escalated to the Executives, even if the time period specified herein has not yet expired.
- (2) The Chief Financial Officer of Purchaser and the [] of Seller (collectively, the "Executives") will meet to consider any Dispute that is referred to them by the Transition Services Representatives, as soon as reasonably possible after such referral but in any event within three (3) Business Days, giving due regard to the nature and the impact of the issue under consideration. If the Dispute is resolved with the unanimous agreement of all Executives, the Executives shall rectify the Dispute in the manner agreed to.
- (3) If the Executives cannot resolve a Dispute referred to them, within seven (7) Business Days of meeting to consider the Dispute, or if either Executive believes, acting reasonably, that there is no possibility that the Dispute will be resolved by the Executives, either Executive may, regardless of the time period specified herein, initiate arbitration with respect to such Dispute, pursuant to the Arbitration Act, 1991 (Ontario) except in the event that the Executives agree in writing to an alternative dispute resolution mechanism. Unless the Parties otherwise agree, the arbitration will be conducted in accordance with the provisions of the National Arbitration Rules of the ADR Institute of Canada, Inc. (the "ADR Rules") and, subject to Rule 3 of the ADR Rules, the Arbitration Act, 1991 (Ontario), based upon the following: (i) there will be one (1) arbitrator agreed to by the Parties. If the Parties cannot agree on the arbitrator, the arbitrator shall be appointed in accordance with the ADR Rules; (ii) notwithstanding anything to the contrary in the ADR Rules, the arbitrator will be instructed to conduct the arbitration proceeding in an expeditious and efficient manner having regard to the nature of such Dispute and the values at risk. In all cases the Parties will instruct the arbitrator to deliver the final award within three (3) months of the date of the notice from one Party to the other Party stating that such Dispute will be referred to arbitration, unless the Parties otherwise agree; (iii) the place of arbitration will be Toronto, Ontario; (iv) the language of the arbitration will be English; and (v) all arbitrations will be subject to the jurisdiction of the courts of Ontario and any awards will be enforceable in the courts of Ontario and in any other court of competent jurisdiction.
- (4) Nothing in this Agreement, including the above dispute resolution procedure, will limit or prevent a Party from seeking interim relief, including by injunction or order for the detention, preservation or inspection of property, or from seeking to enforce the

performance of this Agreement by injunction or specific performance upon application to a court of competent jurisdiction.

11. GENERAL

11.1 Currency

Unless otherwise specified in this Agreement, all references to currency, monetary values and dollars set forth herein shall mean United States (U.S.) dollars.

11.2 Records and Audits

The Seller shall keep and maintain books and records related directly to the performance of the Services consistent with its customary accounting and business practices to the extent applicable as such practices currently exist and as they may change from time to time. At the Purchaser's or a governmental or regulatory authority's request, the Seller shall provide the Purchaser or such governmental or regulatory authority with access to such records for the purpose of copying and/or auditing them, during normal business hours at the Seller's offices, which access shall be provided (i) with the least interference possible to the operations of the Seller and consistent with good commercial practices, (ii) subject to the Purchaser or such governmental or regulatory authority complying with the physical and logical security practices of the Seller, and (iii) subject to the confidentiality obligations of the Purchaser as set out in this Agreement or such governmental or regulatory authority agreeing to reasonable confidentiality obligations.

11.3 No Partnership or Agency

Nothing in this Agreement or any circumstances associated with it or its performance give rise to any relationship of agency, partnership or employer and employee between the Parties and nothing in this Agreement will be deemed in any way or for any purpose to constitute any Party an agent of the other Party in the conduct of such other Party's business. No employee of the Seller will be deemed to be the employee or agent of the Purchaser and nothing in this Agreement shall be construed to make the Purchaser an employer, directly or indirectly, of the Seller employees.

11.4 Assignment

Neither Party may assign or transfer any of its rights, obligations or privileges (by operation of law or otherwise) hereunder without the prior written consent of the other Party.

11.5 Waiver

No delay or omission by either Party in enforcing or exercising any right, power or remedy will impair that right, power or remedy or be construed to be a waiver of it. A waiver by either Party of any of its rights, powers or remedies or of any breach will not be construed to be a waiver of any other right, remedy or power or any succeeding breach. No waiver or discharge of any kind will be valid unless in writing and signed by an authorized representative of the Party against whom such waiver or discharge is sought to be enforced.

11.6 Notices

Any notice, direction or other communication given regarding the matters contemplated by this Agreement (each a "**Notice**") must be in writing, sent by personal delivery, courier facsimile or electronic mail (with read receipt included) and addressed:

To the Seller:

Rayonier Advanced Materials Inc. 1301 Riverplace Boulevard Suite 2300 Jacksonville, Florida 32207 United States of America

Attention: Chief Financial Officer with a copy to: General Counsel

with a copy to:

McCarthy Tétrault LLP 1000, De La Gauchetière Street West Suite 2500 Montréal, Québec H3B 0A2

 Attention:
 Clemens Mayr and Chrystelle Chevalier-Gagnon

 Email:
 cmayr@mccarthy.ca

To the Purchaser:

9437-6001 Quebec Inc. 1800 - 510 West Georgia Street Vancouver, British Columbia V6B 0M3

Attention: Chief Financial Officer

with a copy to:

Norton Rose Fulbright Canada LLP Suite 3000 222 Bay St. Suite 3000, Toronto, Ontario M5K 1E7

 Attention:
 Walied Soliman and Trevor Zeyl

 Email:
 wailed.soliman@nortonrosefulbright.com
 and trevor.zeyl@nortonrosefulbright.com

A Notice is deemed to be delivered and received on the date of delivery if it is a Business Day and the delivery was made prior to 4:00 p.m. (local time in place of receipt) and otherwise on the next Business Day. A Party may change its address for service from time to time by providing a Notice in accordance with the foregoing. Any subsequent Notice must be sent to the Party at its changed address. Any element of a Party's address that is not specifically changed in a Notice shall be assumed not to be changed. Sending a copy of a Notice to a Party's legal counsel as contemplated above is for information purposes only and does not constitute delivery of the Notice to that Party. The failure to send a copy of a Notice to legal counsel does not invalidate delivery of that Notice to a Party.

11.7 Severability

If a court of competent jurisdiction or other competent body decides that any provision of this Agreement is void or otherwise ineffective but would be valid and effective if appropriately modified then such provision will apply with the modification necessary to make it valid and effective. If such a provision cannot be so modified, the provisions' invalidity or ineffectiveness will not affect or impair the validity or legal effect of any other provision of this Agreement.

11.8 Entire Agreement and Amendment

This Agreement and any schedules hereto set out the entire agreement between the Parties with respect to the subject matter of this Agreement and supersedes all prior representations, agreements, negotiations or understandings, whether oral or in writing, between or on behalf of the Parties hereto. No modification or amendment of this Agreement will be binding upon any Party unless in writing, expressly referring this Agreement and duly executed by each Party.

11.9 Conflict

In the event that there is any conflict or inconsistency between these terms and conditions and the Schedules then unless the Schedules specifically provide otherwise, these terms and conditions will prevail.

11.10 Counterparts

This Agreement may be executed in several counterparts, all of which parts taken together, whether provided in original or electronic form, when duly exchanged and delivered, will be deemed to be an original, and will constitute one single agreement between the Parties.

11.11 Third Party Consents

Seller will use commercially reasonable efforts to obtain all waivers, permits, consents or sublicenses to the extent reasonably necessary for purposes of provision of the Services on terms and conditions acceptable to the Purchaser, acting reasonably. In the event that a Person that is not a Party hereto or an affiliate of a Party hereto requires the payment of a fee or other charge to permit the Seller to provide Services to the Purchaser, then the Purchaser agrees that, provided it has consented to the terms and conditions thereof in accordance with the preceding sentence, it shall pay or reimburse the Seller all such fees in full following receipt of an invoice for same.

11.12 Governing Law

- (1) This Agreement shall be governed by and interpreted and enforced in accordance with the laws of the Province of Ontario and the laws of Canada applicable therein.
- (2) Each Party irrevocably attorns and submits to the exclusive jurisdiction of the Ontario courts situated in the City of Toronto and waives objection to the venue of any Proceeding in such court or that such court provides an inconvenient forum.

[Signature page follows.]

IN WITNESS WHEREOF the Parties have executed this Agreement.

RAYONIER A.M. CANADA G.P., herein represented by its managing partner, RAYONIER A.M. CANADA INDUSTRIES INC.

By:

Authorized Signatory

9437-6001 QUEBEC INC.

By:

Authorized Signatory

[signature page to transition services ageement]

SCHEDULE "A" SERVICES

EXHIBIT E GST and QST CERTIFICATE

[**]

EXHIBIT F

PENSION PLAN ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS AGREEMENT is made as of the [•], 2021.

BETWEEN:

[RAYONIER A.M. CANADA G.P.], a general partnership formed under the laws of the Province of Québec (the "**Plan Sponsor**"),

- and -

ITASCA CAPITAL LTD., a corporation incorporated under the laws of the Province of **[British Columbia]** (the "New Plan Sponsor").

RECITALS:

- A. The Plan Sponsor maintains: (i) the Retirement Plan for Organized Employees of Tembec Enterprises Inc.- Kapuskasing Operations (Reg. No. 0342352), (ii) the Régime de rentes pour les employés de Tembec Industries Inc. (Taschereau et employés de bureau de La Sarre (Retraite Québec no. 30723 CRA no. 969535) and (iii) the Régime de retraite des employés syndiqués de Tembec Industries Inc. Division La Sarre N (opérations forestières, garage, sciage et rabotage) (Retraite Québec no. 30729 CRA no. 691865) (collectively, the "Plans");
- B. The defined benefit components of the Plans are funded through a trust agreement with CIBC Mellon and there is a custodian agreement with Manulife for the defined benefit components (each of CIBC Mellon and Manulife, or their respective successor(s) (a "Funding Agent") and each of the trust agreements or custodial agreements (a "Funding Agreement"));
- C. The Plan Sponsor and Rayonier A.M. Canada Industries Inc., as sellers, and the New Plan Sponsor, as purchaser entered into an asset purchase agreement dated as of [•], 2021 (the "**Purchase Agreement**");
- D. Pursuant to the Purchase Agreement, the Plan Sponsor has agreed to transfer sponsorship of the Plans to the New Plan Sponsor and, in connection therewith, to execute and deliver this assignment of its right, title and interest in, to and under the Plans;
- E. The New Plan Sponsor has agreed to assume sponsorship of the Plans together with all right, title and interest in, to and under the Plans; and

F. [The Plan Sponsor and the New Plan Sponsor desire to provide for the continued appointment of the Funding Agents.]

NOW THEREFORE in consideration of the mutual covenants and agreements contained in this Agreement and other good and valuable consideration (the receipt and sufficiency of which are acknowledged), the Parties agree as follows:

- 1. <u>Definitions</u>. The capitalized terms used in this Agreement shall have the meaning ascribed to them in the Purchase Agreement unless otherwise set out herein. For purposes of this Agreement "Parties" means the Plan Sponsor and the New Plan Sponsor.
- Assignment and Assumption of Plans. The Plan Sponsor hereby assigns and transfers unto the New Plan Sponsor all of the 2. Plan Sponsor's right, title and interest in, to and under all of the Plans, including the Funding Agreements. The New Plan Sponsor hereby accepts the assignment and transfers of the Plan and covenants and agrees with the Plan Sponsor that, from and after the date hereof, the New Plan Sponsor will observe, perform and fulfill each and every covenant, proviso, obligation, term and condition of, or applicable to, the Plan Sponsor under the Plans that is applicable at any time from and including the date of this Agreement, to the full and complete exoneration of the Plan Sponsor. The New Plan Sponsor hereby agrees to fully indemnify and save harmless the Plan Sponsor from against any and all liabilities, damages, costs, expenses, causes of action, suits, claims and judgments arising from or in connection with, or resulting from, any breach by the New Plan Sponsor of its obligations hereunder or any breach by the New Plan Sponsor or those for whom the New Plan Sponsor is legally responsible of their obligations or liabilities under the Plans, including the Funding Agreements. The Plan Sponsor hereby agrees to fully indemnify and save harmless the New Plan Sponsor from against any and all liabilities, damages, costs, expenses, causes of action, suits, claims and judgments arising from or in connection with, or resulting from, any breach by the Plan Sponsor of its obligations hereunder or any breach by the Plan Sponsor or those for whom the Plan Sponsor was legally responsible of their obligations or liabilities under the Plans, including the Funding Agreements.
- 3. <u>Conforming Amendments</u>. The Plans, the Funding Agreements and any other agreements between the Plan Sponsor and the Funding Agents in relation to the Plans shall be formally amended where required to give effect to this Agreement, and the Plan Sponsor and the New Plan Sponsor shall fully co-operate with each other in connection with the preparation of such amendments and all requisite filings with the appropriate federal and provincial regulatory authorities. The Funding Agents shall fully co-operate with the New Plan Sponsor in connection with the preparation of any necessary amendments to the Funding Agreements and any other agreements between the Plan Sponsor and the Funding Agents in relation to the Plans.
- 4. <u>Continuation of Plans</u>. It is hereby acknowledged that nothing in this Agreement shall operate in any way as a discontinuance or wind-up of the Plans which are hereby confirmed to continue in accordance with the terms thereof or applicable thereto. For greater certainty, in respect of any Plan that is subsequently wound up, the Parties acknowledge that any amounts payable by or owing to the sponsor of such Plan shall,

on and after the date hereof, become payable by or owing to the New Plan Sponsor, as successor sponsor.

- 5. <u>Severability</u>. If any provision of this Agreement is determined to be illegal, invalid or unenforceable by an arbitrator or any court of competent jurisdiction, that provision will be severed from this Agreement and the remaining provisions shall remain in full force and effect.
- 6. <u>Further Assurances</u>. Each Party, upon the request of the other, agrees to perform such further acts and deliver such further documents as may be reasonably necessary to effectively carry out the terms and intent of this Agreement.
- 7. Non-Assignment. This Agreement may not be assigned by either Party without the written consent of the other Party.
- 8. <u>Successors and Assigns</u>. This Agreement shall enure to the benefit of and be binding upon the Parties hereto and their respective successors and permitted assigns.
- 9. <u>Amendment</u>. This Agreement may be amended, supplemented or otherwise modified by written agreement signed by the Parties.
- 10. <u>Entire Agreement</u>. This Agreement and the Purchase Agreement constitutes the entire agreement between the Parties with respect to the transactions contemplated herein and supersedes all prior agreements, understandings, negotiations and discussions, whether oral or written, of the Parties.
- 11. <u>Governing Law</u>. This Agreement shall be construed in accordance with the laws of the Province of Ontario and the laws of Canada applicable therein. Each Party irrevocably attorns and submits to the exclusive jurisdiction of the Ontario courts situated in the City of Toronto and waives objection to the venue of any proceeding in such court or that such court provides an inconvenient forum.
- 12. <u>Counterparts</u>. This Agreement may be executed by the Parties in separate counterparts, including by way of facsimile, each of which when so executed and delivered shall be an original and shall be binding on the signatory Party, but all such counterparts shall together constitute one and the same instrument.

[Remainder of page intentionally left blank. Signature page follows.]

IN WITNESS WHEREOF the Parties have executed this Agreement on the day and year first above written.

RAYONIER A.M. CANADA G.P., herein represented by its managing partner, RAYONIER A.M. CANADA INDUSTRIES INC.

By:

Name:

Title:

ITASCA CAPITAL LTD.

By:

Name:

Title:

Exhibit G Chip Purchase Setoff Note

Dated: [Closing Date]

Amount: C\$7,900,000

FOR VALUE RECEIVED, the undersigned, 9437-6001 Quebec Inc. (the "**Maker**"), acknowledges itself indebted and unconditionally promises to pay to, or to the order of, (the "**Holder**") at its offices at or such other place as the Holder may, from time to time, designate, the principal amount of Seven Million Nine Hundred Thousand (Cdn\$7,900,000) DOLLARS in lawful money of Canada which portions of the principal amount shall be paid on or before the dates set out in the following table.

Principal Amount	Due Date
C\$1,580,000	April [•], 2022
C\$1,580,000	April [•], 2023
C\$1,580,000	April [•], 2024
C\$1,580,000	April [•], 2025
C\$1,580,000	April [•], 2026

The Maker has the right and privilege of prepaying the whole or any portion of the principal amount of this Note at any time or times without notice, bonus or penalty. In addition, the Maker and the Holder, each have the right and privilege of setting off the principal amount owing under this note against any amounts owing by the Holder or its affiliates to the Maker or its affiliates under that certain Chip Purchase Agreement dated the $[\bullet]$, 2021 among 9437-6001 Quebec Inc. and Rayonier A.M. Canada G.P. The principal amount remaining from time to time unpaid and outstanding shall not bear interest.

The recording by the Holder in its accounts of principal amounts owing and repayments is, in the absence of manifest mathematical error, *prima facie* evidence of such advances and payments provided that the failure of the Holder to record same shall not affect the obligation of the Maker to pay such amounts to the Holder.

To the fullest extent permitted by law, the Maker waives:

- (a) diligence, presentment, demand and protest, and notice of presentment, dishonour, intent to accelerate, acceleration, protest, non-payment, release, compromise, settlement, extension or renewal of this Note; and
- (b) the benefit of all applicable valuation, appraisal and exemption laws.

The Maker agrees that all amounts under this Note are payable without set-off, withholding, deduction, claim, counterclaim, defence or recoupment, all of which are hereby waived by the Maker.

Time is of the essence with this Note.

This Note is binding upon the Maker and its successors and assigns and enures to the benefit of the Holder and its successors and assigns. No party hereto may not any of its rights or obligations hereunder.

This Note is governed by and is to be interpreted and enforced in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein. The Maker irrevocably attorns to the exclusive jurisdiction of the courts of Ontario.

IN WITNESS WHEREOF the Maker has executed and delivered this Note as of the date first above written.

9437-6001 Quebec Inc.

By:

Authorized Signing Officer

Exhibit H Support in Forestry Matters

Name of Service	Support in Forestry Matters
	In the event that [**] accepts an offer of employment from the Purchaser or one of its affiliates at Closing, the Purchaser undertakes to make [**] available to provide support in forestry matters to the Seller.
	During the Service Term(s), [**] will spend up to 50% of his time providing services to the Seller.
	Workplan for [**] to be established by the Seller and the Purchaser, using commercially reasonable efforts, following the closing of transaction.
	Notwithstanding any other provision in this schedule or in the transition services agreement, this service will terminate immediately upon [**] death, incapacity or in the event he is no longer employed with the Purchaser for any reason.
Service Fees	[**]
Service Term	[**]

Exhibit 10.1

Certification

I, Paul G. Boynton, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Rayonier Advanced Materials Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rule 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 6, 2021

/s/ PAUL G. BOYNTON

Paul G. Boynton President and Chief Executive Officer Rayonier Advanced Materials Inc.

Certification

I, Marcus J. Moeltner, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Rayonier Advanced Materials Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rule 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 6, 2021

/s/ MARCUS J. MOELTNER

Marcus J. Moeltner Chief Financial Officer and Senior Vice President, Finance Rayonier Advanced Materials Inc.

Certification

The undersigned hereby certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that to our knowledge:

- 1. The quarterly report on Form 10-Q of Rayonier Advanced Materials Inc. (the "Company") for the period ended March 27, 2021 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

May 6, 2021

/s/ PAUL G. BOYNTON

/s/ MARCUS J. MOELTNER

Paul G. Boynton President and Chief Executive Officer Rayonier Advanced Materials Inc. Marcus J. Moeltner Chief Financial Officer and Senior Vice President, Finance Rayonier Advanced Materials Inc.